

Analyse fonctionnelle, spectrale et probabiliste sur domaines d'extension de Sobolev

Gabriel Claret

► To cite this version:

Gabriel Claret. Analyse fonctionnelle, spectrale et probabiliste sur domaines d'extension de Sobolev. Mathématiques [math]. 2023. dumas-04467822

HAL Id: dumas-04467822 https://dumas.ccsd.cnrs.fr/dumas-04467822

Submitted on 20 Feb 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





Thèse de Master

Analyse fonctionnelle, spectrale et probabiliste sur domaines d'extension de Sobolev

Gabriel Claret

Laboratoire MICS, CentraleSupélec, Université Paris-Saclay Parcours Recherche – CS+R

Membres du jury:

Claude Bardos(Examinateur)Xavier Claeys(Rapporteur)Erick Herbin(Examinateur)Fabien Imbault(Examinateur)Anna Rozanova-Pierrat(Encadrante)Alexander Teplyaev(Examinateur)

Acknowledgments

I thank my supervisor Anna Rozanova-Pierrat for her guidance and her time during this internship, as well as in the past three years during which she made me discover the world of research and gave me the opportunity to learn from her, meet with many researchers, and trusted me with captivating projects.

My thanks go to Alexander Teplyaev for very interesting pointers and conversations on the topic, which give me meaningful insight and allowed me to have a better grasp on this complex subject.

I thank Michael Hinz for helping define the scope of this thesis as well as enlightening discussions which helped shape this project into what it has become.

I also thank Simon Chandler-Wilde and David Hewett for their advice on important results of this thesis.

Finally, I thank Xavier Claeys for having agreed to be referee for this thesis and for his time and commitment, as well as Erick Herbin, Fabien Imbault and Claude Bardos for agreeing to be on the jury.

Contents

1	Introduction	1
2	Framework 2.1 Domain regularity 2.2 Boundary value problems on extension domains 2.3 Probabilistic perspective 2.3.1 Variational-probabilistic correspondence 2.3.2 Probabilistic incorporation of boundary conditions	4 4 8 13 13 13 13
3	Bounded problems	17
	 3.1 Harmonic extension operators	. 18 . 19
4	Transmission problems	21
	4.1 Layer potential operators	. 22
	4.2 Approximation by truncated problems	. 25
	4.3 Neumann-Poincaré operator and boundary values	. 25
	4.4 Poincaré-Steklov operators for transmission problems	. 27
	4.5 Norm estimates and Neumann series	. 28
	4.6 Imaging	. 29
5	Hilbert transform and Cauchy integral	31
	5.1 Lipschitz case: connection with the layer potential operators .	. 32
	5.2 Non-Lipschitz case	. 34
Α	Bounded problems on extension domains	35
	A.1 Harmonic extension operators	. 35
	A.2 Poincaré-Steklov operator	. 42
В	Transmission problems for extension domains	45
	B.1 Layer potential operators	. 45
	B.2 Approximation by truncated problems	. 57
	B.3 Neumann-Poincare operators and boundary values	. 02 67
	B.4 Poincare-Stekiov operators for transmission problems	. 07 69
	B.5 Norm estimates and Neumann series	00 72
	B.6.1 Representation formula	- 12 79
	B.6.2 Subdomain identification	. 12 77
С	Hilbert transform and Cauchy integral on extension domains	80
	U.1 Lipschitz approximation of arbitrary domains	. 80
	U.2 Convergence of the Dirichlet harmonic extension operators	. 82
	U.2.1 Convergence along a sequence of Hilbert spaces	. 82
	U.2.2 Uase of the Dirichlet harmonic extension operators	. 86

Notations

General notations

We will use the following notations:

- a := b when the object a is defined as equal to the object b;
- *n* is an integer non-inferior to 2 (unless stated otherwise) referring to the dimension of the ambient space;
- a domain of \mathbb{R}^n is a nonempty open connected set of \mathbb{R}^n ; all domains considered coincide with the interior of their closure;
- $\lambda^{(n)}$ is the *n*-dimensional Lebesgue measure on \mathbb{R}^n ;
- σ denotes the (n-1)-dimensional Lebesgue measure on the boundary of a Lipschitz domain of \mathbb{R}^n (regardless of that domain);
- if $x \in \mathbb{R}^n$ and r > 0, $B_r(x)$ denotes the open Euclidean ball of center x and radius r;
- if A is a Borel subset of \mathbb{R}^n and $k \in \mathbb{N} \cup \{\infty\}$, $C^k(A)$ denotes the set of real-valued k-times continuously differentiable functions on A, and $C_0^k(A)$ those of them which have a compact support in A;
- compact inclusions are denoted by $\subset \subset$;
- if H is a real Banach space, then $\|\cdot\|_H$ denote the associated norm. $H' := \mathcal{L}(H, \mathbb{R})$ denotes its dual space and $\langle \cdot, \cdot \rangle_{H',H}$ denotes the associated duality pairing; if H is a Hilbert space, $\langle \cdot, \cdot \rangle_H$ denote the associated inner product;
- if K is a subspace of a Hilbert space H, then K^{\perp} denotes the orthogonal complement of K in H;
- if $L: V \to H$ is a bounded linear operator between Hilbert spaces, then $L^*: H' \to V'$ denotes its adjoint operator;
- if Ω is a domain of \mathbb{R}^n (or $\mathbb{C} \simeq \mathbb{R}^2$), then its boundary is denoted by $\partial \Omega$ and its complement domain by $\Omega^c := \mathbb{R}^n \setminus \overline{\Omega}$.

 L^2 spaces will always be identified with their dual spaces, and other Hilbert spaces will always be identified to their bidual according to the Riesz representation theorem.

The term 'well-posedness' of a differential equation is understood in the sense of Hadamard: existence and uniqueness of a solution, which depends continuously on the data (such as boundary conditions).

Notations for functional spaces

If U is an open set of \mathbb{R}^n and X(U) denotes a Hilbert space of real-valued functions defined on U, then:

- X'(U) := (X(U))';
- $u \in X_{\Delta}(U)$ if $u \in X(U)$ and $\Delta u \in L^2(U)$;
- $u \in X_{\langle 0 \rangle}(U)$ if $u \in X(U)$ and $\int_U u \, dx = 0$; if $\Omega \subset U$ is open, $u \in X_{\langle 0 \rangle_\Omega}(U)$ if $u \in X(U)$ and $\int_\Omega u \, dx = 0$;
- $u \in X_{[f]_{\mathrm{Tr}}}(U)$ if $u \in X(U)$ and $\llbracket \mathrm{Tr} u \rrbracket = f$ in the sense of Definition 2.14;
- $u \in X_{[g]_{\partial_{\nu}}}(U)$ if $u \in X(U)$ and $[\![\frac{\partial u}{\partial \nu}]\!] = g$ in the sense of Definition 2.14;

-
$$X_{\mathbb{C}}(U) := X(U) + iX(U)$$
 where $i^2 = -1$.

The notations defined above can be combined.

Specific function spaces

In addition to the usual Sobolev spaces, spaces relevant to our study are listed hereafter. An equality is said to hold 'weakly' if it holds in the distributional sense. By 'gradient norm' is meant the norm $\|\nabla \cdot\|_{L^2}$ and sums of norms are understood as the square root of the sum of the norms squared. If Ω is a one-sided extension domain (Definition 2.5):

- $H^1_f(\Omega) := \left\{ u \in H^1(\Omega) \mid \operatorname{Tr}_i u = f \right\}$ (Definition 2.8);
- $V_1(\Omega) := \{ u \in H^1(\Omega) \mid (-\Delta + 1)u = 0 \text{ weakly on } \Omega \}$ endowed with the H^1 norm on Ω ; $V_1(\Omega^c)$ is defined in the same way replacing Ω with Ω^c ;
- $V_{\langle 0 \rangle}(\Omega) := \left\{ u \in H^1_{\langle 0 \rangle}(\Omega) \mid -\Delta u = 0 \text{ weakly on } \Omega \right\}$ endowed with the gradient norm on Ω ;

-
$$\mathcal{B}_0(\partial\Omega) := \operatorname{Tr}_i \Big(H^1_{\langle 0 \rangle}(\Omega) \Big),$$

$$\mathcal{B}_{0,ext}'(\partial\Omega) := \{g \in \mathcal{B}'(\partial\Omega) \mid \langle g, 1 \rangle_{\mathcal{B}',\mathcal{B}} = 0\};\$$

- if Ω is Lipschitz, $L^2_0(\partial\Omega,\sigma) := \left\{g \in L^2(\partial\Omega,\sigma) \ \Big| \ \int_{\partial\Omega} g \,\mathrm{d}\sigma = 0 \right\}.$

If Ω is a two-sided extension domain (Definition 2.5):

- $H^1_{\partial\Omega}(\mathbb{R}^n) := \left\{ u \in L^2(\mathbb{R}^n) \mid \nabla u|_{\Omega \cup \Omega^c} \in L^2(\Omega \cup \Omega^c)^n \right\}$ endowed with the sum of the L^2 norm on \mathbb{R}^n and the gradient norm on $\Omega \cup \Omega^c$;
- $V_{1,\partial\Omega}(\mathbb{R}^n) := \left\{ u \in H^1_{\partial\Omega}(\mathbb{R}^n) \mid -\Delta u + u = 0 \text{ weakly on } \mathbb{R}^n \setminus \partial\Omega \right\}$ endowed with the sum of the L^2 norm on \mathbb{R}^n and the gradient norm on $\Omega \cup \Omega^c$;

- if n = 2, $V_{van}(\mathbb{R}^2) := \left\{ u \in H^1_{\partial\Omega}(\mathbb{R}^2) \mid u \to 0 \text{ as } |x| \to +\infty \right\};$
- if $n \geq 3$, $W_{\partial\Omega}(\mathbb{R}^n) := \left\{ u \in L^2_{loc}(\mathbb{R}^n) \mid \nabla u \mid_{\Omega \cup \Omega^c} \in L^2(\Omega \cup \Omega^c)^n, \ \langle u \rangle_{\infty} = 0 \right\}$ (defined by Equation (B.12));
- $V_{\partial\Omega}(\mathbb{R}^n) := V_{van}(\mathbb{R}^2)$ if n = 2 and $V_{\partial\Omega}(\mathbb{R}^n) := W_{\partial\Omega,\langle 0 \rangle_{\Omega}}(\mathbb{R}^n)$ endowed with the gradient norm on $\Omega \cup \Omega^c$ if $n \ge 3$.

If Ω is a two-sided extension domain and U is a one-sided extension domain with $\Omega \subset \subset U$:

$$- H^1_{0,\partial\Omega}(U) := \left\{ u \in L^2(U) \mid \nabla u |_{U \setminus \partial\Omega} \in L^2(U \setminus \partial\Omega)^n, \ \mathrm{Tr}^{\partial U} \, u = 0 \right\}.$$

If Ω and D are two-sided extension domains with $D \subset \subset \Omega$:

-
$$H^1_{\partial D}(\Omega) := \left\{ u \in L^2(\Omega) \mid u|_D \in H^1(D), \ u|_{\Omega \setminus \overline{D}} \in H^1(\Omega \setminus \overline{D}) \right\};$$

- $u \in W_{\partial D,\partial\Omega}(\mathbb{R}^n)$ if $u \in L^2_{loc}(\mathbb{R}^n)$, $\nabla u|_{\mathbb{R}^n \setminus (\partial D \cup \partial\Omega)} \in L^2(\mathbb{R}^n \setminus (\partial D \cup \partial\Omega))^n$ and $\langle u \rangle_{\infty} = 0$ (defined by Equation (B.12)).

Operators

Operators relevant to our study are listed hereafter. Note that most of them are defined with and without a tilde: the operators connected to problems for $(-\Delta + 1)$ do not have a tilde while those connected to problems for $-\Delta$ do.

- I denotes the identity operator (regardless of the space).

If Ω is a one-sided extension domain:

- $\operatorname{Tr}_i : H^1(\Omega) \to \mathcal{B}(\partial\Omega)$ is the interior trace operator; it can be seen as $\operatorname{Tr}_i : H^1_{(0)}(\Omega) \to \mathcal{B}_0(\partial\Omega);$
- $\frac{\partial_i}{\partial \nu}$: $H^1_{\Delta}(\Omega) \to \mathcal{B}(\partial \Omega)$ is the weak interior normal derivation operator; it can be seen as $\frac{\partial_i}{\partial \nu}$: $V_{\langle 0 \rangle}(\Omega) \to \mathcal{B}'_0(\partial \Omega)$;
- $\mathcal{D}_{\Omega}: \mathcal{B}(\partial\Omega) \to H^1(\Omega)$ is the Dirichlet 1-harmonic extension operator and $\tilde{\mathcal{D}}_{\Omega}: \mathcal{B}_0(\partial\Omega) \to H^1_{\langle 0 \rangle}(\Omega)$ is the Dirichlet harmonic extension operator (Definitions A.11 and A.12);
- $S_{\Omega} : \mathcal{B}'(\partial\Omega) \to H^1(\Omega)$ is the Neumann 1-harmonic extension operator and $\tilde{S}_{\Omega} : \mathcal{B}_0(\partial\Omega) \to H^1_{\langle 0 \rangle}(\Omega)$ is the Neumann harmonic extension operator (Definition A.7);
- \mathcal{A} : $\mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ and \mathcal{A} : $\mathcal{B}_0(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega)$ are the Poincaré-Steklov operator for the problems on Ω associated to $(-\Delta + 1)$ and $-\Delta$ respectively (Definition A.14).

If Ω is a two-sided extension domain:

- $\operatorname{Tr}_e: H^1(\Omega^c) \to \mathcal{B}(\partial \Omega)$ is the exterior trace operator;
- $\frac{\partial_e}{\partial u}: H^1_{\Delta}(\Omega^c) \to \mathcal{B}(\partial \Omega)$ is the weak exterior normal derivation operator;
- $[[\operatorname{Tr} \cdot]] := \operatorname{Tr}_i \operatorname{Tr}_e$ and $[[\frac{\partial}{\partial \nu}]] := \frac{\partial_i}{\partial \nu} \frac{\partial_e}{\partial \nu}$ are the jump in trace and in normal derivative across $\partial \Omega$ respectively (Definition 2.14);
- $\mathcal{D}_{\partial\Omega}$: $\mathcal{B}(\partial\Omega) \to H^1_{\partial\Omega}(\Omega)$ and $\tilde{\mathcal{D}}_{\partial\Omega}$: $\mathcal{B}(\partial\Omega) \to V_{\partial\Omega}(\mathbb{R}^n)$ are the double layer potential operators for the transmission problems across $\partial\Omega$ associated to $(-\Delta + 1)$ and $-\Delta$ respectively (Propositions B.3 and B.12);
- $S_{\partial\Omega}: \mathcal{B}'(\partial\Omega) \to H^1_{\partial\Omega}(\Omega)$ and $\tilde{S}_{\partial\Omega}: \mathcal{B}'_0(\partial\Omega) \to V_{\partial\Omega}(\mathbb{R}^n)$ are the single layer potential operators for the transmission problems across $\partial\Omega$ associated to $(-\Delta + 1)$ and $-\Delta$ respectively (Theorem B.1 and Proposition B.11);
- $\mathcal{K}_{\partial\Omega}$: $\mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ and $\tilde{\mathcal{K}}_{\partial\Omega}$: $\mathcal{B}_0(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega)$ are the Neumann-Poincaré operators for the transmission problems across $\partial\Omega$ associated to $(-\Delta + 1)$ and $-\Delta$ respectively (Definition B.23);
- $\mathcal{V}_{\partial\Omega} := \mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ and $\mathcal{V}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega)$ are the traces of the single layer potential operators for the transmission problems across $\partial\Omega$ associated to $(-\Delta+1)$ and $-\Delta$ respectively (Equations (B.48) and (B.49));
- $-\mathcal{W}_{\partial\Omega} := \mathcal{B}'(\partial\Omega) \to \mathcal{B}(\partial\Omega)$ and $-\tilde{\mathcal{W}}_{\partial\Omega} : \mathcal{B}'_0(\partial\Omega) \to \mathcal{B}(\partial\Omega)$ are the normal derivatives of the double layer potential operators for the transmission problems across $\partial\Omega$ associated to $(-\Delta+1)$ and $-\Delta$ respectively (Equations (B.48) and (B.49));
- $\mathcal{A}_{\partial\Omega,i}$: $\mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ and $\mathcal{A}_{\partial\Omega,e}$: $\mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ are respectively the interior and exterior Poincaré-Steklov operators for the transmission problems across $\partial\Omega$ associated to $(-\Delta + 1)$ (Theorem B.28).

1 Introduction

Whether it be in medical imaging for the detection of cancer tumors, in the study of metal alloys to detect impurities or in the military field to identify buried objects, imaging problems can arise in a wide range of situations. The idea of such a problem can be described as follows: a domain contains an inclusion characterized by a change in conductivity; we wish to identify the inclusion from data obtained on the boundary of the domain. Naturally, when solving such problems, modelling the boundary under study accounts for an essential step. In the classical theory, all boundaries are assumed to be regular, that is at least Lipschitz, and rather often once continuously differentiable. In that framework, imaging problems have been widely studied: in [61], a representation formula is proved for the solutions to the imaging problem (Problem (4.19) on page 29) and used to identify the inclusion from a single boundary measurement under some hypotheses (for example, when the inclusion is a disk). The book [1] synthesises diverse imaging results from single boundary measurements on Lipschitz boundaries, but also multiple boundary measurements using the Poincaré-Steklov (or Dirichlet-to-Neumann) operator to identify the inclusion. Note that imaging problems are closely connected to Calderón's problem (conductivity identification from boundary measurements), see for instance [19, 20], as well as inverse scattering problems (reconstruction of obstacles from scattered waves), see for instance [35] for the study of such problems on C^2 domains and [70, 95] on Lipschitz domains.

In the classical case, the imaging problem is solved introducing operators called the single and double layer potential operator [85, 91], which are defined as integrals over the boundary using the fundamental solution to the problem. Those operators link a Neumann or Dirichlet boundary condition respectively to a solution having that condition as jump in trace or in normal derivative respectively. Therefore, the layer potential operators are closely connected to the study of transmission problems (Problem (4.1) on page 21) – at this point, across Lipschitz boundaries; the link between the solution to the problem and the layer potential operators is known as Green's third identity [92]. Imaging considerations aside, transmission problems themselves have been widely studied in a Lipschitz framework. In [3] is proved the transmission problem on the whole Euclidean space can be approximated by truncated transmission problems, and that the associated solutions converge as the truncation domain grows. It is also possible look into the boundary values of the solution (determined from boundary jumps alone), which leads to the introduction of the Neumann-Poincaré operator [77, 91] as a singular integral over a the boundary. It is well known that operator can be symmetrized [80] and that its (real) spectrum is included in $\left[-\frac{1}{2},\frac{1}{2}\right]$ [44, 62]. The fundamental solution to the problem, used to define the layer potential operators can be understood from a probabilistic perspective [8], relying on the connection between the analytical concept of Dirichlet forms and the probabilistic concept of Feller semi-groups [31].

Although the Lipschitz model allow to set boundary conditions using definitions in the strong sense, real object often are far more irregular, and it is not uncommon for such shapes to be best modelled using fractals (*i.e.* self-similar or scale invariant geometries) [72, 73]. This consideration gives rise to new questions: how irregular can the domain be? how to define boundary data in that case? The notions of trace operators for irregular domains (see 94) for d-set boundaries (Definition 2.3), [53] for d-upper regular boundaries (Definition 2.4) with $d \in [n-2, n[$ and [14] for H^1 -extension domains (Definition 2.5), of weak normal derivative [67] and of Sobolev spaces of functions defined on irregular sets [26] allow to consider boundary value problems on a broad class of non-Lipschitz domains, including fractal domains, from a variational perspective. In [25, 26], scattering problem are studied in the case of an obstacle described by a fractal screen (*i.e.*, a relatively open fractal set in \mathbb{R}^{n-1}), and leads to the study of layer potential operators in that case. Though dealing with irregular shapes as well, our framework is different for our domain is open in \mathbb{R}^n , which prevents from using boundary condition in the strong sense. The Dirichlet and Neumann boundary conditions are then understood as elements of an abstract trace space and its dual respectively.

In this thesis, we aim at solving imaging problems set on H^1 -Sobolev extension domains. That class contains smooth domains, Lipschitz domains, fractal domains, but unlike the latter, extension domains are not defined using a specific boundary measure, relying on the framework from [14]. This allows to consider domains with varying boundary dimension, such as multi-fractal domains, or partly smooth and partly fractal domains. In addition, the results obtained do not depend on any boundary measure, including in the case of Lipschitz boundaries.

This manuscript is organised as follows: notations which will be used throughout this manuscript are listed on pages iii–vi. In Section 2, we introduce the class of extension domains on which our problems are set. We introduce the trace operator and the weak normal derivatives in our framework and characterize the trace operator as an isometry (Theorem 2.9). Moreover, we explain the probabilistic approach to solving boundary value problems. In Sections 3 to 5, we give an overview of results regarding boundary value and imaging problems on Lipschitz domains and introduce the results which we generalize to extension domains, while the new results are stated and proved in Appendix A to C. For that matter and although this thesis can be read linearly, we advise the reader to refer to the corresponding subsection in the appendix after having read a subsection in the body of the text (both subsections have the same title apart from Section 5: all of Appendix C refers to Subsection 5.2).

In Section 3 and Appendix A, we study bounded problems set on our class of extension domains with Dirichlet and Neumann boundary conditions for the operators $(-\Delta + 1)$ and $-\Delta$. We introduce Hilbert spaces on which those problems are well-posed and introduce the corresponding (1-)harmonic extension operators. We prove those operators are isometric (Theorems A.8)

and A.13). Then, we generalize the notion of Poincaré-Steklov operator in the case of extension domains as a linear and continuous operator, and prove it can be regarded as the Riesz isometry between the trace space and its dual space (Theorem A.17). Our isometry results are condensed in Diagram (3.7).

In Section 4 and Appendix B, we focus on the study of transmission problems across the boundary of an extension domain for $(-\Delta + 1)$ and $-\Delta$. In Subsection 4.1 and Appendix B.1, we identify spaces on which that transmission problem is well-posed and define the associated single and double layer potential operators as linear and continuous operators. Our method relies on a variational approach and is focused on the link those operators make between the boundary conditions of the problem and the solution. We prove those operators can be expressed as convolutions on the boundary with the fundamental solution, which generalize the classical formulae (Definition B.1 and Propositions B.6, B.11 and B.12). Finally, we express the normal derivative of the fundamental solution to the Laplacian, used to define the double layer potential in the classical case, in terms of the harmonic measure (Theorem 4.1). In Subsection 4.2 and Appendix B.2, we study the truncated transmission problem and define associated layer potential operators as linear and continuous operators. We prove the convergence of those operators to the layer potentials for the transmission problem on the whole space as the truncation domain grows (Theorems B.18 and B.22). In Subsection 4.4 and Appendix B.4, we use the layer potential operators in the framework of extension domains to generalize the notion of Poincaré-Steklov operator for transmission problems. We define two operators (one interior and one exterior) and characterize them in terms of solutions to the transmission problem. In Subsection 4.3 and Appendix B.3, we generalize the notion of Neumann-Poincaré operator using the boundary relations of the double layer potential operator. We prove its adjoint is connected to the single layer potential operator in the same way as it is in the Lipschitz case (Proposition B.24). We also define other operators associated to boundary values of the layer potentials and generalize the symmetrization properties of the Neumann-Poincaré operator and the notion of Calderón projectors (Theorem B.27). In Subsection 4.5, we use the boundary values of the layer potentials to define norms on the trace space and its dual, equivalent to the usual ones. We prove contracting properties of the Neumann-Poincaré operator (Theorem B.29) and that its real spectrum is included in $\left|-\frac{1}{2},\frac{1}{2}\right|$ (Theorem B.31). Finally, we estimate the parameters of the equivalence of the norms defined in this subsection. In Subsection 4.6 and Appendix B.6, we generalize the representation formula for the solutions to two-phased transmission problems in the case of extension domains, using the generalized layer potential and Neumann-Poincaré operators (Theorem B.34). We apply it to the identification of inclusions from a single boundary measurement in the monotonous case for extension domains and in the case of disks (inside an extension domain).

In Section 5 and Appendix C, we study the Hilbert transform and its connection to the layer potential operators in the complex plane. We extend

our definitions of layer potential and Neumann-Poincaré operators to complexvalued condition, and express the Hilbert transform in terms of single and double layer potential operators in the Lipschitz case (Theorem 5.2). We give two possible definitions of the Hilbert transform in \mathbb{R}^2 for extension domains. In an attempt to prove those definitions coincide, we prove any domain can be approximated by Lipschitz domains from both the inside and the outside (Theorem C.3), as well as the convergence of Dirichlet harmonic extensions in the context of converging domains (Theorem C.17).

2 Framework

In this section, we introduce the class of domains on which our study will be carried out: the class of one- and two-sided extension domains (Subsection 2.1). We define Dirichlet and Neumann boundary conditions for such domains by means of the trace operators and the weak normal derivative and prove the trace operator defines a partial isometry (Subsection 2.2). In Subsection 2.3, we describe the probabilistic approach to the study of boundary value problems.

2.1 Domain regularity

A general framework for the study of boundary value problems relies on the notion of Lipschitz domains [51, Definition 2.4.5].

Definition 2.1. (Lipschitz domain) A domain Ω of \mathbb{R}^n is said to be a Lipschitz domain if there exist constants L, a, r > 0 such that for all $x \in \partial \Omega$, there exist an orthogonal system of coordinates the origin of which is x = 0, a cylinder $K = B \times]-a, a[$ where B is an (n-1)-dimensional open ball of radius r > 0, and a L-Lipschitz continuous function $\varphi : B \rightarrow]-a, a[$ with $\varphi(0) = 0$ such that:

- (i) $\partial \Omega \cap K = \{(y, \varphi(y)) \mid y \in B\},\$
- (*ii*) $\Omega \cap K = \{(y, x_n) \mid y \in B, x_n > \varphi(y)\}.$

In other words, a Lipschitz domain is a domain with a boundary which is locally the graph of a Lipschitz continuous function (condition (i)), and which lies on one side of its boundary (condition (ii)). This class of domains contains smooth domains such as C^1 -domains, but also domains with a boundary which is not differentiable everywhere, as long as the singularity points form a σ null set. Polygonal shapes and pre-fractal domains are prime examples of Lipschitz domains. However, fractal objects such as the Von Koch snowflake (approximated by Figure 1) are nowhere differentiable, and therefore do no satisfy the Lipschitz property.

The fundamental difference between Lipschitz domains and fractal domains can be understood in terms of Hausdorff dimension of the boundary.



Figure 1: A fifth generation pre-fractal curve of the Von Koch snowflake. (Wikipedia)

Definition 2.2. (Hausdroff measures and dimension in \mathbb{R}^n) Let X be a Borel set of \mathbb{R}^n , and let s > 0. The s-dimensional Hausdorff measure of X is defined by:

$$H^{s}(X) := \lim_{r \to 0} \inf \left\{ \sum_{i \in \mathbb{N}} \operatorname{diam}(A_{i})^{s} \middle| \forall i \in \mathbb{N}, \left(A_{i} \subset \mathbb{R}^{n} \text{ is a Borel set,} \right. \\ \left. \operatorname{diam}(A_{i}) < r \right), \text{ and } X \subset \bigcup_{i \in \mathbb{N}} A_{i} \right\}.$$

The Hausdorff dimension of X is defined by:

$$\dim_H(X) := \inf \{ s > 0 \mid H^s(X) = 0 \}.$$

While Lipschitz domains have boundaries of Hausdorff dimension n-1, the Hausdorff dimension of fractal domains can be any number in]n-2, n[. In particular, it can take non-integer values, as in the case for the Von Koch snowflake which is of dimension $\frac{\ln 4}{\ln 3} \in]1, 2[$. One idea to widen the class of domains we consider could be to focus on all domains with a boundary of Hausdorff dimension in]n-2, n[.

We may now introduce a class of boundaries which includes fractal shapes such as the Von Koch snowflake: Ahlfors *d*-regular sets (or *d*-sets) [57, 58]. In the following definition, we assume $d \in [n - 2, n[$ as it corresponds to the boundaries which are under study; note that the notion of *d*-set can be defined for all $d \in [0, n]$.

Definition 2.3. (*d-set*) Let $d \in [n-2, n[$. A Borel set F of \mathbb{R}^n is said to be a *d-set if there exists a Borel measure* μ supported by F and constants $c_1, c_2 > 0$

such that:

$$\forall x \in F, \forall r \in]0,1], \quad c_1 r^d \le \mu(F \cap B_r(x)) \le c_2 r^d.$$

A measure μ satisfying the condition from Definition 2.3 is called '*d*-measure'. It was proved in [57, Proposition 1, p. 30] that, on a given *d*-set, all *d*-measures are equivalent: Definition 2.3 could be stated using the *d*-dimensional Hausdorff measure as μ . In particular, the existence of a *d*-measure supported by the boundary of a domain Ω ensures the Hausdorff dimension of $\partial\Omega$ is *d*. Naturally, the boundary of any Lipschitz domain is an (n-1)-set. The Von Koch snowflake is an example of a $\frac{\ln 4}{\ln 3}$ -set.

It goes without saying the class of d-sets is much more general than the class of Lipschitz boundaries for the study of boundary value problems, provided we manage to define boundary condition for such shapes. To go even further, as we explained before, all d-sets have a fixed boundary dimension, and we can be interested in getting rid of that condition. That leads to the introduction of d-upper regular boundaries, studied notably in [53].

Definition 2.4. (*d-upper regular set*) Let $d \in [n-2, n[$. A Borel set F of \mathbb{R}^n is said to be *d-upper regular if there exists a Borel measure* μ supported by F and a constant c > 0 such that:

$$\forall x \in F, \ \forall r \in]0,1], \quad \mu(F \cap B_r(x)) \le c \, r^d.$$

In the case of a *d*-upper regular boundary, the Hausdorff dimension can vary, as long as it is being bounded below by *d*. For that matter, the class of *d*upper regular set greatly expands that of *d*-sets. Nonetheless, results obtained for domains with such boundaries will depend on the boundary measure μ *a priori*. It will be pointed out in Subsection 2.2 it is possible to define those boundary conditions for an ever wider class of domains, which is the class of H^1 -extension domains [50].

Definition 2.5. (H^1 -extension domain) A domain Ω of \mathbb{R}^n is said to be an H^1 -extension domain if there exists a bounded linear extension operator $E: H^1(\Omega) \to H^1(\mathbb{R}^n)$, that is:

- (i) $\forall u \in H^1(\Omega), \ \mathrm{E}u|_{\Omega} = u,$
- (*ii*) $\forall u \in H^1(\Omega), \| \mathbb{E}u \|_{H^1(\mathbb{R}^n)} \le C \| u \|_{H^1(\Omega)},$

where C > 0 depends only on Ω (and n).

The class of H^1 -extension domains is very broad: it contains Lipschitz domains [22, 87], *d*-sets such as the Von Koch snowflake, and even domains with varying boundary dimension (*e.g.*, a Von Koch snowflake on one side, and a Lipschitz boundary on the other side). Also, no boundary measure is specified. A characterization of H^1 -extension domains is given in [50]: **Theorem 2.6.** A domain Ω of \mathbb{R}^n is an H^1 -extension domain if and only if it is an n-set (Definition 2.3 for d = n), and it holds $H^1(\Omega) = C_2^1(\Omega)$ in the sense of equivalent norms, where $C_2^1(\Omega)$ is the space of sharp maximal functions defined by:

$$C_2^1(\Omega) := \left\{ u \in L^2(\Omega) \ \middle| \\ \left(u_{2,\Omega}^{\#} : x \mapsto \sup_{r>0} \frac{1}{r} \inf_{c \in \mathbb{R}} \frac{1}{\lambda^{(n)}(B_r(x))} \int_{\Omega \cap B_r(x)} |u - c| \, \mathrm{d}y \right) \in L^2(\Omega) \right\},$$

endowed with the norm $||u||_{C_2^1(\Omega)} := ||u||_{L^2(\Omega)} + ||u_{2,\Omega}^{\#}||_{L^2(\Omega)}$.

The *n*-set condition ensures extension domains do not possess an outward cusp, *i.e.* they do not become infinitely thin. The domain in Figure 2 is an example of a domain which is not an extension domain. Note that this domain also has an inward cusp, therefore its complement is not an extension domain either.



Figure 2: A domain with a cusp.

We introduce the class of domains on which our study will be carried out.

Definition 2.7. (One- and two-sided extension domain) Let Ω be a domain of \mathbb{R}^n . If Ω is bounded and an H^1 -extension domain, then it is a one-sided extension domain.

In addition, if its complement domain $\Omega^c := \mathbb{R}^n \setminus \overline{\Omega}$ is an H^1 -extension domain, and the Hausdorff dimension of $\partial \Omega$ is included in [n-1, n], then Ω is a two-sided extension domain.

Considering two-sided extension domains will allow to define boundary conditions from both the inside and the outside of the domain, which will be necessary as we consider transmission problems in Subsection 4. The condition on the dimension of the boundary in Definition 2.7 is to be understood locally, as the dimension can vary along $\partial\Omega$, and ensures Ω is 'hermetically sealed' from Ω^c .

2.2 Boundary value problems on extension domains

The study of transmission problems requires the definition of two types of boundary conditions: Dirichlet boundary conditions and Neumann boundary conditions.

The Dirichlet boundary condition is defined using the trace operator. On Lipschitz boundaries, that operator was defined and studied by Marschall [74], shortly after generalized to *d*-set boundaries by Wallin [94] in 1991, and later on generalized to *d*-upper regular boundaries [53]. Those approaches rely on the existence of a boundary measure (Lebesgue's measure in the former case, a *d*-measure in the latter) to define that operator. In 2009, Biegert [14] uses a different approach based on the H^1 -extension property to define the trace operator on H^1 -extension domains. Here, we rely on the fact that this definition of the trace operator does not depend on the way an element of $H^1(\Omega)$ is extended to $H^1(\mathbb{R}^n)$ [14, Theorem 6.1] to give an equivalent definition, similar to that of Wallin on *d*-sets [94, Theorem 1] (see also [39, 53]).

Definition 2.8. (Interior trace operator) Let Ω be a one-sided extension domain of \mathbb{R}^n . The interior trace operator $\operatorname{Tr}_i : H^1(\Omega) \to \operatorname{Tr}_i(H^1(\Omega))$ is defined quasi-everywhere for $x \in \partial \Omega$ by:

$$\operatorname{Tr}_{i} u(x) = \lim_{r \to 0^{+}} \frac{1}{\lambda^{(n)}(\Omega \cap B_{r}(x))} \int_{\Omega \cap B_{r}(x)} u(y) \, \mathrm{d}y.$$
(2.1)

The 'quasi-everywhere' definition is understood in terms of the capacity [14, Definition 5.1]:

$$\begin{split} \mathrm{Cap}(U) &:= \inf \bigg\{ \; \|u\|_{H^1(\Omega)}^2 \; \Big| \; u \in \overline{C_0^0(\overline{\Omega}) \cap H^1(\Omega)}^{\|\cdot\|_{H^1(\Omega)}} \\ & \text{ and } \; u \geq 1 \; \text{on a neighborhood of } U \bigg\}, \end{split}$$

where U is a subset of $\overline{\Omega}$ and the neighbourhood is in the sense of the trace topology of \mathbb{R}^n on Ω . The use of that capacity allows to circumvent the absence of a specific boundary measure in the definition of H^1 -extension domains (Definition 2.5). The following theorem synthesizes important properties of the interior trace operator.

Theorem 2.9. (*Trace theorem*) Let Ω be a one-sided extension domain of \mathbb{R}^n . Then:

1. the image of the trace operator $\mathcal{B}(\partial\Omega) := \operatorname{Tr}_i(H^1(\Omega))$ endowed with the norm

$$||f||_{\mathcal{B}(\partial\Omega)} := \min\{||v||_{H^1(\Omega)} \mid f = \operatorname{Tr}_i v\},$$
(2.2)

is a Hilbert space.

- 2. Ker $\operatorname{Tr}_i = H_0^1(\Omega) = \overline{C_0^{\infty}(\Omega)}^{\|\cdot\|_{H^1(\Omega)}}$ is the closure in $H^1(\Omega)$ of the set of indefinitely differentiable functions with compact support in Ω .
- 3. the trace operator $\operatorname{Tr}_i : H^1(\Omega) \to \mathcal{B}(\partial\Omega)$ is a partial isometry with operator norm equal to 1.
- 4. $\operatorname{Tr}_i: V_1(\Omega) \to \mathcal{B}(\partial \Omega)$ defines an isometry, where

$$V_1(\Omega) = \left\{ v \in H^1(\Omega) \mid (-\Delta + 1)v = 0 \text{ weakly on } \Omega \right\}$$
(2.3)

is the space of (weakly) 1-harmonic functions, endowed with the standard norm $\|\cdot\|_{H^1(\Omega)}$. It holds:

$$V_1(\Omega) = \operatorname{Ker}(\operatorname{Tr}_i)^{\perp} = \left(H_0^1(\Omega)\right)^{\perp}.$$
(2.4)

Proof. Point 1 follows partially from the definition, see also [52, Theorem 1],[53, Theorem 5.1], [39, Theorem 2], where the norm was defined as:

$$||f||_{\operatorname{Tr}_i} = \inf\{||v||_{H^1(\Omega)} \mid f = \operatorname{Tr}_i v\}.$$

Here, we specify that the infimum is reached. To prove that, it is sufficient to use the continuity of the trace operator from $H^1(\Omega)$ to its image and Stampacchia's theorem [18, Theorem V.6], ensuring the existence of $u \in H^1_f(\Omega)$, where

$$H^1_f(\Omega) := \left\{ v \in H^1(\Omega) \mid \operatorname{Tr}_i v = f \text{ on } \partial\Omega \right\},$$
(2.5)

the unique minimizer of $\|\cdot\|_{H^1(\Omega)}$ on $H^1_f(\Omega)$. It is equivalent to the weak well-posedness on $H^1(\Omega)$ of the non-homogeneous Dirichlet boundary value problem for $(-\Delta + 1)$:

$$\begin{cases} -\Delta u + u = 0 \quad \text{on } \Omega, \\ \operatorname{Tr}_{i} u = f. \end{cases}$$
(2.6)

The weak solution to Problem (2.6) is then understood in the sense of the variational formulation on $H^1_f(\Omega)$:

$$\forall v \in H^1_f(\Omega), \quad \langle u, v \rangle_{H^1(\Omega)} = \|u\|^2_{H^1(\Omega)}, \tag{2.7}$$

which, in particular, implies:

$$\forall v \in H^1_f(\Omega), \quad \Big(\|v\|_{H^1(\Omega)} = \|f\|_{\mathcal{B}(\partial\Omega)} \quad \Longleftrightarrow \quad u = v \Big). \tag{2.8}$$

To prove Point 2, we use the following characterization from [47, Corollary 2.3.1 and Example 2.3.1]:

$$H_0^1(\mathbb{R}^n \setminus \partial \Omega) = \left\{ v \in H^1(\mathbb{R}^n) \mid \tilde{v} = 0 \text{ q.e. on } \partial \Omega \right\}.$$

 $u \in H^1(\Omega)$ can be extended to $H^1_0(\mathbb{R}^n \setminus \partial \Omega)$ if and only if $u \in H^1_0(\Omega)$, and it can be extended to the space on the right hand side of the equality if and only if $\operatorname{Tr}_i u = 0$. Points 3 and 4 are corollaries of the previous points, of Stampacchia's theorem in particular. Let us prove (2.4). For $u \in H^1(\Omega)$, by Green's formula (see Definition 2.12 below), the following equivalences hold:

$$u \in V_1(\Omega) \iff -\Delta u + u = 0 \text{ weakly on } \Omega$$
$$\iff \forall v \in \operatorname{Ker}(\operatorname{Tr}_i), \quad \langle u, v \rangle_{H^1(\Omega)} = 0$$
$$\iff u \in \operatorname{Ker}(\operatorname{Tr}_i)^{\perp}.$$

Since $\operatorname{Tr}_i : \operatorname{Ker}(\operatorname{Tr}_i)^{\perp} \to \mathcal{B}(\partial\Omega)$ is bijection, by (2.8), it is an isometry. Therefore, since $\operatorname{Ker}(\operatorname{Tr}_i)^{\perp} \subset H^1(\Omega)$ and it is a closed Hilbert space endowed with $\|\cdot\|_{H^1}$ (as an orthogonal complement), the trace operator $\operatorname{Tr}_i : H^1(\Omega) \to \mathcal{B}(\partial\Omega)$ is a partial isometry of operator norm equal to 1.

As it was mentioned before, in the case of a two-sided extension domain, we also define an exterior trace operator in a similar way since Ω^c is an extension domain.

Definition 2.10. (Exterior trace operator) Let Ω be a two-sided extension domain of \mathbb{R}^n . The exterior trace operator $\operatorname{Tr}_e : H^1(\Omega^c) \to \mathcal{B}(\partial\Omega)$ is defined quasi-everywhere for $x \in \partial\Omega$ by:

$$\operatorname{Tr}_{e} u(x) = \lim_{r \to 0^{+}} \frac{1}{\lambda^{(n)}(\Omega^{c} \cap B_{r}(x))} \int_{\Omega^{c} \cap B_{r}(x)} u(y) \, \mathrm{d}y.$$
(2.9)

The trace operators allow to generalize the notion of 'value at the boundary' and define Dirichlet-type boundary conditions. The idea of the pointwise Definitions 2.8 and 2.10 is to assign to the trace of u at a point $x \in \partial \Omega$ the average of the values taken by u infinitely close to x. A similar result to Theorem 2.9 can be obtained using the exterior trace operator instead of the interior one. In particular, their ranges coincide, in the sense of equivalent norms.

Proposition 2.11. Let Ω be a two-sided extension domain of \mathbb{R}^n . Then, $\mathcal{B}(\partial \Omega) = \text{Tr}_e(H^1(\Omega^c))$ and the norm defined by:

$$||f||_{\mathcal{B}(\partial\Omega)} := \min\{||v||_{H^1(\Omega^c)} \mid f = \operatorname{Tr}_e v\},$$
(2.10)

defines a norm on $\mathcal{B}(\partial\Omega)$ equivalent to $\|\cdot\|_{\mathcal{B}(\partial\Omega)}$.

Proof. We prove the infimum is reached in the same way as in Theorem 2.9. By definition of an H^1 -extension domain, their norms depend on Ω alone, which also defines its complement. Equivalently, an element $u \in H^1(\mathbb{R}^n)$ can be viewed as an extension from either Ω or Ω^c , with coinciding interior and exterior traces on $\partial\Omega$. The Neumann boundary condition is defined using the normal derivation operator, defined in the classical sense by $\nu \cdot \nabla$ at a boundary point, where ν is the outward normal unit vector. Since a Lipschitz boundary is differentiable almost everywhere, that definition of the normal derivative can be used for such boundaries almost everywhere. However, as explained in Subsection 2.1, the normal vector exist for no $x \in \partial \Omega$ a priori when working in the general class of extension domains (or even *d*-sets), since the boundary may be nowhere differentiable (as in the case of the Von Koch snowflake, or any *d*-set for d > n-1). Therefore, in the case of extension domains, we cannot use the usual definition of the normal derivative as it is done in the regular case. Nonetheless, we may use Green's formula to define a weak normal derivative [67] as a linear continuous functional on the image of the trace, that is as an element of $\mathcal{B}'(\partial \Omega)$.

Definition 2.12. (Weak interior normal derivative) Let Ω be a one-sided extension domain of \mathbb{R}^n . Let $u \in H^1_{\Delta}(\Omega)$, where that space is defined by (also on pages iii-vi):

$$H^1_{\Delta}(\Omega) := \left\{ v \in H^1(\Omega) \mid \Delta v \in L^2(\Omega) \right\}.$$
(2.11)

The weak interior normal derivative of u is the element $\psi \in \mathcal{B}'(\partial \Omega)$ such that for all $v \in H^1(\Omega)$, it holds:

$$\langle \psi, \operatorname{Tr}_{i} v \rangle_{\mathcal{B}', \mathcal{B}} = \int_{\Omega} (\Delta u) v \, \mathrm{d}x + \int_{\Omega} \nabla u \cdot \nabla v \, \mathrm{d}x,$$
 (2.12)

and we denote $\frac{\partial_i u}{\partial \nu} := \psi$.

The space $H^1_{\Delta}(\Omega)$ is a Hilbert space endowed with the norm [36, Proposition 1.1]:

$$\|\cdot\|_{H^{1}_{\Delta}(\Omega)} := \sqrt{\|\cdot\|^{2}_{H^{1}(\Omega)} + \|\Delta\cdot\|^{2}_{L^{2}(\Omega)}}.$$
(2.13)

Note that, in addition to the definition of the weak normal derivative, Definition 2.12 states that Green's formula holds for all $v \in H^1(\Omega)$, *i.e.* that the left-hand side in Equation (2.12) only depends on $\operatorname{Tr}_i v$ rather than v itself [67].

Once again, in the case of a two-sided extension domain, we may also define a weak exterior normal derivative in a similar way.

Definition 2.13. (Weak exterior normal derivative) Let Ω be a two-sided extension domain of \mathbb{R}^n . Let $u \in H^1_{\Delta}(\Omega^c)$. The weak exterior normal derivative of u is the element $\varphi \in \mathcal{B}'(\partial\Omega)$ such that for all $v \in H^1(\Omega^c)$, it holds:

$$\langle \varphi, \operatorname{Tr}_{e} v \rangle_{\mathcal{B}', \mathcal{B}} = -\int_{\Omega^{c}} (\Delta u) v \, \mathrm{d}x - \int_{\Omega^{c}} \nabla u \cdot \nabla v \, \mathrm{d}x,$$
 (2.14)

and we denote $\frac{\partial_e u}{\partial \nu} := \varphi$.

In the case of a Lipschitz boundary, the weak normal derivatives coincide with those defined in the strong sense. Note the sign convention used in Definition 2.13, which corresponds to using the outward normal vector for the definition of the exterior normal derivative in the regular case. That choice makes the definition of the boundary jump values – which will be used as boundary conditions for transmission problems in Subsection 4 – more intuitive.

Definition 2.14. (Boundary jump values) Let Ω be a two-sided extension domain of \mathbb{R}^n , and define:

$$H^1_{\partial\Omega}(\mathbb{R}^n) := \left\{ u \in L^2(\mathbb{R}^n) \mid \nabla u|_{\Omega \cup \Omega^c} \in L^2(\Omega \cup \Omega^c)^n \right\}.$$
(2.15)

The jump in trace of $u \in H^1_{\partial\Omega}(\mathbb{R}^n)$ across $\partial\Omega$ is defined by:

$$\llbracket \operatorname{Tr} u \rrbracket := \operatorname{Tr}_i u - \operatorname{Tr}_e u.$$

In addition, if $\Delta u|_{\Omega \cup \Omega^c} \in L^2(\Omega \cup \Omega^c)$, then the jump in normal derivative of u across $\partial \Omega$ is defined by:

$$\left[\frac{\partial u}{\partial \nu}\right] := \frac{\partial_i u}{\partial \nu} - \frac{\partial_e u}{\partial \nu}.$$

Remark 2.15. In Definition 2.14, the gradient ∇u is defined in the weak sense as a distribution on \mathbb{R}^n . The statement ' $\nabla u|_{\Omega \cup \Omega^c} \in L^2(\Omega \cup \Omega^c)^n$ ' is to be understood in the following sense:

$$\exists \psi \in L^2(\Omega \cup \Omega^c), \ \forall \varphi \in C_0^\infty(\Omega \cup \Omega^c), \quad \langle \nabla u, \varphi \rangle_{(C_0^\infty)', C_0^\infty(\Omega \cup \Omega^c)} = \int_{\Omega \cup \Omega^c} \psi \varphi \, \mathrm{d}x,$$

where $C_0^{\infty}(\Omega \cup \Omega^c)$ is defined on pages *iii*-vi and $(C_0^{\infty}(\Omega \cup \Omega^c))'$ the space of distributions on $\Omega \cup \Omega^c$. Since all elements of $C_0^{\infty}(\Omega \cup \Omega^c)$ vanish at the boundary, that space (and consequently $L^2(\Omega \cup \Omega^c)$) 'ignores' the behavior of the distribution ∇u on the boundary.

In what follows, such statements involving weak derivatives and L^2 spaces are always to be understood in that sense.

The space $H^1_{\partial\Omega}(\mathbb{R}^n)$ defined by (2.15) is a Hilbert space endowed with the norm:

$$\|\cdot\|_{H^1_{\partial\Omega}(\mathbb{R}^n)} := \sqrt{\|\cdot\|_{L^2(\mathbb{R}^n)}^2} + \|\nabla\cdot\|_{L^2(\Omega\cup\Omega^c)^n}^2.$$

It is slightly different from the space $H^1(\Omega \cup \Omega^c)$ which contains distributions only defined on $\Omega \cup \Omega^c - i.e.$ linear continuous forms with domain $C_0^{\infty}(\Omega \cup \Omega^c)$, the elements of which vanish near the boundary. $H^1_{\partial\Omega}(\mathbb{R}^n)$ on the other hand can be seen as a space of distributions on \mathbb{R}^n , which can be applied to functions in $C_0^{\infty}(\mathbb{R}^n)$ which contains functions non-null near the boundary. In that sense, setting the transmission problem on $H^1_{\partial\Omega}(\mathbb{R}^n)$ instead of $H^1(\Omega \cup \Omega^c)$ will give more regularity to the solution, for it will be identified on the whole space \mathbb{R}^n , and not only on $\Omega \cup \Omega^c$. We also insist on the fact that the space $H^1(\mathbb{R}^n)$ is not suitable for the study of transmission problems, for being in that space is not compatible with having a non homogeneous jump in trace.

2.3 Probabilistic perspective

2.3.1 Variational-probabilistic correspondence

Although the most intuitive approach to a boundary value problem is an analytical approach, such problems are also widely studied from a probabilistic perspective. The link between those approaches lies in the connections between the concepts of Dirichlet forms and Markov processes. Let us start with reminding the definition of Dirichlet forms [31, Definition 1.1.2, Theorem 1.1.3].

Definition 2.16. (Dirichlet form) Let (E, m) be a σ -finite measurable space. A pair $(\mathcal{E}, dom(\mathcal{E}))$ is said to be a Dirichlet form on $L^2(E, m)$ if

- (i) $(\mathcal{E}, dom(\mathcal{E}))$ is a symmetric form on $L^2(E, m)$: $dom(\mathcal{E})$ is a dense linear subspace of $L^2(E, m)$ and \mathcal{E} is a non-negative definite symmetric bilinear form defined on $dom(\mathcal{E}) \times dom(\mathcal{E})$;
- (ii) $(\mathcal{E}, dom(\mathcal{E}))$ is closed: $dom(\mathcal{E})$ is complete for the inner product

$$\mathcal{E}_1(u,v) := \langle u, v \rangle_{L^2(E,m)} + \mathcal{E}(u,v), \qquad u, v \in dom(\mathcal{E});$$

(iii) $(\mathcal{E}, dom(\mathcal{E}))$ is Markovian:

$$\forall u \in dom(\mathcal{E}), \quad v := (0 \lor u) \land 1 \in dom(\mathcal{E}) \quad and \quad \mathcal{E}(v, v) \le \mathcal{E}(u, u).$$

If Ω is a domain of \mathbb{R}^n , a very standard example of Dirichlet form which will be of interest here is the form $(\mathcal{E}^{\Delta}, H^1(\Omega))$, where \mathcal{E}^{Δ} is the bilinear form associated to the Laplacian $-\Delta$ on Ω :

$$\mathcal{E}^{\Delta}: (u,v) \in H^1(\Omega)^2 \longmapsto \int_{\Omega} \nabla u \cdot \nabla v \, \mathrm{d}x.$$

Let us now recall the definition of Markov processes [31, Definiton A.1.8] (see also [16, Chapter I, Definition 1.1] for an alternative definition).

Definition 2.17. (Markov process) Let $(E, \mathcal{B}(E))$ be a measurable space (endowed with its Borel σ -field). Consider $E_{\partial} := E \cup \{\partial\}$, where ∂ is an isolated point, endowed with the σ -field:

$$\mathcal{B}(E_{\partial}) := \mathcal{B}(E) \cup \{A \cup \{\partial\}, A \in \mathcal{B}(E)\}.$$

 $X = (S, \mathcal{F}, (X_t)_{t \in [0,\infty]}, (\mathbb{P}_x)_{x \in E_{\partial}})$ is said to be a Markov process on $(E, \mathcal{B}(E))$ if:

(i) for all $x \in E_{\partial}$, $(S, \mathcal{F}, (X_t)_{t \in [0,\infty]}, \mathbb{P}_x)$ is a stochastic process with time parameter set $[0, \infty]$ and state space $(E_{\partial}, \mathcal{B}(E_{\partial}))$, such that

$$\forall \omega \in S, \quad X_{\infty}(\omega) = \partial;$$

- (ii) for all $t \ge 0$ and $A \in \mathcal{B}(E_{\partial})$, the function $x \in E_{\partial} \mapsto \mathbb{P}_{x}(X_{t} \in A)$ is $\mathcal{B}(E_{\partial})$ -measurable;
- (iii) (Markov property) there exists a filtration $(\mathcal{F}_t)_{t \in [0,\infty]}$ such that:

$$\forall x \in E_{\partial}, \ \forall s, t \ge 0, \ \forall A \in \mathcal{B}(E_{\partial}), \\ \mathbb{P}_{x}(X_{s+t} \in A \mid \mathcal{F}_{t}) = \mathbb{P}_{X_{t}}(X_{s} \in A) \quad \mathbb{P}_{x}\text{-}a.s.$$

- (iv) $\forall t \ge 0$, $\mathbb{P}_{\partial}(X_t = \partial) = 1$;
- (v) (Normality) $\forall x \in E_{\partial}, \quad \mathbb{P}_x(X_0 = x) = 1.$

One particular topic of interest for a Markov process X is its transition function $(P_t)_{t\geq 0}$:

$$\forall t \ge 0, \quad P_t : (x, A) \in E \times \mathcal{B}(E) \longmapsto \mathbb{P}_x(X_t \in A).$$

The transition function $(P_t)_{t\geq 0}$ of a Markov process X is such that [31, Theorem A.1.9]:

1. for all $s, t \ge 0$ and all $f: E \to \mathbb{R}$ bounded measurable, it holds:

$$P_s P_t f = P_{s+t} f,$$

where

$$P_t f: x \in E \longmapsto \int_E f(y) P_t(x, \mathrm{d}y);$$

2. for all $x \in E$, $P_0(x, \cdot)$ is the Dirac measure concentrated at x.

Dirichlet forms are analytical objects while Markov processes (and their transition functions) are probabilistic. However, under some assumptions, a correspondence can be made between both concepts. Assume the transition function $(P_t)_{t\geq 0}$ of a Markov process on a measurable space (E, m) with m σ -finite is such that:

(i) $(P_t)_{t\geq 0}$ is *m*-symmetric, *i.e.* for all $t\geq 0$ and all $f,g:E\to \mathbb{R}_+$ measurable, it holds:

$$\int_{E} (P_t f) g \, \mathrm{d}m = \int_{E} f(P_t g) \, \mathrm{d}m;$$

(ii) for all $x \in E$ and $f : E \to \mathbb{R}$ bounded continuous on E, it holds:

$$\lim_{t \to 0} P_t f(x) = f(x),$$

then [31, Lemma 1.1.14], if $(T_t)_{t\geq 0}$ denotes the family of symmetric linear operators on $L^2(E, m)$ defined by:

$$\forall f : E \to \mathbb{R} \text{ bounded measurable}, \quad T_t f(x) := \int_E f(y) P_t(x, \mathrm{d}y), \quad (2.16)$$

 $(T_t)_{t\geq 0}$ is a strongly continuous contraction semi-group on $L^2(E,m)$ (*i.e.*, a Feller semi-group), that is for $s,t\geq 0$ and $f\in L^2(E,m)$:

$$T_s \circ T_t = T_{s+t}, \quad ||T_t f||_{L^2(E,m)} \le ||f||_{L^2(E,m)} \quad \text{and} \quad \lim_{t \to 0} ||T_t f - f||_{L^2(E,m)} = 0.$$

That semi-group can be used to define a Dirichlet form $(\mathcal{E}, dom(\mathcal{E}))$ on $L^2(E, m)$ [31, Theorem 1.1.3]:

$$dom(\mathcal{E}) := \left\{ f \in L^2(E,m) \mid \lim_{t \to 0} \frac{1}{t} \langle f - T_t f, f \rangle_{L^2(E,m)} < +\infty \right\},$$
$$\forall f, g \in dom(\mathcal{E}), \quad \mathcal{E}(f,g) := \lim_{t \to 0} \frac{1}{t} \langle f - T_t f, g \rangle_{L^2(E,m)}.$$

Conversely, a Dirichlet form $(\mathcal{E}, dom(\mathcal{E}))$ on $L^2(E, m)$ can be used to define a strongly continuous contraction semi-group [31, Theorem 1.1.3]:

$$T_t: f \in L^2(E,m) \longmapsto \lim_{\alpha \to \infty} e^{-t\alpha} \sum_{k=0}^{+\infty} \frac{(t\alpha)^k}{k!} (\alpha G_\alpha)^k f,$$

where G_{α} , $\alpha \geq 0$, is defined using the Riesz theorem by:

$$\forall f \in L^2(E,m), \quad G_{\alpha}f \in dom(\mathcal{E}) \quad \text{and} \quad \big(\forall v \in dom(\mathcal{E}), \\ \langle f, v \rangle_{L^2(E,m)} = \langle G_{\alpha}f, v \rangle_{L^2(E,m)} + \alpha \mathcal{E}(G_{\alpha}f, v) \big).$$

If $(T_t)_{t\geq 0}$ is associated to a Markov process on (E, m) in the sense of (2.16), then its transition function is *m*-symmetric.

2.3.2 Probabilistic incorporation of boundary conditions

In Subsection 2.3.1, we highlighted the fundamental connection between a variational approach and a probabilistic approach to solve elliptic problems. Here, we describe the probabilistic method to incorporate different types of boundary conditions.

Stopped and killed processes. Dirichlet boundary conditions are accounted for by considering processes (described in Subsection 2.3.1) stopped upon reaching the boundary. The most standard example is that of harmonic functions expressed in terms of Brownian diffusions stopped as they exit the domain, see for instance [40, Part 2, Sections III.8, IV.3 and VIII.13] where the link between harmonic functions and martingales through Îto's formula is highlighted. That link is thoroughly used in [8, Chapter II] to establish a number of properties of harmonic functions, including their reconstruction from Dirichlet boundary data using Brownian diffusions:

$$\forall x \in \Omega, \quad u(x) = \mathbb{E}_x(f(B_\tau)),$$

where Ω is a domain of \mathbb{R}^n , u is harmonic (in the strong sense of a C^2 function) on Ω and coincides with $f \in C^0(\partial\Omega)$ on $\partial\Omega$, B is an *n*-dimensional Brownian motion and τ is its exit time from Ω . This result can be generalized to harmonic functions in the variational sense [6], as it will be pointed out in Subsection 3.1.

Similarly to the variational approach, setting a homogeneous boundary condition can be used to focus on the influence of the source term, which leads to the introduction of killed processes, see for instance [32, Chapter 2]. In that case, we adjoin one cemetery point to the domain Ω , which symbolizes the boundary, and consider the process is trapped at that point when it first exits the domain. In the case of the Brownian motion, if the source term is given by $s \in L^{\infty}(\Omega)$ and setting s = 0 for the cemetery point, then the heat diffusion at time $t \geq 0$ is given by [32, Section 2.1]:

$$\forall x \in \Omega, \quad u(t, x) = \mathbb{E}_x(s(B_t)).$$

One important remark is that the spaces in which the Dirichlet boundary condition and the source term are chosen here are not the intuitive spaces we would use with a variational approach, that is $\mathcal{B}(\partial\Omega)$ and $L^2(\Omega)$ respectively. The probabilistic approach to boundary value problems is not only an alternative to the variational approach, those perspectives are also complementary in that sense. We also refer to [9, 31, 41, 42, 47] for the study of stopped and killed processes.

Reflected processes. Neumann boundary conditions are accounted for by considering processes reflected upon hitting the boundary. The most common approach, see for instance [10] for normally reflected processes on Lipschitz boundaries, is to introduce the semi-group associated to the reflected process, in the spirit of Subsection 2.3.1, and deduce the properties of the diffusion from its study. We also refer to [28, 29, 30, 55, 86] for the study of (normally) reflected processes. In [21] is defined a Brownian motion which is obliquely reflected on the boundary, which accounts for a non-homogeneous Neumann boundary condition. While the papers mentioned here mostly introduce a semi-group which incorporates the boundary condition, it is also possible to define a proper probabilistic normal derivative on non-smooth domains as it is done in [49], which coincides – up to a multiplicative constant – with Definition 2.12.

Absorbed processes. Although it is not the main focus of this study, we say a few words about absorbing boundary conditions of order 1 and 2, that is Robin and Wentzell-type boundary conditions. Robin (or third type or mixed) boundary conditions are expressed in the following way:

$$\left(\frac{\partial u}{\partial \nu} + \alpha \operatorname{Tr} u\right)\Big|_{\partial\Omega} = h,$$

where h is given and α is a complex coefficient. This boundary condition corresponds to a solution which is partly absorbed and partly reflected, depending

on the coefficient α . Problems with such a boundary condition were treated using a probabilistic approach in [82] in 1990 on C^3 boundaries (regularity needed for technical reasons in the proof). That approach relies on the study of the semi-group connected to the solution, similarly to (2.16), and allows to solve the problem with lighter hypotheses on the spectrum of the operator than in the variational approach. Recently, a similar approach allowed to treat problems with a Robin boundary conditions on a *d*-set boundary in [24].

Another type of absorbing boundary conditions is the Wentzell boundary condition, first introduced in [90] in 1959, which is expressed as:

$$\left(\frac{\partial u}{\partial \nu} + \alpha \operatorname{Tr} u + \beta \Delta_{\partial \Omega} u\right)\Big|_{\partial \Omega} = h_{2}$$

where β is another complex coefficient, and $\Delta_{\partial\Omega}$ denotes the tangential Laplacian along $\partial\Omega$. In addition to the absorption-reflection effect of the Robin boundary condition, the Wentzell boundary condition also contains a term $(\beta\Delta_{\partial\Omega}u)$ which refers to an adhesion of the process to the boundary: the process enters the boundary at a certain time and exits at another. A probabilistic study of this problem is carried out in [93], by means of the Dirichlet form associated to the diffusing process, and in [68] on a Von Koch snowflake, using the convergence of the associated Dirichlet forms along the pre-fractal sequence to define a Wentzell boundary condition without defining the tangential Laplacian *per say*. We also refer to [15, 46, 56] for the study of problems with Wentzell boundary conditions.

3 Bounded problems

Let us begin our study of problems set on extension domains by considering bounded problems, *i.e.* problems set on a bounded Ω . In this section, we will consider in turn a Dirichlet and a Neumann boundary condition, that is:

$$\begin{cases} (-\Delta + k)u = 0 & \text{on } \Omega \\ \operatorname{Tr}_{i} u = f \end{cases}$$
(3.1)

on the one hand, and:

$$\begin{cases} (-\Delta + k)u = 0 & \text{on } \Omega\\ \frac{\partial_i u}{\partial \nu} = g \end{cases}$$
(3.2)

on the other hand, with $k \in \{0, 1\}$, and f, g given boundary conditions.

In Subsection 3.1 and Appendix A.1, we establish the well-posedness of the problems above when Ω is a one-sided extension domain, and prove the associated (1-) harmonic extension operators can be viewed as isometries. We prove both the Dirichlet and Neumann problem have the same range, and in Subsection 3.2 and Appendix A.2, we highlight that connection through the definition of the Poincaré-Steklov operator. We prove that operator defines an isometry as well, and explicit the connections between the latter and the (1-)harmonic extension operators.

3.1 Harmonic extension operators

The first question one can ask is whether Problems (3.1) and (3.2) are wellposed for k = 1 and k = 0, and on what spaces. The case k = 1 is simpler since 1 is not an eigenvalue for either the Dirichlet Laplacian or the Neumann Laplacian: both problems are well-posed on $H^1(\Omega)$. Introducing the space of weakly 1-harmonic functions on Ω :

$$V_1(\Omega) := \left\{ u \in H^1(\Omega) \mid (-\Delta + 1)u = 0 \text{ weakly on } \Omega \right\},$$
(3.3)

we can define a Dirichlet 1-harmonic extension operator associated to Problem (3.1), denoted by $\mathcal{D}_{\Omega} : \mathcal{B}(\partial\Omega) \to V_1(\Omega)$, as well as a Neumann 1-harmonic extension operator associated to Problem (3.2), denoted by $\mathcal{S}_{\Omega} : \mathcal{B}'(\partial\Omega) \to V_1(\Omega)$. In the same way, we can define on the same spaces exterior 1-harmonic extension operators \mathcal{D}_{Ω^c} and \mathcal{S}_{Ω^c} with range $V_1(\Omega^c)$ for the problems set on the complement domain Ω^c (Definitions A.7 and A.11).

However, 0 is an eigenvalue for the Neumann Laplacian, as any constant is a solution to Problem (3.2) for k = 0 and g = 0: the problem is ill-posed on $H^1(\Omega)$. We are led to introduce a space similar to $H^1(\Omega)$, but on which constants have been removed:

$$H^1_{\langle 0 \rangle}(\Omega) := \left\{ u \in H^1(\Omega) \ \left| \ \int_{\Omega} u \, \mathrm{d}x = 0 \right\}, \right.$$

on which $\|\nabla \cdot\|_{L^2(\Omega)^n}$ defines a norm, equivalent to the usual H^1 norm by Poincaré's inequality. The interior trace operator restricted to $H^1_{\langle 0 \rangle}(\Omega)$ is no longer onto; let us denote $\mathcal{B}_0(\partial \Omega) := \operatorname{Tr}_i(H^1_{\langle 0 \rangle}(\Omega)) \subset \mathcal{B}(\partial \Omega)$. When the problem is set on $H^1(\Omega)$ (as in the case k = 1), all boundary conditions are understood in terms of the range of the trace: Dirichlet boundary conditions in $\mathcal{B}(\partial \Omega)$, Neumann boundary conditions in $\mathcal{B}'(\partial \Omega)$. Now that the problem is set on $H^1_{\langle 0 \rangle}(\Omega)$, it seems natural to adapt the space of conditions to the range of the trace on $H^1_{\langle 0 \rangle}(\Omega)$: Dirichlet boundary conditions in $\mathcal{B}_0(\partial \Omega)$ and Neumann boundary conditions in its dual space, $\mathcal{B}'_0(\partial \Omega)$. For boundary conditions understood in that sense, Problems (3.1) and (3.2) are well-posed on $H^1_{\langle 0 \rangle}(\Omega)$ for k = 0. Introducing the space of (weakly) harmonic functions on Ω in which the constants have been removed:

$$V_{\langle 0 \rangle}(\Omega) := \left\{ u \in H^1_{\langle 0 \rangle}(\Omega) \mid -\Delta u = 0 \text{ weakly on } \Omega \right\},\$$

we can define a Dirichlet harmonic extension operator associated to Problem (3.1), denoted by $\tilde{\mathcal{D}}_{\Omega} : \mathcal{B}_0(\partial\Omega) \to V_{\langle 0 \rangle}(\Omega)$, as well as a Neumann harmonic extension operator associated to Problem (3.2), denoted by $\tilde{\mathcal{S}}_{\Omega} : \mathcal{B}'(\partial\Omega) \to V_{\langle 0 \rangle}(\Omega)$ (Definitions A.7 and A.12). In Appendix A.1, we prove the space $\mathcal{B}_0(\partial\Omega)$ is an hyperplane of $\mathcal{B}(\partial\Omega)$ with complement the space of constant functions and equip it with a norm so that $\operatorname{Tr}_i : H^1_{\langle 0 \rangle}(\Omega) \to \mathcal{B}_0(\partial\Omega)$ defines an isometry. We define normal derivatives for elements of $H^1_{\langle 0 \rangle}(\Omega)$ with Laplacian in $L^2(\Omega)$ as elements of $\mathcal{B}'_0(\partial\Omega)$, which we identify with the subspace of $\mathcal{B}'(\partial\Omega)$ of functions null on constants. We prove that Problems (3.1) and (3.2) are well-posed on the spaces $H^1(\Omega)$ if k = 1 and $H^1_{\langle 0 \rangle}(\Omega)$ if k = 0, and define the associated (1-)harmonic extension operators S_{Ω} , \mathcal{D}_{Ω} and \tilde{S}_{Ω} and $\tilde{\mathcal{D}}_{\Omega}$ as linear continuous operators on the spaces mentioned above. We prove all four of those operators inherit the isometric properties of the trace operator (on the proper spaces) and define isometries themselves.

The Dirichlet harmonic extension from $\partial\Omega$ to Ω can also be understood from a probabilistic perspective, using the method described in Subsection 2.3, for any bounded Borel set Ω . Let $(B_t)_{t\geq 0}$ be an *n*-dimensional Brownian motion. Let τ be its exit time from Ω :

$$\tau := \inf\{t \ge 0, B_t \in \partial\Omega\}.$$

If $f \in C^0(\partial \Omega)$, then the function defined on Ω by:

$$u: x \longmapsto \mathbb{E}_x(f(B_\tau)) = \int_{\partial\Omega} f(y) \mathbb{P}_x(B_\tau \in \mathrm{d}y)$$
(3.4)

is the weak solution to the problem [6, Proposition 2.5]:

$$\begin{cases} \Delta u = 0 & \text{on } \Omega, \\ \operatorname{Tr}_{i} u = f & \text{on } \partial \Omega. \end{cases}$$
(3.5)

Note that for all $x \in \Omega$, $\tau < \infty$ a.s. since Ω is bounded. If n = 2, it is well known the Brownian trajectories are dense in the plane a.s., therefore $\tau < \infty$ a.s. and by [6, Proposition 2.5] once again, the solution to the exterior Dirichlet problem

$$\begin{cases} \Delta u = 0 & \text{on } \Omega^c, \\ \operatorname{Tr}_e u = f & \text{on } \partial\Omega, \end{cases}$$
(3.6)

are given by (3.4), only defined on Ω^c instead of Ω .

While the probabilistic approach allows to define a harmonic extension for any bounded Borel subset of \mathbb{R}^n , that extension can only be applied to bounded continuous functions on the boundary (which ensures formula (3.4) is well-defined). On the other hand, the variational approach allows to define the extension operator $\tilde{\mathcal{D}}_{\Omega}$ on a more general class of functions, $\mathcal{B}_0(\partial\Omega)$ (and even $\mathcal{B}(\partial\Omega)$), provided however that the domain Ω be an extension domain.

3.2 Poincaré-Steklov operator

As it was pointed out in Subsection 3.1, the spaces of solutions of Problems (3.1) and (3.2) coincide (for a given $k \in \{0,1\}$). The idea of the Poincaré-Steklov operator, also referred to as 'Dirichlet-to-Neumann' operator, is to highlight the correspondence between those problems, as the mapping $\operatorname{Tr}_i u \mapsto \frac{\partial_i u}{\partial \nu}$ where $u \in H^1(\Omega)$ is a solution to the Dirichlet problem (3.1). Generally speaking, considering the operator $(-\Delta + k)$ with $k \in \mathbb{R}$, we may define that operator when k is outside of the spectrum $\sigma(\Delta_D)$ of the Dirichlet Laplacian (Problem (3.1)). It is known that for an arbitrary domain Ω , all Dirichlet eigenvalues of $-\Delta$ are strictly positive and form an unbounded sequence $(\lambda_j)_{j\in\mathbb{N}}$ with $\lambda_j \to +\infty$ [45, Paragraph 6.5.1, Theorem 1]: k may only be an eigenvalue of the Dirichlet Laplacian Δ_D if k < 0 (*i.e.*, in the case of the Helmholtz operator). If $k \in \mathbb{R} \setminus \sigma(\Delta_D)$, the Dirichlet problem for $(-\Delta + k)$ has a unique solution $u \in H^1(\Omega)$ for each boundary condition $f \in \mathcal{B}(\partial\Omega)$. Hence, we can define the Poincaré-Steklov operator \mathcal{A}_k using that weak solution and establish similar properties regarding continuity and invertibility to those stated in [4] in the case of a d-set boundary $(d \in]n-2, n[)$, and generalized in [84] in the case of a domain with Jonsson boundary measures.

Conversely, when $k \in \mathbb{R} \setminus \sigma(\Delta_D)$ is also outside the spectrum $\sigma(\Delta_N)$ of the Neumann Laplacian (Problem (3.2)), the Neumann problem for $(-\Delta + k)$ has a unique solution $u \in H^1(\Omega)$ for each boundary condition $g \in \mathcal{B}'(\partial\Omega)$. In that case, the Poincaré-Steklov operator is invertible. Otherwise, we need to circumvent the ill-posedness of the Neumann problem by restricting the space of boundary conditions for Problems (3.1) and (3.2) so the Dirichlet and Neumann k-harmonic extension operators (defined as in Subsection 3.1) have the same range: that is why we defined the Dirichlet harmonic extension operator $\tilde{\mathcal{D}}_{\Omega}$ on $\mathcal{B}_0(\partial\Omega)$ instead of $\mathcal{B}(\partial\Omega)$ in the case $k = 0 \in \sigma(\Delta_N) \setminus \sigma(\Delta_D)$.

In Appendix A.2, we define the Poincaré-Steklov operator associated to the problem for $(-\Delta + k)$ with $k \in \mathbb{R} \setminus \sigma(\Delta_D)$ on a one-sided extension domain Ω (Definition A.14), and prove it is linear, continuous and coincides with its adjoint. Focusing on the cases $k = 1 \notin \sigma(\Delta_N)$ and $k = 0 \in \sigma(\Delta_N)$, we identify spaces on which they are invertible, and rely on the properties of the trace operator Tr_i to prove they define (Riesz) isometries (Theorem A.17). Finally, we connect the Poincaré-Steklov operator with the (1-)harmonic extension operators defined in Subsection 3.1.

Condensing all isometry results for the trace operator Tr_i , the 1-harmonic extension operators S_{Ω} and \mathcal{D}_{Ω} , and the Poincaré-Steklov operator for $(-\Delta+1)$ $\mathcal{A} = \mathcal{A}_1$, we obtain Diagram (3.7), where ϕ denotes the isometry from the Riesz representation theorem from a Hilbert space to its dual (regardless of that space):

An analogous diagram can be drawn with the operators associated with the Laplacian instead, by removing the constants in all the spaces. Doing so, the Poincaré-Steklov operator for $-\Delta$, $\tilde{\mathcal{A}} = \mathcal{A}_0$, becomes invertible. An analogous diagram of isomorphisms can also be drawn for the exterior problem associated to $(-\Delta + 1)$. The operators mentioned will not be isometric given our definition of $\|\cdot\|_{\mathcal{B}(\partial\Omega)}$ in Theorem 2.9 which relies on the interior trace operator. However, as it was discussed in Proposition 2.11, we could define an equivalent norm using the operator Tr_e instead. Considering that norm for $\mathcal{B}(\partial\Omega)$ (and modifying the norm on $\mathcal{B}'(\partial\Omega)$ accordingly), the operators in the diagram for the exterior problem would be isomorphisms.

4 Transmission problems

In this section, we consider the following transmission problem:

$$\begin{cases} (-\Delta + k)u = 0 & \text{on } \mathbb{R}^n \backslash \partial\Omega, \\ [Tr \, u] = -f, \\ [\frac{\partial u}{\partial \nu}] = g, \end{cases}$$
(4.1)

where $k \in \{0, 1\}$ and f, g are given boundary condition. The spirit of such a problem is to identify the solution based on the influence of the boundary (*i.e.* its jumps across the boundary), rather than its boundary values themselves as in Section 3. The choice of the values of k is due to the fact that 1 is not an eigenvalue of the transmission Laplacian, while 0 is. Naturally, the results described here can be adapted to other values of k, bearing that distinction in mind.

In Subsection 4.1, we introduce the layer potential operators in the Lipschitz case, which are used to find the solution the transmission problem. Those operators are defined using the fundamental solution to the operator under study, and we generalize the link between that function and the harmonic measure in the case of extension domains. The notions of layer potential operators are generalized in Appendix B.1 as linear continuous operators. In Subsection 4.2 and Appendix B.2, we consider the truncated transmission problem (*i.e.*, the transmission problem set on a large truncation domain instead of \mathbb{R}^n) and prove convergence results of the solutions to those of the original transmission problem in the case of extension domains. In Subsection 4.4, we aim towards generalizing the notion of Poincaré-Steklov operators for transmission problems by giving two possible definitions; we prove those definitions coincide in Appendix B.4. In Subsection 4.3, we describe the operators relevant to the study of boundary values for the layer potential operators in the Lipschitz case, notably the Neumann-Poincaré operator. We generalize those notions in the framework of two-sided extension domains and prove symmetrization formulae for the Neumann-Poincaré operator in Appendix B.3. In Subsection 4.5 and Appendix B.5, we recall the link between contracting operators and Neumann series. We introduce norms on $\mathcal{B}(\partial\Omega)$ and $\mathcal{B}'(\partial\Omega)$ using the operators from the previous section and prove contracting properties for the Neumann-Poincaré operator. We prove the real spectrum of that operator is included in $]-\frac{1}{2},\frac{1}{2}[$. Finally, in Subsection 4.6, we state results proved in the Lipschitz framework regarding the resolution of imaging problems: the representation formula for the solution and the identification of subdomains in the monotonous case and in the case of disks. We generalize those results in the framework of extension domains in Appendix B.6.

4.1 Layer potential operators

In the classical case -i.e., when the boundary is Lipschitz -, solving the transmission problem (4.1) for k = 0 leads to the introduction of the so-called *layer potential operators* (see for instance [1, 61, 75]), defined by:

$$\tilde{\mathcal{S}}_{\partial\Omega} g(x) = \int_{\partial\Omega} \tilde{G}(x-y)g(y)\,\sigma(\mathrm{d}y), \quad x \in \mathbb{R}^n,$$

for the single layer potential operator, and

$$\tilde{\mathcal{D}}_{\partial\Omega}f(x) = \int_{\partial\Omega} \frac{\partial \tilde{G}}{\partial \nu_y}(x-y)g(y)\,\sigma(\mathrm{d}y), \quad x \in \mathbb{R}^n \backslash \partial\Omega,$$

for the double layer potential operator, where \tilde{G} is the fundamental solution to $-\Delta$ on \mathbb{R}^n :

$$\tilde{G}: x \in \mathbb{R}^n \longmapsto \begin{cases} \frac{1}{2\pi} \ln |x| & \text{if } n = 2, \\ \frac{1}{(2-n)\omega_n} |x|^{2-n} & \text{if } n \ge 3, \end{cases}$$

where ω_n is the area of the unit sphere of \mathbb{R}^n . Here, the elements f and gare in $L^2(\partial\Omega, \sigma)$, and $\int_{\partial\Omega} g \, d\sigma = 0$. For such f and g, the layer potentials $\tilde{S}_{\partial\Omega} g$ and $\tilde{\mathcal{D}}_{\partial\Omega} f$ are harmonic on $\mathbb{R}^n \setminus \partial\Omega$, and the solution to the transmission problem (4.1) for k = 0 can be expressed in terms of the layer potentials as [85, Proposition 3.5.1]:

$$u = \tilde{\mathcal{S}}_{\partial\Omega} g - \tilde{\mathcal{D}}_{\partial\Omega} f.$$
(4.2)

This formula is known as Green's third identity.

In the case of a two-sided extension domain Ω , the classical definitions of the layer potentials can no longer be used: Ω has no specified boundary measure, and even then, the normal derivative $\frac{\partial}{\partial\nu}\tilde{G}$ is understood as an element of $\mathcal{B}'(\partial\Omega)$ and might not be integrable; same goes for the boundary condition $g \in \mathcal{B}'_0(\partial\Omega)$. That is why, in order to generalize the notion of layer potential operators in the case of two-sided extension domains, we focus on the connection those operators have with the solution to the transmission problem (Formula (4.2)).

In Appendix B.1, we introduce spaces on which the transmission problems (4.1) for k = 1 and k = 0 are well-posed. We use a variational approach to define the layer potential operators for the transmission problems (4.1) for k = 1 and k = 0 and prove those operators can be written in terms of convolutions with the fundamental solution to the associated problem, which goes to show our new definitions generalize the classical definitions of those operators. Such an approach was used in [7], only in a Lipschitz framework. We prove the newly defined operators are linear and continuous, and generalize Green's third identity (4.2) in that framework.

We generalize the connection between the normal derivative of the fundamental solution, used to define the double layer potential operator, and the harmonic measure, proved in [8, Proposition 3.11] in the case of strongly harmonic functions on regular domains. Let Ω be a one-sided extension domain of \mathbb{R}^n . Let Γ be the Green function for the Laplacian on Ω with homogeneous Dirichlet boundary condition:

$$\forall x \in \Omega, \qquad \begin{cases} \Delta \Gamma(x, \cdot) = \delta_x & \text{on } \Omega, \\ \operatorname{Tr} \Gamma(x, \cdot) = 0 & \text{on } \partial \Omega. \end{cases}$$
(4.3)

If G denotes the fundamental solution to the Laplacian on \mathbb{R}^n , then for $x \in \Omega$, it holds:

$$\Gamma(x,\cdot) = \tilde{G}(x-\cdot) - h_x \quad \text{on } \Omega, \tag{4.4}$$

where h_x is harmonic on Ω and has the same trace as $\tilde{G}(x-\cdot)$ on $\partial\Omega$. Denoting $\Gamma_x := \Gamma(x, \cdot)$, it follows that:

$$\forall x \in \Omega, \quad \Gamma_x \in H^1_0(\Omega). \tag{4.5}$$

The harmonic correction can either be expressed using a variational approach as $h_x = \tilde{\mathcal{D}}(\operatorname{Tr} \tilde{G}(x - \cdot))$, where $\tilde{\mathcal{D}}$ is the harmonic extension defined in Subsection 3.1, or using the probabilistic formula (3.4).

Theorem 4.1. Let Ω be a one-sided extension domain of \mathbb{R}^n . Then, for all $x \in \Omega$ and $f \in \mathcal{B}(\partial \Omega)$ bounded, it holds:

$$\left\langle \frac{\partial \Gamma_x}{\partial \nu}, f \right\rangle_{\mathcal{B}', \mathcal{B}(\partial \Omega)} = \frac{1}{\omega_n} \int_{\partial \Omega} f(y) \mathbb{P}_x(B_\tau \in \mathrm{d}y),$$
 (4.6)

where ω_n is the measure of the unit sphere in \mathbb{R}^n .

Proof. We adapt the proof of [8, Proposition 3.11]. Let $x \in \Omega$. Let $\varepsilon > 0$ small enough so that $\overline{B_{\varepsilon}(x)} \subset \Omega$, where $B_{\varepsilon}(x)$ is the ball of \mathbb{R}^n of center x and radius ε . Then, by Green's formula on $\Omega \setminus \overline{B_{\varepsilon}(x)}$, it holds (normal derivatives in duality pairings are in $\mathcal{B}'(\partial\Omega)$ on $\partial\Omega$; those in integrals on in $L^2(\partial B_{\varepsilon}(x), \sigma)$ on $\partial B_{\varepsilon}(x)$):

$$\begin{cases} \left\langle \frac{\partial u}{\partial \nu}, \operatorname{Tr} \Gamma_x \right\rangle_{\mathcal{B}', \mathcal{B}(\partial \Omega)} - \int_{\partial B_{\varepsilon}(x)} \frac{\partial u}{\partial \nu} \Gamma_x \, \mathrm{d}\sigma = \int_{\Omega \setminus \overline{B_{\varepsilon}(x)}} \nabla u \cdot \nabla \Gamma_x \, \mathrm{d}y, \\ \left\langle \frac{\partial \Gamma_x}{\partial \nu}, \operatorname{Tr} u \right\rangle_{\mathcal{B}', \mathcal{B}(\partial \Omega)} - \int_{\partial B_{\varepsilon}(x)} \frac{\partial \Gamma_x}{\partial \nu} u \, \mathrm{d}\sigma = \int_{\Omega \setminus \overline{B_{\varepsilon}(x)}} \nabla u \cdot \nabla \Gamma_x \, \mathrm{d}y, \end{cases}$$
(4.7)

where σ denotes the (n-1)-dimensional Lebesgue measure.

Since u is harmonic on $B_{\varepsilon}(x)$, it holds:

$$\int_{\partial B_{\varepsilon}(x)} \frac{\partial u}{\partial \nu} \Gamma_x \, \mathrm{d}\sigma = \int_{B_{\varepsilon}(x)} \nabla u \cdot \nabla \Gamma_x \, \mathrm{d}y \xrightarrow[\varepsilon \to 0]{} 0, \tag{4.8}$$

using the regularity of Γ_x from (4.5). In the same way, it holds:

$$\int_{\partial B_{\varepsilon}(x)} \frac{\partial h}{\partial \nu} u \, \mathrm{d}\sigma \xrightarrow[\varepsilon \to 0]{} 0,$$

where h is defined by (4.4). Moreover, since $B_{\varepsilon}(x)$ is a ball, it holds [1, Lemma 2.2]:

$$\int_{\partial B_{\varepsilon}(x)} \frac{\partial G(x, \cdot)}{\partial \nu} u \, \mathrm{d}\sigma = \frac{\varepsilon^{1-n}}{\omega_n} \int_{\partial B_{\varepsilon}(x)} u \, \mathrm{d}\sigma \ \underset{\varepsilon \to 0}{\longrightarrow} \frac{1}{\omega_n} u(x)$$

The continuity of u by Weyl's lemma [96, Lemma 2] guarantees the convergence. Hence:

$$\int_{\partial B_{\varepsilon}(x)} \frac{\partial \Gamma_x}{\partial \nu} u \, \mathrm{d}\sigma \, \underset{\varepsilon \to 0}{\longrightarrow} \, \frac{1}{\omega_n} u(x) = \frac{1}{\omega_n} \int_{\partial \Omega} f(y) \, \mathbb{P}_x(B_{\tau} \in \mathrm{d}y), \tag{4.9}$$

using the representation (3.4) of the harmonic function u in terms of the harmonic measure on $\partial\Omega$.

Rewriting the Green formulae (4.7) with the boundary conditions (4.3) and (3.5), the limits (4.8) and (4.9), and calculating the difference yields (4.6).

Corollary 4.2. Let Ω be a one-sided extension domain of \mathbb{R}^n . Let $x \in \Omega$ and assume there exists a d-upper regular measure μ_x on $\partial\Omega$ for some $d \in]n-2, n[$ (i.e., a measure as in Definition 2.4) such that $\frac{\partial}{\partial\nu}\Gamma_x \in L^1(\partial\Omega, \mu_x)$. Then, the normal derivative of the Green function Γ_x is expressed in terms of the harmonic measure $\mathbb{P}_x(B_\tau \in dy)$ and the boundary measure μ_x as:

$$\frac{\partial \Gamma_x}{\partial \nu}(y) = \frac{1}{\omega_n} \frac{\mathbb{P}_x(B_\tau \in \mathrm{d}y)}{\mu_x(\mathrm{d}y)}.$$

4.2 Approximation by truncated problems

A usual approximation [3, 4] of the transmission problem (4.1) consists in considering that the exterior domain Ω^c has been truncated. In other words, we consider a two-sided extension domain Ω of \mathbb{R}^n , and a bounded one-sided extension domain U such that $\Omega \subset \subset U$. We wish to approximate the transmission problem on the whole space (4.1) with $k \in \{0, 1\}$, by a problem set on U:

$$\begin{cases} -\Delta u + ku = 0 & \text{on } U \backslash \partial \Omega, \\ [Tr \, u]_{\partial \Omega} = -f, & \left[\left[\frac{\partial u}{\partial \nu} \right]_{\partial \Omega} = g, \\ Tr_i^{\partial U} = 0, \end{cases}$$
(4.10)

with $f \in \mathcal{B}(\partial\Omega)$ and $g \in \mathcal{B}'(\partial\Omega)$ if k = 1 and $g \in \mathcal{B}'_0(\partial\Omega)$ if k = 0, where $\llbracket \cdot \rrbracket_{\partial\Omega}$ denotes the boundary jump value across $\partial\Omega$ and $\operatorname{Tr}_i^{\partial U}$ denotes the interior trace operator on ∂U (similar notations will be used for trace operators on $\partial\Omega$). Note that, in this subsection, only the interior trace operator on ∂U will be considered, as well as normal derivatives on $\partial\Omega$; we will simply denote them by $\operatorname{Tr}^{\partial U}$ and $\frac{\partial_i}{\partial\nu}, \frac{\partial_e}{\partial\nu}$ respectively.

Naturally, our main interest regarding the truncated problem is its convergence towards the transmission problem set on the whole space, when the truncation domain U grows to \mathbb{R}^n . In Appendix B.2, we define the single and double layer potential operators for the truncated problem (4.10) for k = 1 and k = 0 in the case of extension domains, and prove the space of solutions to the truncated problem can be regarded as the space of solutions to the problem on the whole space (4.1) (using the Dirichlet boundary condition on U). In that space, we establish the convergence of the layer potential operators for the truncated problems to those on the whole space when the truncating domains grows at least as fast as a growing sequence of balls, generalizing the result from [3] in the Lipschitz case and [4] on *d*-sets.

4.3 Neumann-Poincaré operator and boundary values

The layer potential operators allow to determine the solution to the transmission problem (4.1) from its jumps in trace and normal derivative. One thing we can be interested in knowing about that solution is what its boundary values actually are, since all that was fixed when setting the problem where the jumps. From Proposition B.7, it is known any solution to the transmission problem can be split into a single layer potential part and a double layer potential part, and it seems natural to keep that decomposition in mind when determining the boundary values. The trace value of a single layer potential (denoted by $\tilde{\mathcal{V}}_{\partial\Omega}$ if k = 0) as well as the normal derivative value of a double layer potential (denoted by $-\tilde{\mathcal{W}}_{\partial\Omega}$ if k = 0) are easily defined [63, 89, 91], due to the absence of jumps in the quantities in question. From their definition, double layer potentials have a jump in trace. Knowing that jump, it is enough to have access to the average of the traces to recover its trace values. In the case of the transmission problem (4.1) for k = 0 across a Lipschitz boundary, that average is given by the Neumann-Poincaré operator [1, 64, 80], defined as the following singular integral:

$$\tilde{\mathcal{K}}_{\partial\Omega}f(x) = \frac{1}{\omega_n} \text{ p.v.} \int_{\partial\Omega} \frac{\langle y - x, \nu_y \rangle}{|x - y|^n} f(y) \,\sigma(\mathrm{d}y),$$

with p.v. the Cauchy principal value, ω_n the area of the unit sphere of \mathbb{R}^n and ν the outward normal vector field to $\partial\Omega$. It holds:

$$\tilde{\mathcal{K}}_{\partial\Omega} = \frac{1}{2} \left(\mathrm{Tr}_i + \mathrm{Tr}_e \right) \circ \tilde{\mathcal{D}}_{\partial\Omega}.$$
(4.11)

In the classical case, that operator is understood as $\tilde{\mathcal{K}}_{\partial\Omega}$: $L^2(\partial\Omega, \sigma) \rightarrow L^2(\partial\Omega, \sigma)$. One important property of the Neumann-Poincaré operator is that its adjoint, expressed as:

$$\tilde{\mathcal{K}}^*_{\partial\Omega}f(x) = \frac{1}{\omega_n} \text{ p.v.} \int_{\partial\Omega} \frac{\langle x - y, \nu_x \rangle}{|x - y|^n} f(y) \,\sigma(\mathrm{d}y),$$

satisfies (see for instance [66, Proposition 12]):

$$\tilde{\mathcal{K}}^*_{\partial\Omega} = \frac{1}{2} \left(\frac{\partial_i}{\partial\nu} + \frac{\partial_e}{\partial\nu} \right) \circ \tilde{\mathcal{S}}_{\partial\Omega}.$$
(4.12)

The four boundary operators introduced above are used to defined the Claderón projectors [80, Theorem 3.1.3]:

$$\tilde{\mathcal{C}}_i := \frac{1}{2}I + \tilde{\mathcal{M}}$$
 and $\tilde{\mathcal{C}}_e := \frac{1}{2}I - \tilde{\mathcal{M}},$

where

$$\tilde{\mathcal{M}} := \begin{pmatrix} -\tilde{\mathcal{K}}_{\partial\Omega} & \tilde{\mathcal{V}}_{\partial\Omega} \\ \\ \tilde{\mathcal{W}}_{\partial\Omega} & \tilde{\mathcal{K}}_{\partial\Omega}^* \end{pmatrix} : H^{\frac{1}{2}}(\partial\Omega) \times H^{-\frac{1}{2}}(\partial\Omega) \to H^{\frac{1}{2}}(\partial\Omega) \times H^{-\frac{1}{2}}(\partial\Omega),$$

the study of which allows to prove the Neumann-Poincaré operators and boundary value operators can be symmetrized:

$$\begin{cases} \tilde{\mathcal{K}}_{\partial\Omega}\tilde{\mathcal{V}}_{\partial\Omega} = \tilde{\mathcal{V}}_{\partial\Omega}\tilde{\mathcal{K}}_{\partial\Omega}^{*}, \\ \tilde{\mathcal{W}}_{\partial\Omega}\tilde{\mathcal{K}}_{\partial\Omega} = \tilde{\mathcal{K}}_{\partial\Omega}^{*}\tilde{\mathcal{W}}_{\partial\Omega}, \end{cases} \quad \text{and} \quad \begin{cases} \tilde{\mathcal{K}}_{\partial\Omega}^{2} + \tilde{\mathcal{V}}_{\partial\Omega}\tilde{\mathcal{W}}_{\partial\Omega} = \frac{1}{4}I, \\ (\tilde{\mathcal{K}}_{\partial\Omega}^{*})^{2} + \tilde{\mathcal{W}}_{\partial\Omega}\tilde{\mathcal{V}}_{\partial\Omega} = \frac{1}{4}I. \end{cases}$$
(4.13)

In Appendix B.3, we define the trace of the single layer potential operator and the normal derivative of the double layer potential operator in the case of two-sided extension domains, as linear and continuous on the trace spaces and their duals. We generalize the notion of Neumann-Poincaré operator for the transmission problem (4.1) for k = 1 and k = 0, using Relation (4.11) as our definition, and prove the connection (4.12) between its adjoint and our generalized single layer potential operator still holds. We prove the Neumann-Poincaré operator and its adjoint are continuous. Finally, we generalize the notions of Calderón projectors and prove the symmetrization formulae (4.13) in the framework of two-sided extension domains.

4.4 Poincaré-Steklov operators for transmission problems

We wish to define Poincaré-Steklov operators for the transmission problem associated to $(-\Delta + 1)$ on \mathbb{R}^n . In the case of the bounded problem set on Ω , the interior Poincaré-Steklov operator $\mathcal{A} : \mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ is defined by (see Subsection 3.2):

$$\mathcal{A}(\mathrm{Tr}_i \, u) = \frac{\partial_i u}{\partial \nu},$$

for every $u \in V_1(\Omega)$ (defined by (2.3)). Although one could be tempted to proceed in the same way, *i.e.* map a Dirichlet boundary condition to a Neumann boundary condition, the solutions to the transmission problem are such that the jump in trace and the jump in normal derivative can be chosen independently. If the space of solutions to the transmission problem is denoted by:

$$V_{1,\partial\Omega}(\mathbb{R}^n) := \left\{ u \in H^1_{\partial\Omega}(\mathbb{R}^n) \mid -\Delta u + u = 0 \text{ weakly on } \mathbb{R}^n \backslash \partial\Omega \right\},\$$

then it holds (Theorem B.7):

$$V_{1,\partial\Omega}(\mathbb{R}^n) = \mathcal{S}_{\partial\Omega}(\mathcal{B}'(\partial\Omega)) \oplus \mathcal{D}_{\partial\Omega}(\mathcal{B}(\partial\Omega)), \qquad (4.14)$$

where the direct sum is orthogonal with respect to the norm on $H^1_{\partial\Omega}(\mathbb{R}^n)$ by Green's formula: for $f \in \mathcal{B}(\partial\Omega)$ and $g \in \mathcal{B}'(\partial\Omega)$,

$$\langle \mathcal{S}_{\partial\Omega} g, \mathcal{D}_{\partial\Omega} f \rangle_{H^{1}_{\partial\Omega}} = \int_{\mathbb{R}^{n}} (\mathcal{S}_{\partial\Omega} g) (\mathcal{D}_{\partial\Omega} f) \, \mathrm{d}x + \int_{\Omega \cup \Omega^{c}} \nabla (\mathcal{S}_{\partial\Omega} g) \cdot \nabla (\mathcal{D}_{\partial\Omega} f) \, \mathrm{d}x \\ = \left\langle \left[\left[\frac{\partial}{\partial \nu} (\mathcal{D}_{\partial\Omega} f) \right] \right], \left[[\mathrm{Tr}(\mathcal{S}_{\partial\Omega} g)] \right] \right\rangle_{\mathcal{B}', \mathcal{B}} = 0$$

The latter spaces are described by:

$$\mathcal{S}_{\partial\Omega}(\mathcal{B}'(\partial\Omega)) = \left\{ u \in V_{1,\partial\Omega}(\mathbb{R}^n) \mid \llbracket \operatorname{Tr} u \rrbracket = 0 \right\},\\ \mathcal{D}_{\partial\Omega}(\mathcal{B}(\partial\Omega)) = \left\{ u \in V_{1,\partial\Omega}(\mathbb{R}^n) \mid \llbracket \frac{\partial u}{\partial\nu} \rrbracket = 0 \right\}$$

Instead of considering any solution to the transmission problem, it seems natural from Decomposition (4.14) to consider solutions with either a homogeneous jump in trace $(u = S_{\partial\Omega}[\frac{\partial u}{\partial \nu}])$ or a homogeneous jump in normal derivative $(u = -\mathcal{D}_{\partial\Omega}[\text{Tr} u])$. In this way, the Poincaré-Steklov operators we wish to

define must link the jump in trace of a certain solution $u = -\mathcal{D}_{\partial\Omega}[[\operatorname{Tr} u]]$ to the jump in normal derivative of *another* solution $v = \mathcal{S}_{\partial\Omega}[[\frac{\partial u}{\partial \nu}]]$, where u and v are connected somehow. From [89], in the Lipschitz case, the (interior) Poincaré-Steklov operator for the transmission problem can be defined as follows:

$$\mathcal{V}_{\partial\Omega}^{-1}\left(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega}\right) : \mathcal{B}(\partial\Omega) \longrightarrow \mathcal{B}'(\partial\Omega) \\ - \left[\left[\operatorname{Tr} u\right]\right] \xrightarrow{(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega})} \operatorname{Tr}_{i} u = \operatorname{Tr} v \xrightarrow{\mathcal{V}_{\partial\Omega}^{-1}} \left[\left[\frac{\partial v}{\partial\nu}\right]\right]$$
(4.15)

where $u = -\mathcal{D}_{\partial\Omega} \llbracket \operatorname{Tr} u \rrbracket$ and $v = \mathcal{S}_{\partial\Omega} \llbracket \frac{\partial v}{\partial \nu} \rrbracket$.

As explained before, this operator links two solutions to the transmission problem: a solution u with no jump in normal derivative, and a solution vwith no jump in trace. The connection between those solutions is done using the interior trace of u. However, one could choose to use the normal derivative of u as the connection, and define the interior Poincaré-Steklov operator as follows:

$$-\left(\frac{1}{2}I + \mathcal{K}_{\partial\Omega}^{*}\right)^{-1} \mathcal{W}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \longrightarrow \mathcal{B}'(\partial\Omega)$$
$$-\left[\left[\operatorname{Tr} u\right]\right] \xrightarrow{-\mathcal{W}_{\partial\Omega}} \frac{\partial u}{\partial\nu} = \frac{\partial_{i}v}{\partial\nu} \xrightarrow{\left(\frac{1}{2}I + \mathcal{K}_{\partial\Omega}^{*}\right)^{-1}} \left[\left[\frac{\partial v}{\partial\nu}\right]\right] \quad (4.16)$$

where $u = -\mathcal{D}_{\partial\Omega} \llbracket \operatorname{Tr} u \rrbracket$ and $v = \mathcal{S}_{\partial\Omega} \llbracket \frac{\partial v}{\partial \nu} \rrbracket$.

In Appendix B.4, we prove the operators defined by (4.15) and (4.16) coincide, and generalize the notion of interior Poincaré-Steklov operator for the transmission problem (4.1) for k = 1 in the framework of two-sided extension domains, and define the exterior Poincaré-Steklov operator in a similar way. Moreover, we give an interpretation of those operators in terms of solutions to the transmission problem.

4.5 Norm estimates and Neumann series

We are interested in the spectral properties of the Neumann-Poincaré operators defined in Subsection 4.3. When defined on $L^2(\partial\Omega, \sigma)$ for Lipschitz boundaries, the spectrum of the operator $\tilde{\mathcal{K}}^*_{\partial\Omega}$ (hence, that of its adjoint $\tilde{\mathcal{K}}_{\partial\Omega}$) lies in $\left[-\frac{1}{2}, \frac{1}{2}\right]$ [1, 44, 62]. That property is used to give an explicit formula to two-phased transmission problems (Equation (4.19)), see Subsection 4.6.

It is well-known and easy to verify that the resolvent of an operator L can be expressed in terms of the associated Neumann series:

$$\sum_{k=0}^{+\infty} L^k = (I - L)^{-1}, \qquad (4.17)$$

provided the series converges. In particular, that equality is valid anytime ||L|| < 1. In the Lipschitz case [89, Theorem 3.2], it is proved the operators
$(-\frac{1}{2}I + \tilde{\mathcal{K}}_{\partial\Omega})$ is (strictly) contractive for the norm on $L^2(\partial\Omega, \sigma)$ defined by:

$$\|\cdot\|_{\tilde{\mathcal{V}}_{\partial\Omega}^{-1}} := \sqrt{\langle \tilde{\mathcal{V}}_{\partial\Omega}^{-1} \cdot, \cdot \rangle_{L^2(\partial\Omega,\sigma)}}, \qquad (4.18)$$

where $\tilde{\mathcal{V}}_{\partial\Omega}$: $L^2(\partial\Omega, \sigma) \to L^2(\partial\Omega, \sigma)$ is the trace of the single layer potential operator for the Laplacian, defined in Subsection 4.3. In particular, the corresponding Neumann series (4.17) converges.

In Appendix B.5, we define norms in the spirit of Equation (4.18) associated to $\mathcal{V}_{\partial\Omega}$, $\mathcal{V}_{\partial\Omega}^{-1}$ and $\mathcal{W}_{\partial\Omega}$, as well as $\tilde{\mathcal{V}}_{\partial\Omega}$, $\tilde{\mathcal{V}}_{\partial\Omega}^{-1}$ and $\tilde{\mathcal{W}}_{\partial\Omega}$ in the framework of two-sided extension domains: those norms are defined on the trace space and its dual space. In our case, those operators are defined using the trace spaces $\mathcal{B}(\partial\Omega)$ and $\mathcal{B}_0(\partial\Omega)$ and their dual $\mathcal{B}'(\partial\Omega)$ and $\mathcal{B}'_0(\partial\Omega)$. We use the continuity of the operators mentioned above to prove those norms are equivalent to the usual norms on $\mathcal{B}(\partial\Omega)$, $\mathcal{B}'(\partial\Omega)$ and $\mathcal{B}_0(\partial\Omega)$, $\mathcal{B}'_0(\partial\Omega)$, and later on express the equivalence constants using the two-sided extension properties of the domain under study. We generalize the contraction property of the operator $(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega})$ from [89] using the generalized definition of the operator from Appendix B.3, prove a similar property for the operator $(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega})$, and use results from perturbation theory to show the real spectra of the operators $\mathcal{K}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}(\partial\Omega)$ and $\mathcal{K}^*_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ are included in $] - \frac{1}{2}, \frac{1}{2}[$, which generalizes the Lipschitz case [27, 83]. We prove similar results for the operators $\tilde{\mathcal{K}}_{\partial\Omega} : \mathcal{B}_0(\partial\Omega) \to \mathcal{B}_0(\partial\Omega)$ and $\tilde{\mathcal{K}^*_{\partial\Omega} : \mathcal{B}'_0(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega)$.

4.6 Imaging

Consider a bounded domain Ω inside which is an inclusion $D \subset \Omega$, characterized by its conductivity $k \in]0, 1[\cup]1, +\infty[$ (Figure 3). We try to identify the inclusion D based only on boundary values on $\partial\Omega$ of the solution to the following problem:

$$\begin{cases} \nabla \cdot \left(\left(1 + (k-1) \mathbb{1}_D \right) \nabla u \right) = 0 \quad \text{on } \Omega, \\ \frac{\partial_i u}{\partial \nu} \Big|_{\partial \Omega} = g, \\ \int_{\Omega} u \, \mathrm{d}x = 0, \end{cases}$$
(4.19)

where g is a given Neumann boundary condition and $\mathbb{1}_D$ denotes the indicator function of the subset D.

In the Lipschitz case [1, 61], the Neumann boundary condition is understood as $g \in L^2_0(\partial\Omega, \sigma)$, where:

$$L_0^2(\partial\Omega,\sigma) = \left\{ g \in L^2(\partial\Omega,\sigma) \ \left| \ \int_{\partial\Omega} g \, \mathrm{d}\sigma = 0 \right\} \right\}.$$

That space is consistent with being the normal derivative of a harmonic function, by Green's formula. Using the classical definition of layer potentials on



Figure 3: Configuration of an imaging problem in the case of extension domains.

 $L^2(\partial\Omega, \sigma)$ given in Subsection 4.1, the solution u to the imaging problem (4.19) can be expressed as [61, Theorem 2.1]:

$$u = H - \tilde{\mathcal{S}}_{\partial D} \varphi$$
 where $H = \tilde{\mathcal{S}}_{\partial \Omega} g - \tilde{\mathcal{D}}_{\partial \Omega} f$, (4.20)

with

$$\left(\frac{k+1}{2(k-1)}I + \tilde{\mathcal{K}}^*_{\partial D}\right)\varphi = \left.\frac{\partial H}{\partial\nu}\right|_{\partial D},\,$$

defined using the adjoint of the Neumann-Poincaré operator in the classical sense, see Subsection 4.3, and $f = \text{Tr}_i u$ on $\partial\Omega$. That representation formula is the key ingredient when it comes to subdomain identification; we are specifically interested in two identification results [1] said 'with one measurement' (only one value of g is needed): the identification in the monotonous case which allows to prove two possible inclusions are the same under the assumption that one contains the other, and the identification in the case of disks in \mathbb{R}^2 , which identifies inclusions of a fixed shape.

In Appendix B.6, we generalize the representation formula (4.20) in the case of a two-sided extension domain Ω . The Neumann boundary condition is understood in the sense of $\mathcal{B}'_0(\partial\Omega)$ and the operators involved are understood in the generalized variational sense. Then, we generalize the identification results in the monotonous case and in the case of disks of \mathbb{R}^2 . Note that, although the latter result only applies to disks in \mathbb{R}^2 , which are smooth, the larger domain Ω is still a two-sided extension domain: our approach relies on the abstract definition of the layer potential operators and on the general notions of Dirichlet and Neumann boundary conditions, in the sense of the trace operators and the weak normal derivatives.

5 Hilbert transform and Cauchy integral

In this section, we study the Hilbert transform and the Cauchy integrals as operators connected to the layer potential operators. The most fundamental definition of the Hilbert transform can be seen as a convolution with the kernel $y \in \mathbb{R} \mapsto \frac{1}{\pi y}$. That kernel being singular at point 0, that convolution is actually defined as a singular integral [8, Chapter IV, Section 2]:

$$\mathfrak{H}f: x \longmapsto \lim_{\substack{\varepsilon \to 0 \\ N \to \infty}} \frac{1}{\pi} \int_{\varepsilon < |y| < N} \frac{f(x-y)}{y} \, \mathrm{d}y,$$

for $f : \mathbb{R} \to \mathbb{R}$ such that the limit exists. In other words, the Hilbert transform is defined as the principal value of the convolution described above [88, Prologue]. Understanding the action of this Hilbert transform can be done by determining its frequency response [8, Chapter IV, Section 2, Proposition 2.3]:

Proposition 5.1. Let $f \in C_0^1(\mathbb{R})$. Then, the Fourier transform of the Hilbert transform of f is given by:

$$\widehat{\mathfrak{H}f}(\xi) = -i\operatorname{sgn}(\xi)\widehat{f}(\xi)$$
 a.e.

Therefore, the Hilbert transform acts as a rotation of $\pm \frac{\pi}{2}$ of the frequency components, depending on their sign.

The Hilbert transform is also a topic of interest from a probabilistic perspective, and many of its properties are proved using probabilistic tools, such as martingale transforms and projections. Let $(B_t)_{t\geq 0}$ be an *n*-dimensional Brownian motion and $(\mathcal{F}_t)_{t\in[0,\infty]}$ be the filtration generated by the process. If $X \in L^2(\mathcal{F}_\infty)$, then there exists a process $(M_t)_{t\geq 0}$ $(\mathcal{F}_t)_{t\geq 0}$ -adapted and satisfying:

$$\mathbb{E}\left(\int_0^{+\infty} |M_s|^2 \,\mathrm{d}s\right) < +\infty,$$

such that [69, Section 5.3, Theorem 5.7]:

$$X = \mathbb{E}(X) + \int_0^{+\infty} H_s \cdot \mathrm{d}B_s.$$

Then, if A is an $n \times n$ real matrix, the martingale transform of X by A is defined by [11, Section 1]:

$$A * X := \int_0^{+\infty} AH_s \cdot \mathrm{d}B_s \ \in L^2(\mathcal{F}_\infty).$$

Considering projections of those martingale transforms (*i.e.* considering conditional expectations of those processes) allows to get insightful inequalities, which can be used to prove the Hilbert transform is of strong type (p, p) (see [97, Definition 9.3]) for all $p \in]1, +\infty[$, see [11, 12].

In the complex plane \mathbb{C} , alternative definitions of the Hilbert transform [63, 79, 81] use the boundary of a Lipschitz domain $\partial\Omega$ as singularity points, rather than 0:

$$\mathcal{H}_{\partial\Omega}f(z) := \frac{1}{\pi} \int_{\partial\Omega} \frac{f(\zeta)}{\zeta - z} \,\mathrm{d}\zeta,$$

where $f : \partial \Omega \to \mathbb{C}$ is such that the integral is defined and $z \in \mathbb{C}$. In that case, the Hilbert transform is directly connected to the Cauchy integral [81]:

$$\Phi_{\partial\Omega}f = \frac{1}{2i}\mathcal{H}_{\partial\Omega}f.$$

This integral has been widely studied for Lipschitz curves [23, 34, 33], but also on more general curves, while still assuming the existence of a regular enough boundary measure [76]. We wish to use the generalized definitions of the layer potential operators given in Subsection 4.1 to define the Hilbert transform and Cauchy integral over the boundary of any two-sided extension domain.

In Subsection 5.1, we highlight the connection between Hilbert transform and layer potential operators in the complex plane in the Lipschitz case. In Subsection 5.2, we introduce two possible definitions of the Hilbert transform in the case of a two-sided extension domain and describe a method to prove those definitions coincide. In Appendix C, we prove results following that method: the dyadic approximation of general domains and the convergence of the Dirichlet harmonic extension operators.

5.1 Lipschitz case: connection with the layer potential operators

First, let us highlight the connection between the layer potential operators and the Hilbert transform (or the Cauchy integral) in the complex plane. Let Ω be a Lipschitz domain of \mathbb{C} (*i.e.* the image of a Lipschitz domain of \mathbb{R}^2 by the canonical bijection $\mathbb{R}^2 \to \mathbb{C}$). In a complex framework, we can extend the definitions of the layer potential operators and of the Neumann-Poincaré operator for the transmission problem for the Laplacian across a Lipschitz boundary, given in [1] for instance:

$$\begin{split} \tilde{\mathcal{D}}_{\partial\Omega} &: H^{\frac{1}{2}}_{\mathbb{C}}(\partial\Omega) \longrightarrow H^{1}_{\mathbb{C}}(\mathbb{C} \backslash \partial\Omega) \\ f_{R} &+ if_{I} \longmapsto \tilde{\mathcal{D}}_{\partial\Omega} f_{R} + i\tilde{\mathcal{D}}_{\partial\Omega} f_{I} \end{split}$$

for the double layer potential,

$$\begin{split} \tilde{\mathcal{S}}_{\partial\Omega} &: H_{\mathbb{C}}^{-\frac{1}{2}}(\partial\Omega) \longrightarrow H_{\mathbb{C}}^{1}(\mathbb{C} \setminus \partial\Omega) \\ g_{R} &+ ig_{I} \longmapsto \tilde{\mathcal{S}}_{\partial\Omega} f_{R} + i\tilde{\mathcal{S}}_{\partial\Omega} f_{I} \end{split}$$

for the single layer potential, and

$$\tilde{\mathcal{K}}_{\partial\Omega} : H^{\frac{1}{2}}_{\mathbb{C}}(\partial\Omega) \longrightarrow H^{\frac{1}{2}}_{\mathbb{C}}(\partial\Omega)
f_R + if_I \longmapsto \tilde{\mathcal{K}}_{\partial\Omega}f_R + i\tilde{\mathcal{K}}_{\partial\Omega}f_I,$$

for the Neumann-Poincaré operator. The definitions of those complex-valued operators display their algebraic decompositions.

Let $f \in H^{\frac{1}{2}}_{\mathbb{C}}(\partial\Omega)$. The Cauchy integral of f over $\partial\Omega$ counter-clockwise is defined by:

$$\Phi_{\partial\Omega}f(z) := \frac{1}{2i\pi} \int_{\partial\Omega} \frac{f(\zeta)}{\zeta - z} \,\mathrm{d}\zeta, \quad z \in \mathbb{C}.$$

It is holomorphic on $\mathbb{C}\setminus\partial\Omega$ [79, Section 15]. Moreover, it is connected to the Neumann-Poincaré operator in the following way [63, Section 6]:

$$\tilde{\mathcal{K}}_{\partial\Omega}f(z) = \int_{\partial\Omega} f(\zeta) \operatorname{Re}\left(\frac{\mathrm{d}\zeta}{2i\pi(\zeta-z)}\right), \quad z \in \partial\Omega.$$
(5.1)

If $f_R \in H^{\frac{1}{2}}(\partial \Omega)$ (\mathbb{R} -valued), then (5.1) yields:

$$\widetilde{\mathcal{K}}_{\partial\Omega}f_R(z) = \operatorname{Re}(\Phi_{\partial\Omega}f_R(z)), \quad z \in \partial\Omega.$$
(5.2)

Since $\Phi_{\partial\Omega} f_R$ is holomorphic on $\mathbb{C} \setminus \partial\Omega$, its real part is harmonic and by (5.2), $\Phi_{\partial\Omega} f_R$ can be written as:

$$\Phi_{\partial\Omega}f_R(z) = -\tilde{\mathcal{D}}_{\partial\Omega}f_R(z) + ih_R(z), \quad z \in \mathbb{C},$$
(5.3)

where h_R is the harmonic conjugate of $-\tilde{\mathcal{D}}_{\partial\Omega}f_R$. By the Plemelj formulae [79, Section 17], it holds $[[\operatorname{Tr} \Phi_{\partial\Omega}f_R]] = f_R$, which implies h_R (which is harmonic) can be expressed as a single layer potential.

If $f_I \in iH^{\frac{1}{2}}(\partial\Omega)$ (*i* \mathbb{R} -valued), then (5.1) yields:

$$\tilde{\mathcal{K}}_{\partial\Omega}(if_I)(z) = \operatorname{Im}(\Phi_{\partial\Omega}(if_I)(z)), \quad z \in \partial\Omega.$$
(5.4)

In the same way as before, $\Phi_{\partial\Omega}(if_I)$ can be written as:

$$\Phi_{\partial\Omega}(if_I)(z) = \tilde{\mathcal{S}}_{\partial\Omega}g_I(z) - i\tilde{\mathcal{D}}_{\partial\Omega}(if_I)(z), \quad z \in \mathbb{C},$$

where $\tilde{\mathcal{S}}_{\partial\Omega}g_I$ is the anti-harmonic conjugate of $-\tilde{\mathcal{D}}_{\partial\Omega}(if_I)$.

Those considerations yield the following theorem.

Theorem 5.2. Let Ω be a Lipschitz domain of \mathbb{C} . Let $f \in H^{\frac{1}{2}}_{\mathbb{C}}(\partial\Omega)$.

- 1. If f is real-valued, then $\operatorname{Re}(\Phi_{\partial\Omega}f)$ is a double layer potential and $\operatorname{Im}(\Phi_{\partial\Omega}f)$ is a single layer potential;
- 2. If f is purely imaginary-valued, then $\operatorname{Re}(\Phi_{\partial\Omega}f)$ is a single layer potential and $\operatorname{Im}(\Phi_{\partial\Omega}f)$ is a double layer potential.

5.2 Non-Lipschitz case

Generalizing a singular integral operator such as the Hilbert transform to twosided extension domains can be rather tricky, due for instance to the lack of a specified boundary measure. It is often more fruitful to rely on properties of the operator in question to define it from another angle, as it was done in Subsection 4.1. In the case of the Hilbert transform, we wish to use its connection with the Neumann-Poincaré operator given by equation (5.1), and the consequent Theorem 5.2 to give a definition on irregular domains. However, the Plemelj formulae [79, Section 17], which allow to identify the harmonic conjugate in equation (5.3) as a single layer potential in the Lipschitz case, are proved on Lipschitz boundaries, relying on geometrical considerations. Therefore, at this point, we can give two definitions of the Cauchy integral, one as a holomorphic function using the harmonic conjugate:

Definition 5.3. (Cauchy integral on two-sided extension domains of \mathbb{C} , version 1) Let Ω be a two-sided extension domain of \mathbb{C} . For $f = f_R + if_I \in \mathcal{B}_{\mathbb{C}}(\partial\Omega)$, we define the Cauchy integral on $\partial\Omega$ as:

$$\begin{cases} \Phi^{1}_{\partial\Omega}f_{R} := -\tilde{\mathcal{D}}_{\partial\Omega}f_{R} + ih_{R}, \\ \Phi^{1}_{\partial\Omega}(if_{I}) = \bar{h}_{I} - i\tilde{\mathcal{D}}_{\partial\Omega}(if_{I}), \end{cases}$$

where h_R is the harmonic conjugate of $-\tilde{\mathcal{D}}_{\partial\Omega}f_R$ and \bar{h}_I is the anti-harmonic conjugate of $-\tilde{\mathcal{D}}_{\partial\Omega}(if_I)$.

The other definition is as the sum of layer potentials:

Definition 5.4. (Cauchy integral on two-sided extension domains of \mathbb{C} , version 2) Let Ω be a two-sided extension domain of \mathbb{C} . For $f = f_R + if_I \in \mathcal{B}_{\mathbb{C}}(\partial\Omega)$, we define the Cauchy integral on $\partial\Omega$ as:

$$\begin{cases} \Phi_{\partial\Omega}^2 f_R := -\tilde{\mathcal{D}}_{\partial\Omega} f_R + i\tilde{\mathcal{S}}_{\partial\Omega} g_R, \\ \Phi_{\partial\Omega}^2 (if_I) = \tilde{\mathcal{S}}_{\partial\Omega} g_I - i\tilde{\mathcal{D}}_{\partial\Omega} (if_I), \end{cases}$$

where $\tilde{S}_{\partial\Omega}g_R$ is its single layer potential part (Proposition B.7) of the harmonic conjugate of $-\tilde{\mathcal{D}}_{\partial\Omega}f_R$ and $\tilde{S}_{\partial\Omega}g_I$ is the single layer potential part of the anti-harmonic conjugate of $-\tilde{\mathcal{D}}_{\partial\Omega}(if_I)$.

One conjecture which can be made at this point is that, as in the regular case, those definitions coincide. One way to prove it is to establish the convergence of the layer potential operators as Lipschitz domains converge to an extension domain. In Appendix C.1, we prove that any bounded domain can be approximated by a sequence of Lipschitz domains, from both the inside and the outside. In Appendix C.2, we prove, under some hypotheses, the convergence of the harmonic extension operators defined in Subsection 3.1 when approximating extension domains. What remains to be done is to use an energy minimization argument to reconstruct the layer potential operators from the harmonic extensions, and establish the convergence of the projectors related to that energy minimization to deduce the convergence of the layer potential operators.

A Bounded problems on extension domains

A.1 Harmonic extension operators

By the Riesz representation theorem, the Neumann boundary value problem (3.2) for k = 1 on a one-sided extension domain Ω of \mathbb{R}^n is weakly wellposed, in the sense of the variational formulation on $H^1(\Omega)$ (see also [5]):

$$\forall v \in H^1(\Omega), \quad \langle u, v \rangle_{H^1(\Omega)} = \langle g, \operatorname{Tr}_i v \rangle_{\mathcal{B}', \mathcal{B}}$$
(A.1)

with the *a priori* estimate:

$$\|u\|_{H^1(\Omega)} \le \|g\|_{\mathcal{B}'(\partial\Omega)}.\tag{A.2}$$

The continuity constant in (A.2) is equal to 1, since $\|\operatorname{Tr}_i\|_{\mathcal{L}(H^1(\Omega), \mathcal{B}(\partial\Omega))} = 1$.

Let us define the space of weak solutions to the Neumann boundary value problem for $(-\Delta + 1)$:

$$X_{(-\Delta+1)_N} := \{ u \in H^1(\Omega) \mid \exists g \in \mathcal{B}'(\partial\Omega) \text{ s.t. } (A.1) \text{ holds} \}$$

endowed with the standard H^1 norm. Given the continuity of the solution with respect to $g \in \mathcal{B}'(\partial\Omega)$ (by (A.2)), the space $(X_{(-\Delta+1)_N}, \|\cdot\|_{H^1(\Omega)})$ is closed in $H^1(\Omega)$, *i.e.* it is the Hilbert space. Since any 1-harmonic function is a solution to problem (A.1) associated to its own interior normal derivative, it holds $X_{(-\Delta+1)_N} = V_1(\Omega)$, where $V_1(\Omega)$ is defined by (2.3), and coincides with Ker(Tr_i)^{\perp} by (2.4).

By the well-posedness of (A.1) with the continuity estimate (A.2), the operator of the inverse problem $\mathcal{B}'(\partial\Omega) \to X_{(-\Delta+1)_N}$ is linear and bounded; it is called the Neumann 1-harmonic extension operator on Ω .

The case k = 0 is more tricky since 0 is an eigenvalue of the Laplacian with Neumann boundary condition: any constant is a solution to the homogeneous problem. For Problem (3.2) with k = 0 to be well-posed, we need to work on a space which is similar to $H^1(\Omega)$, but in which the only constant is null:

$$H^{1}_{\langle 0 \rangle}(\Omega) := \left\{ v \in H^{1}(\Omega) \ \middle| \ \int_{\Omega} v \, \mathrm{d}x = 0 \right\}.$$
(A.3)

Note that the constants have been removed in a way which does not involve any boundary measure. That space is endowed with the norm $\|\nabla \cdot\|_{L^2(\Omega)^n}$: since Ω is bounded and $\{0\}$ is closed, then $H^1_{\langle 0 \rangle}(\Omega)$ is a closed subspace of $H^1(\Omega)$, *i.e.*, $(H^1_{\langle 0 \rangle}(\Omega), \|\cdot\|_{H^1(\Omega)})$ is a Hilbert space. As Poincaré's inequality holds on $H^1_{\langle 0 \rangle}(\Omega)$ by [43], [54, Corollary 3.4], the semi-norm $\|\nabla \cdot\|_{L^2(\Omega)^n}$ on $H^1(\Omega)$ becomes a norm on $H^1_{\langle 0 \rangle}(\Omega)$ equivalent to $\|\cdot\|_{H^1(\Omega)}$. Therefore, we consider the Hilbert space $H^1_{\langle 0 \rangle}(\Omega)$, endowed with the inner product:

$$\langle u, v \rangle_{H^1_{\langle 0 \rangle}(\Omega)} := \int_{\Omega} \nabla u \cdot \nabla v \, \mathrm{d}x.$$

We prove the restriction of the trace $\operatorname{Tr}_i : H^1_{\langle 0 \rangle}(\Omega) \to \mathcal{B}(\partial \Omega)$ is not onto and introduce the range of that restriction:

$$\mathcal{B}_0(\partial\Omega) := \operatorname{Tr}_i \left(H^1_{\langle 0 \rangle}(\Omega) \right). \tag{A.4}$$

Lemma A.1. Let Ω be a one-sided extension domain of \mathbb{R}^n . Then, it holds:

$$\mathcal{B}(\partial\Omega) = \mathcal{B}_0(\partial\Omega) \oplus \mathbb{R}.$$

Proof. By [38, Theorem 7], for every $f \in \mathcal{B}(\partial\Omega)$, there exists a unique solution $u \in H^1(\Omega)$ to (3.1) for k = 0. Moreover, there exists a unique $u_0 \in H^1_{\langle 0 \rangle}(\Omega)$ such that $\Delta u = 0$ weakly on Ω and $\frac{\partial_i}{\partial \nu} u = \frac{\partial_i}{\partial \nu} u_0$. Since Ω is connected, Green's formula for the harmonic function $u - u_0$ implies there exists $c \in \mathbb{R}$ such that $u - u_0 = c$, hence f is uniquely decomposed as $f = \operatorname{Tr}_i u_0 + c$, with $\operatorname{Tr}_i u_0 \in \mathcal{B}_0(\partial\Omega)$.

By analogy with Theorem 2.9, we state the following theorem.

Theorem A.2. The space $\mathcal{B}_0(\partial\Omega)$ is a Hilbert space endowed with the norm:

$$\|f\|_{\mathcal{B}_0(\partial\Omega)} := \min\left\{\|v\|_{H^1_{\langle 0\rangle}(\Omega)} \mid \operatorname{Tr}_i v = f \text{ on } \partial\Omega\right\}.$$

That minimum is reached for a unique value of $u \in H^1_{\langle 0 \rangle}(\Omega)$, which is the unique element such that $\Delta u = 0$ on Ω and $\operatorname{Tr}_i u = f$ on $\partial \Omega$. Consequently, the trace operator $\operatorname{Tr}_i : V_{\langle 0 \rangle}(\Omega) \to \mathcal{B}_0(\partial \Omega)$ defines an isometry between the subspace of harmonic functions in $H^1_{\langle 0 \rangle}(\Omega)$:

$$V_{\langle 0 \rangle}(\Omega) := \left\{ u \in H^1_{\langle 0 \rangle}(\Omega) \mid \Delta u = 0 \text{ weakly on } \Omega \right\},\tag{A.5}$$

endowed with the norm $\|\nabla \cdot\|_{L^2(\Omega)^n}$, and $\mathcal{B}_0(\partial\Omega)$. In addition, the adjoint of the trace operator $\operatorname{Tr}_i^* : \mathcal{B}'_0(\partial\Omega) \to V'_{\langle 0 \rangle}(\Omega)$, and the inverse operators $\operatorname{Tr}_i^{-1} : \mathcal{B}_0(\partial\Omega) \to V_{\langle 0 \rangle}(\Omega)$ and $(\operatorname{Tr}_i^*)^{-1} : V'_{\langle 0 \rangle}(\Omega) \to \mathcal{B}'_0(\partial\Omega)$ are isometric as well. In particular, it holds:

$$\forall g \in \mathcal{B}'_0(\partial\Omega), \quad \|g\|_{\mathcal{B}'_0(\partial\Omega)} = \|\mathrm{Tr}_i^* g\|_{H^1(\Omega)'}, \tag{A.6}$$

$$\forall u \in V_{\langle 0 \rangle}(\Omega), \quad \|u\|_{H^1_{\langle 0 \rangle}(\Omega)} = \|\operatorname{Tr}_i u\|_{\mathcal{B}_0(\partial\Omega)}. \tag{A.7}$$

The following proposition is a direct consequence of the theorem above.

Proposition A.3. For every $f \in \mathcal{B}_0(\partial \Omega)$, it holds:

$$||f||_{\mathcal{B}(\partial\Omega)} \le ||f||_{\mathcal{B}_0(\partial\Omega)}.$$

Moreover, the equality is strict whenever $f \neq 0$. In particular, the injection of $\mathcal{B}_0(\partial \Omega)$ into $\mathcal{B}(\partial \Omega)$ is continuous.

Proof. Let us denote by u_1 the only element of $V_1(\Omega)$ such that $\operatorname{Tr}_i u_1 = f$, and by $u_{\langle 0 \rangle}$ the only element of $V_{\langle 0 \rangle}(\Omega)$ such that $\operatorname{Tr}_i u_{\langle 0 \rangle} = f$. Then, by definition of the norm on $\mathcal{B}(\partial \Omega)$ and Theorems 2.9 and A.2, we have:

$$||u_1||_{H^1(\Omega)} = ||f||_{\mathcal{B}(\partial\Omega)} \le ||u_{\langle 0 \rangle}||_{H^1_{\langle 0 \rangle}(\Omega)} = ||f||_{\mathcal{B}_0(\partial\Omega)}.$$

Furthermore, the inequality becomes an equality if and only if $u_1 = u_{\langle 0 \rangle}$, which implies $u_1 = 0$ and f = 0.

In the theorems defining the interior normal derivative in the weak sense [67], the surjectivity of the trace operator is a key element. Therefore, for every $u \in H^1_{\langle 0 \rangle, \Delta}(\Omega) := \{ u \in H^1_{\langle 0 \rangle}(\Omega) \mid \Delta u \in L^2(\Omega) \}$, it appears natural to define its normal derivative as an element of the dual space $\mathcal{B}'_0(\partial\Omega)$, in the same way as it was done in Definitions 2.12 and 2.13.

Definition A.4. (Weak interior normal derivative in $\mathcal{B}'_0(\partial\Omega)$) The weak interior normal derivative in $\mathcal{B}'_0(\partial\Omega)$ of $u \in H^1_{\langle 0 \rangle, \Delta}(\Omega)$ is the element $\psi \in \mathcal{B}'_0(\partial\Omega)$ such that for all $v \in H^1_{\langle 0 \rangle}(\Omega)$, it holds:

$$\langle \psi, \operatorname{Tr}_i v \rangle_{\mathcal{B}'_0, \mathcal{B}_0} = \int_{\Omega} (\Delta u) v \, \mathrm{d}x + \int_{\Omega} \nabla u \cdot \nabla v \, \mathrm{d}x,$$

and we denote $\frac{\partial_i^{\mathrm{o}} u}{\partial \nu} = \psi.$

Obviously, for every $u \in H^1_{\langle 0 \rangle, \Delta}(\Omega)$, it holds $u \in H^1_{\Delta}(\Omega)$, defined by Equation (2.11). Therefore, we could also use Definition 2.12 to define $\frac{\partial_i u}{\partial \nu} \in \mathcal{B}'(\partial\Omega)$. However, that choice could be misleading: since Problem (3.2) for k = 0 is set on $H^1_{\langle 0 \rangle}(\Omega)$, only the restriction $\frac{\partial_i u}{\partial \nu}|_{\mathcal{B}_0} = \frac{\partial_i^{\circ} u}{\partial \nu} : \mathcal{B}_0(\partial\Omega, \mu) \to \mathbb{R}$ intervenes in the weak formulation of the problem (see (A.9) below). Therefore, the identification $\frac{\partial_i u}{\partial \nu} = g$ is only valid when we apply those functions to elements of $\mathcal{B}_0(\partial\Omega)$, which means the Neumann boundary condition of Problem (3.2) for k = 0 must be understood in the space $\mathcal{B}'_0(\partial\Omega)$.

One might nonetheless wish to understand that condition in $\mathcal{B}'(\partial\Omega)$. In that case, since it holds (Lemma A.1):

$$\mathcal{B}(\partial\Omega) = \mathcal{B}_0(\partial\Omega) \oplus \mathbb{R},$$

we only need to set the value of $\langle \frac{\partial_i^{\circ u}}{\partial \nu}, 1 \rangle_{\mathcal{B}', \mathcal{B}}$ when u is (weakly) harmonic. By Green's formula, it must hold:

$$\left\langle \frac{\partial_i^{\mathrm{o}} u}{\partial \nu}, 1 \right\rangle_{\mathcal{B}', \mathcal{B}} = \int_{\Omega} \Delta u \, \mathrm{d}x = 0.$$

Therefore, setting 0 on the constants is the only possible extension of a Neumann boundary condition $g \in \mathcal{B}'_0(\partial\Omega)$ of problem (3.2) for k = 0 to $\mathcal{B}'(\partial\Omega)$. The following theorem states that considering a Neumann boundary condition in $\mathcal{B}'_0(\partial\Omega)$ or extended to an element of $\mathcal{B}'(\partial\Omega)$ in such a way is equivalent.

Theorem A.5. Let us define the set:

$$\mathcal{B}'_{0,ext}(\partial\Omega) := \{ g \in \mathcal{B}'(\partial\Omega) \mid \langle g, 1 \rangle_{\mathcal{B}',\mathcal{B}} = 0 \}$$
(A.8)

of extensions of the elements of $\mathcal{B}'_0(\partial\Omega)$ to $\mathcal{B}'(\partial\Omega)$. Then, the restriction operator:

$$\cdot |_{\mathcal{B}_0} : \mathcal{B}'_{0,ext}(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega)$$
$$g \mapsto g|_{\mathcal{B}_0}$$

defines an isometry between $\mathcal{B}'_{0,ext}(\partial\Omega)$ and $\mathcal{B}'_0(\partial\Omega)$.

Proof. Let $g \in \mathcal{B}'_{0,ext}(\partial\Omega)$. Since $\mathcal{B}(\partial\Omega) = \mathcal{B}_0(\partial\Omega) \oplus \mathbb{R}$, it holds:

$$\begin{split} \|g\|_{\mathcal{B}'} &= \sup_{\|f\|_{\mathcal{B}}=1} \left| \langle g, f \rangle_{\mathcal{B}', \mathcal{B}} \right| \\ &= \max \left(\sup_{\|f_0\|_{\mathcal{B}_0}=1} \left| \langle g|_{\mathcal{B}_0}, f_0 \rangle_{\mathcal{B}'_0, \mathcal{B}_0} \right|, \quad \underbrace{\left| \langle g, c \rangle_{\mathcal{B}', \mathcal{B}} \right|}_{=0} \right) \quad \text{for } c \in \mathbb{R}, \ \|c\|_{\mathcal{B}} = 1 \\ &= \sup_{\|f_0\|_{\mathcal{B}_0}=1} \left| \langle g|_{\mathcal{B}_0}, f_0 \rangle_{\mathcal{B}'_0, \mathcal{B}_0} \right| \\ &= \|g\|_{\mathcal{B}_0}\|_{\mathcal{B}'_0}. \end{split}$$

Therefore $\cdot|_{\mathcal{B}_0}$ defines an isometry between $\mathcal{B}'_{0,ext}(\partial\Omega)$ and $\mathcal{B}'_0(\partial\Omega)$.

Problem (3.2) for k = 0 is understood in the following variational sense:

$$\forall v \in H^1_{\langle 0 \rangle}(\Omega), \quad \int_{\Omega} \nabla u \cdot \nabla v \, \mathrm{d}x = \langle g, \mathrm{Tr}_i v \rangle_{\mathcal{B}'_0, \mathcal{B}_0}, \tag{A.9}$$

with $g \in \mathcal{B}'_0(\partial \Omega)$ and $\operatorname{Tr}_i : H^1_{\langle 0 \rangle}(\Omega) \to \mathcal{B}_0(\partial \Omega)$.

As for the operator $(-\Delta + 1)$, we can use the Riesz representation theorem and the isometric properties from Theorem A.2, to obtain the following result.

Proposition A.6. Let Ω be a one-sided extension domain of \mathbb{R}^n . For all $g \in \mathcal{B}'_0(\partial \Omega)$ there exists a unique solution $u \in H^1_{\langle 0 \rangle}(\Omega)$ of the variational formulation (A.9). Moreover,

$$\|\nabla u\|_{L^2(\Omega)^n} \le \|g\|_{\mathcal{B}'(\partial\Omega)}.$$

We provide the analogous construction for the Neumann value problem (3.2) associated to $-\Delta$ in the class of functions with the mean value zero on the

boundary. The problem is weakly well-posed by Proposition A.6 on $H^1_{\langle 0 \rangle}(\Omega)$ (see (A.3)). This time, we set

$$X_{(-\Delta)_N} := \{ u \in H^1_{\langle 0 \rangle}(\Omega) \mid \exists g \in \mathcal{B}'_0(\partial \Omega) \text{ s.t. } (A.9) \text{ holds} \},$$
(A.10)

endowed with the norm $||u||_{H^{1}_{\langle 0 \rangle}(\Omega)} = ||\nabla u||_{L^{2}(\Omega)^{n}}$. Then $(X_{(-\Delta)_{N}}, ||\cdot||_{H^{1}_{\langle 0 \rangle}(\Omega)})$ is a Hilbert subspace of $H^{1}(\Omega)$. Since any element of $V_{\langle 0 \rangle}(\Omega)$ (see (A.5) for the definition) is a solution to problem (A.9) associated to its own interior normal derivative, it holds $X_{(-\Delta)_{N}} = V_{\langle 0 \rangle}(\Omega)$, and $(V_{\langle 0 \rangle}(\Omega), ||\cdot||_{H^{1}_{\langle 0 \rangle}(\Omega)})$ is a Hilbert space.

Definition A.7. (Neumann (1-)harmonic extension operator) For any one-sided extension domain Ω of \mathbb{R}^n , let us define:

(i) the Neumann 1-harmonic extension operator on Ω by:

$$S_{\Omega} : \mathcal{B}'(\partial\Omega) \to S_{\Omega}(\mathcal{B}'(\partial\Omega)) = V_1(\Omega) \subset H^1(\Omega)$$
$$g \mapsto u_g$$

where u_g is the unique solution to the variational formulation (A.1) corresponding to $g \in \mathcal{B}'(\partial\Omega)$:

$$\forall v \in H^1(\Omega), \quad \langle \mathcal{S}_\Omega g, v \rangle_{H^1(\Omega)} = \langle g, \operatorname{Tr}_i v \rangle_{\mathcal{B}', \mathcal{B}}.$$
(A.11)

(ii) the Neumann harmonic extension operator on Ω by:

$$\tilde{S}_{\Omega} : \mathcal{B}'_{0}(\partial\Omega) \to \tilde{S}_{\Omega}(\mathcal{B}'_{0}(\partial\Omega)) = V_{\langle 0 \rangle}(\Omega) \subset H^{1}_{\langle 0 \rangle}(\Omega)$$
$$g \mapsto u_{q}$$

where u_g is the unique solution to the variational formulation (A.9) corresponding to $g \in \mathcal{B}'_0(\partial\Omega)$:

$$\forall v \in H^1_{\langle 0 \rangle}(\Omega), \quad \int_{\Omega} \nabla(\tilde{\mathcal{S}}_{\Omega}g) \cdot \nabla v \, \mathrm{d}x = \langle g, \mathrm{Tr}_i v \rangle_{\mathcal{B}'_0, \mathcal{B}_0}, \tag{A.12}$$

As it was shown previously, from the well-posedness of the Neumann value problems for $(-\Delta+1)$ and $-\Delta$, we can also deduce the Neumann (1-)harmonic extension operators are isometric.

Theorem A.8. If Ω is a one-sided extension domain of \mathbb{R}^n , then

(i) the Neumann 1-harmonic extension operator, $S_{\Omega} : \mathcal{B}'(\partial \Omega) \to V_1(\Omega)$, is an isometry:

$$\forall g \in \mathcal{B}'(\partial\Omega), \quad \|\mathcal{S}_{\Omega}g\|_{H^1(\Omega)} = \|g\|_{\mathcal{B}'(\partial\Omega)}, \tag{A.13}$$

and thus is linear and continuous;

(ii) the Neumann harmonic extension operator, $\tilde{S}_{\Omega} : \mathcal{B}'_{0}(\partial\Omega) \to V_{\langle 0 \rangle}(\Omega)$, is an isometry:

 $\forall g \in \mathcal{B}'_0(\partial\Omega), \quad \|\tilde{\mathcal{S}}_{\Omega}g\|_{H^1_{(0)}(\Omega)} = \|g\|_{\mathcal{B}'_0(\partial\Omega)}, \tag{A.14}$

and thus is linear and continuous.

Proof. This is a direct consequence of the well-posedness of the variational formulations and Theorems 2.9 and A.2 respectively. \Box

Remark A.9. Since $S_{\Omega} : \mathcal{B}'(\partial\Omega) \to V_1(\Omega)$ and $\tilde{S}_{\Omega} : \mathcal{B}'(\partial\Omega) \to V_{\langle 0 \rangle}(\Omega)$ are linear bijections between infinite dimensional Hilbert spaces, they cannot be compact operators. However, since Ω is a Sobolev extension domain, the embedding $H^1(\Omega) \subset L^2(\Omega)$ is compact and we can deduce the operators $S_{\Omega} : \mathcal{B}'(\partial\Omega) \to L^2(\Omega)$ and $\tilde{S}_{\Omega} : \mathcal{B}'_0(\partial\Omega) \to L^2(\Omega)$ are compact.

The isometric properties also hold for the trace of the Neumann (1-)harmonic extension operators, by analogy with [5], as a composition of two isometries:

Theorem A.10. Let Ω be a one-sided extension domain of \mathbb{R}^n . Then, the trace of the Neumann (1-)harmonic extension operators on Ω :

- (i) $S|_{\partial\Omega} := \operatorname{Tr}_i \circ S_{\Omega} : g \mapsto \operatorname{Tr}_i u_g$ where u_g is the solution to (A.1) associated to g, defines an isometry from $\mathcal{B}'(\partial\Omega)$ to $\mathcal{B}(\partial\Omega)$ which coincides with its adjoint. In particular, the bilinear form defined by $(g,h) \mapsto \langle g, S_{\Omega} |_{\partial\Omega} h \rangle_{\mathcal{B}',\mathcal{B}}$ is equal to the usual inner product on $\mathcal{B}'(\partial\Omega)$.
- (ii) $\tilde{S}|_{\partial\Omega} := \operatorname{Tr}_i \circ \tilde{S}_{\Omega}$, mapping g to $\operatorname{Tr}_i u_g$ where u_g is the solution to (A.9) associated to $g \in \mathcal{B}'_0(\partial\Omega)$, defines an isometry from $\mathcal{B}'_0(\partial\Omega)$ to $\mathcal{B}_0(\partial\Omega)$ and coincides with its adjoint.

Proof. Given its definition, $S|_{\partial\Omega}$ is a linear operator. For $g, h \in \mathcal{B}'(\partial\Omega)$, it holds:

$$\begin{aligned} \langle g, \, \mathbb{S}|_{\partial\Omega} \, h \rangle_{\mathcal{B}',\mathcal{B}} &= \langle g, \, \mathrm{Tr}_i \, u_h \rangle_{\mathcal{B}',\mathcal{B}} \\ &= \langle u_g, \, u_h \rangle_{H^1(\Omega)} \\ &= \langle h, \, \mathbb{S}|_{\partial\Omega} \, g \rangle_{\mathcal{B}',\mathcal{B}} \\ &= \langle \mathbb{S}|_{\partial\Omega} \, g, \, h \rangle_{\mathcal{B}'',\mathcal{B}'}, \end{aligned}$$

since $\mathcal{B}(\partial\Omega)$ is a Hilbert space and can be identified to its bidual $\mathcal{B}''(\partial\Omega)$, where u_g and u_h are the unique solutions of (A.1) respectively associated to g and h. Hence $\mathcal{S}|_{\partial\Omega}$ coincides with its adjoint.

Furthermore, by Theorem A.8, for $g \in \mathcal{B}'(\partial \Omega)$:

$$\langle g, \mathfrak{S}|_{\partial\Omega} g \rangle_{\mathcal{B}',\mathcal{B}} = \|u_g\|^2_{H^1(\Omega)} = \|g\|^2_{\mathcal{B}'(\partial\Omega)},$$

Therefore, $(g, h) \mapsto \langle g, S|_{\partial\Omega} h \rangle$ is equal to the usual inner product on $\mathcal{B}'(\partial\Omega)$. Hence, as the composition of two isometries, $S|_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to \mathcal{B}(\partial\Omega)$ is an isometry itself.

In the same way, we prove (ii).

As for the Neumann (1-)harmonic extensions, we use the well-posedness of the corresponding Dirichlet boundary problems and define the Dirichlet (1-)extension operators.

Definition A.11. (Dirichlet 1-harmonic extension operator) For any one-sided extension domain Ω of \mathbb{R}^n , let us define \mathcal{D}_{Ω} , the Dirichlet 1-harmonic extension operator on Ω by:

$$\mathcal{D}_{\Omega}: \mathcal{B}(\partial\Omega) \to \mathcal{D}_{\Omega}(\mathcal{B}(\partial\Omega)) = V_1(\Omega) \subset H^1(\Omega)$$
$$f \mapsto u^f$$

where u^{f} is the unique weak solution to the variational formulation (2.6) associated to f:

$$\forall v \in H^1_f(\Omega), \quad \langle \mathcal{D}_\Omega f, v \rangle_{H^1(\Omega)} = \left\langle \frac{\partial_i(\mathcal{D}_\Omega f)}{\partial \nu}, f \right\rangle_{\mathcal{B}', \mathcal{B}}.$$
 (A.15)

In Definition A.11, the space $\mathcal{D}_{\Omega}(\mathcal{B}(\partial\Omega)) = V_1(\Omega)$ is the space of weak solutions of the Dirichlet $(-\Delta+1)$. It is the same as in Definition A.7 considered as the space of weak solutions for the Neumann $(-\Delta+1)$.

For the Laplace equation with Dirichlet boundary conditions on a one-sided extension domain Ω of \mathbb{R}^n :

$$\begin{cases} -\Delta u = 0 & \text{on } \Omega, \\ \operatorname{Tr}_{i} u = f, \end{cases}$$
(A.16)

we directly use [38, Theorem 7]: for all $f \in \mathcal{B}(\partial\Omega)$, there exists a unique weak solution $u \in H^1(\Omega)$ with $\operatorname{Tr}_i u = f$, and it is such that:

$$\forall v \in H_0^1(\Omega), \quad \int_{\Omega} \nabla u \cdot \nabla v \, \mathrm{d}x = 0.$$
 (A.17)

Hence, it allows us to define the Dirichlet harmonic extension operator in the same way:

Definition A.12. (Dirichlet harmonic extension operator) For any onesided extension domain Ω of \mathbb{R}^n , let us define $\tilde{\mathbb{D}}$, the Dirichlet harmonic extension operator by:

$$\tilde{\mathcal{D}}_{\Omega}: \mathcal{B}_0(\partial\Omega) \to \tilde{\mathcal{D}}_{\Omega}(\mathcal{B}_0(\partial\Omega)) = V_{\langle 0 \rangle}(\Omega) \subset H^1(\Omega)$$
$$f \mapsto u^f$$

where u^f is the unique solution to the variational formulation (A.17) associated to $f \in \mathcal{B}_0(\partial \Omega)$.

As before, the operator $\tilde{\mathcal{D}}_{\Omega}$ is understood as mapping the Dirichlet boundary data to the associated weak solution, *i.e.* in Definition A.12 the space $V_{\langle 0 \rangle}(\Omega)$ is the space of the weak solutions for the Dirichlet $-\Delta$. As in the case of $(-\Delta + 1)$, it holds $\tilde{\mathcal{D}}_{\Omega}(\mathcal{B}_0(\partial \Omega)) = \tilde{\mathcal{S}}_{\Omega}(\mathcal{B}'_0(\partial \Omega))$ (equal to $V_{\langle 0 \rangle}(\Omega)$ for $-\Delta$).

We deduce the continuity and the isometric properties of those Dirichlet (1-)harmonic extension operators from the isometry of the corresponding trace operators, $\operatorname{Tr}_i : V_1(\Omega) \to \mathcal{B}(\partial\Omega)$ (see Theorem 2.9) and $\operatorname{Tr}_i : V_{\langle 0 \rangle}(\Omega) \to \mathcal{B}_0(\partial\Omega)$ respectively.

Theorem A.13. Let Ω is a one-sided extension domain of \mathbb{R}^n . Then, the Dirichlet (1-)harmonic extension operators $\mathcal{D}_{\Omega} : \mathcal{B}(\partial\Omega) \to V_1(\Omega)$ and $\tilde{\mathcal{D}}_{\Omega} : \mathcal{B}_0(\partial\Omega) \to V_{\langle 0 \rangle}(\Omega)$ are isometries:

$$\forall f \in \mathcal{B}(\partial\Omega), \quad \|f\|_{\mathcal{B}(\partial\Omega)} = \|\mathcal{D}_{\Omega}f\|_{H^{1}(\Omega)}, \\ \forall f \in \mathcal{B}_{0}(\partial\Omega), \quad \|f\|_{\mathcal{B}_{0}(\partial\Omega)} = \|\tilde{\mathcal{D}}_{\Omega}f\|_{H^{1}_{(\Omega)}(\Omega)}.$$
 (A.18)

In this sense, \mathcal{D}_{Ω} and $\tilde{\mathcal{D}}_{\Omega}$ are equal to $\operatorname{Tr}_{i}^{-1}$ (defined on the adequate spaces). Consequently, understood as $\mathcal{D}_{\Omega} : \mathcal{B}(\partial\Omega) \to H^{1}(\Omega)$ and $\tilde{\mathcal{D}}_{\Omega} : \mathcal{B}(\partial\Omega) \to H^{1}(\Omega)$, those operators are linear and continuous.

A.2 Poincaré-Steklov operator

As explained in Subsection 3.2, we generalize the notion of Poincaré-Steklov operator in the framework of one-sided extension domains.

Definition A.14. (Poincaré-Steklov operator) Let Ω be a one-sided extension domain of \mathbb{R}^n . Let $k \in \mathbb{R} \setminus \sigma(\Delta_D)$ and let $u \in H^1(\Omega)$ be the weak solution to:

$$\begin{cases} (-\Delta + k)u = 0 & \text{on } \Omega\\ \operatorname{Tr}_{i} u = f \in \mathcal{B}(\partial\Omega). \end{cases}$$
(A.19)

The linear mapping defined as:

$$\mathcal{A}_k : \mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$$
$$\operatorname{Tr}_i u \mapsto \left. \frac{\partial_i u}{\partial\nu} \right|_{\partial\Omega}$$

is called the Poincaré-Steklov (or Dirichlet-to-Neumann) operator associated to $(-\Delta + k)$.

Having defined the operator $\mathcal{A}_k : \mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ in Definition A.14 enables to look into the same properties regarding continuity and invertibility as the ones stated for the Poincaré-Steklov operator based on the Laplace equation (*i.e.* with k = 0; see [4] for the statement on *d*-sets and the general setting of [13, p. 5904] for Lipschitz domains and also [75, Theorem 4.10], still holding in our general case). Since $k \in \mathbb{R} \setminus \sigma(\Delta_D)$, for all $\lambda \in \mathbb{C}$, there exists a solution to the Dirichlet problem (A.19) if $f \in \mathcal{B}(\partial\Omega)$ satisfies:

$$\left\langle \frac{\partial_i v}{\partial \nu}, f \right\rangle_{\mathcal{B}', \mathcal{B}} = 0,$$

for every solution $v \in H_0^1(\Omega)$ of the homogeneous problem:

$$\begin{cases} -\Delta v = (\lambda - k)v \quad \text{on } \Omega, \\ \operatorname{Tr}_{i} v = 0. \end{cases}$$
(A.20)

If $\lambda = 0$, since k is not eigenvalues of the Dirichlet Laplacian Δ_D , it follows that the only solution to Problem (A.20) is v = 0. This ensures \mathcal{A}_k is well defined on all of $\mathcal{B}(\partial\Omega)$. If k is an eigenvalue of the Neumann Laplacian, then Ker $\mathcal{A}_k \neq \{0\}$.

Let us consider the adjoint operator $\mathcal{A}_k^* : \mathcal{B}(\partial \Omega) \to \mathcal{B}'(\partial \Omega)$ (since $\mathcal{B}(\partial \Omega) \simeq \mathcal{B}''(\partial \Omega)$). Let us show that it coincides with \mathcal{A}_k .

Calculating the difference between two Green formulae for $u, v \in H^1_{\Delta}(\Omega)$ (defined by Equation (2.11)), we find that:

$$\left\langle \frac{\partial_i u}{\partial \nu}, \operatorname{Tr}_i v \right\rangle_{\mathcal{B}, \mathcal{B}} - \left\langle \operatorname{Tr}_i u, \frac{\partial_i v}{\partial \nu} \right\rangle_{\mathcal{B}, \mathcal{B}'} = \int_{\Omega} (\Delta u) v \, \mathrm{d}x - \int_{\Omega} u(\Delta v) \, \mathrm{d}x.$$

If $u \in H^1(\Omega)$ is a weak solution to (A.19), then $\Delta u = ku \in L^2(\Omega)$ in the sense of distributions, hence $\Delta u \in L^2(\Omega)$. In other words, weak solutions to (A.19) necessarily belong to $H^1_{\Delta}(\Omega)$. Therefore, since u and v are solutions to (A.19):

$$\int_{\Omega} (\Delta u) v \, \mathrm{d}x - \int_{\Omega} u(\Delta v) \, \mathrm{d}x = k \int_{\Omega} uv \, \mathrm{d}x - k \int_{\Omega} uv \, \mathrm{d}x = 0,$$

and we directly obtain that for all $\phi := \operatorname{Tr}_i u \in \mathcal{B}(\partial\Omega)$ and $f := \operatorname{Tr}_i v \in \mathcal{B}(\partial\Omega)$:

$$\langle \mathcal{A}_k \phi, f \rangle_{\mathcal{B}', \mathcal{B}} = \langle \phi, \mathcal{A}_k f \rangle_{\mathcal{B}, \mathcal{B}'},$$

i.e. the operator \mathcal{A}_k is closed and coincides with its adjoint. Since $\mathcal{B}'(\partial\Omega)$ is a Banach space, by the closed graph theorem, the Poincaré-Steklov operator \mathcal{A}_k is continuous.

The following theorem summarizes the results above.

Theorem A.15. Let Ω be a one-sided extension domain of \mathbb{R}^n and let $k \in \mathbb{R} \setminus \sigma(\Delta_D)$. Then the Poincaré-Steklov operator $\mathcal{A}_k : \mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ from Definition A.14 is a linear bounded operator coinciding with its adjoint. It is such that Ker $\mathcal{A}_k = \{0\}$ if and only if $k \notin \sigma(\Delta_N)$.

Remark A.16. If the operator Poincaré-Steklov is considered for a closed space of weak solutions to $(-\Delta+k)$ on which the Neumann eigenvalue problem does not have k as an eigenvalue, then it becomes injective. For example, it happens for k = 0 if we only consider solution satisfying $\int_{\Omega} u \, dx = 0$.

Therefore, choosing spaces on which the trace operator is isometric, we can define the Poincaré-Steklov operator as an isometry in the cases k = 1 and k = 0.

Theorem A.17. Let Ω be a one-sided extension domain of \mathbb{R}^n . Then, the (interior) Poincaré-Steklov operator

$$\mathcal{A}_1 =: \mathcal{A} : \operatorname{Tr}_i u \in \mathcal{B}(\partial \Omega) \mapsto \left. \frac{\partial_i u}{\partial \nu} \right|_{\partial \Omega} \in \mathcal{B}'(\partial \Omega),$$

(resp. $\mathcal{A}_0 =: \tilde{\mathcal{A}} : \operatorname{Tr}_i u \in \mathcal{B}_0(\partial\Omega) \mapsto \frac{\partial_i u}{\partial \nu}\Big|_{\partial\Omega} \in \mathcal{B}'_0(\partial\Omega)$), associated to the weak solutions in $V_1(\Omega)$, (resp. in $V_{\langle 0 \rangle}(\Omega)$) to the problem

$$\begin{cases} (-\Delta + 1)u = 0 & \text{on } \Omega, \\ \operatorname{Tr} u = f \in \mathcal{B}(\partial\Omega), \end{cases}$$
(A.21)

(resp. of $-\Delta u = 0$ on Ω with $\operatorname{Tr} u = f \in \mathcal{B}_0(\partial \Omega)$), is an isometry.

For any $k \in \mathbb{R} \setminus (\sigma(\Delta_D) \cup \sigma(\Delta_N))$, the Poincaré-Steklov operator \mathcal{A}_k : $\mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ from Definition A.14 is bijective. For example, the operator \mathcal{A}_1 (which we denote by \mathcal{A}) is bijective. For such a k, we can define both the Neumann extension operator and the Dirichlet extension operator for the problem $(-\Delta + k)$ on Ω , as the operator reconstructing the solution to the problem from the Neumann or Dirichlet boundary conditions respectively. From that perspective, the action of the Poincaré-Steklov is a shift from a Dirichlet boundary condition to a Neumann boundary condition. Therefore, the (k-)harmonic extension operators and the Poincaré-Steklov operator are deeply connected. The following results, detailed for $k = 1 \notin \sigma(\Delta_N)$ and $k = 0 \in \sigma(\Delta_N)$, highlight that connection. For $k = 0 \in \sigma(\Delta_N)$, it is possible to make \mathcal{A}_0 (denoted by \mathcal{A}) bijective as well by working on subspaces in which constants have been removed.

Proposition A.18. Let Ω be a one-sided extension domain of \mathbb{R}^n . Then it holds:

- (i) $\mathcal{D}_{\Omega} = \mathcal{S}_{\Omega} \circ \mathcal{A}$ and $\tilde{\mathcal{D}}_{\Omega} = \tilde{\mathcal{S}}_{\Omega} \circ \tilde{\mathcal{A}}$,
- (*ii*) $\mathcal{A} = \frac{\partial_i}{\partial \nu} \circ \mathcal{D}_{\Omega}$ and $\tilde{\mathcal{A}} = \frac{\partial_i}{\partial \nu} \circ \tilde{\mathcal{D}}_{\Omega}$, (*ii*) $\mathcal{A} = \mathcal{S}_{\Omega}^{-1} \circ \mathcal{D}_{\Omega}$, and $\tilde{\mathcal{A}} = \tilde{\mathcal{S}}_{\Omega}^{-1} \circ \tilde{\mathcal{D}}_{\Omega}$,

where $S : \mathcal{B}'(\partial\Omega) \to V_1(\Omega), \ \mathcal{D} : \mathcal{B}(\partial\Omega) \to V_1(\Omega), \ \mathcal{A} : \mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ on the one hand, as well as $\tilde{S} : \mathcal{B}'_0(\partial\Omega) \to V_{\langle 0 \rangle}(\Omega), \ \tilde{D} : \mathcal{B}_0(\partial\Omega) \to V_{\langle 0 \rangle}(\Omega),$ $\tilde{\mathcal{A}}: \mathcal{B}_0(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega)$ on the other hand are isometries, and $\frac{\partial_i}{\partial\nu}$ is an isometry either from $V_1(\Omega)$ to $\mathcal{B}'(\partial\Omega)$ (when applied to \mathcal{D}), or from $V_{\langle 0 \rangle}(\Omega)$ to $\mathcal{B}'_0(\partial\Omega)$ (when applied to $\tilde{\mathbb{D}}$).

Proof. Let us prove it for the operators associated with $(-\Delta + 1)$ (the case of $-\Delta$ is analogous).

(i) Let $f \in \mathcal{B}(\partial\Omega)$. $\mathcal{A}f \in \mathcal{B}'(\partial\Omega)$ therefore $S_{\Omega} \circ \mathcal{A}f$ is a weak solution to $(-\Delta + 1)u = 0$ on $H^1(\Omega)$. In addition, it holds:

$$\operatorname{Tr}_{i}\left(\mathfrak{S}\circ\mathcal{A}f\right)=\mathcal{A}^{-1}\left(\frac{\partial_{i}}{\partial\nu}(\mathfrak{S}_{\Omega}\circ\mathcal{A}f)\right)=\mathcal{A}^{-1}(\mathcal{A}f)=f.$$

Therefore, $S_{\Omega} \circ \mathcal{A}f \in H^1(\Omega)$ is a weak solution to the transmission problem (2.6), hence $\mathcal{D}_{\Omega}f = S_{\Omega} \circ \mathcal{A}f$.

(*ii*) The range of the double layer potential \mathcal{D}_{Ω} is $V_1(\Omega) \subset H^1_{\Delta}(\Omega)$, therefore its interior normal derivative can be defined as an element of $\mathcal{B}'(\partial\Omega)$. Let $f \in \mathcal{B}(\partial\Omega)$. Then $u := \mathcal{D}_{\Omega}f$ is such that $\operatorname{Tr}_i u = f$ and $(-\Delta + 1)u = 0$ weakly on Ω . Therefore, it holds:

$$\mathcal{A}f = \mathcal{A}(\operatorname{Tr}_i u) = \frac{\partial_i u}{\partial \nu} = \frac{\partial_i}{\partial \nu} \circ \mathcal{D}_{\Omega}f.$$

(iii) Since $\mathcal{D}_{\Omega} = \operatorname{Tr}_{i}^{-1} : \mathcal{B}(\partial\Omega) \to V_{1}(\Omega) \text{ and } \mathcal{A}^{-1} = \operatorname{Tr}_{i} \circ \mathcal{S}_{\Omega}$ by Theorem A.10, we directly find $\mathcal{A} = (\operatorname{Tr}_{i} \circ \mathcal{S})^{-1} = \mathcal{S}_{\Omega}^{-1} \circ \mathcal{D}_{\Omega}$.

The relations described above are synthesized in Diagram (3.7). In the spirit of point *(ii)* of Proposition A.18, we use the Dirichlet extension operator to describe the Poincaré-Steklov operator $\tilde{\mathcal{A}} = \mathcal{A}_0$ in the variational sense. For all $f, \varphi \in \mathcal{B}(\partial\Omega)$, since $\operatorname{Tr}_i(\tilde{\mathcal{D}}_\Omega \varphi) = \varphi$, it holds, by Green's formula:

$$\langle \tilde{\mathcal{A}}f, \varphi \rangle_{\mathcal{B}', \mathcal{B}} = \left\langle \frac{\partial_i \tilde{\mathcal{D}}_{\Omega} f}{\partial \nu}, \varphi \right\rangle_{\mathcal{B}', \mathcal{B}} = \int_{\Omega} \nabla \tilde{\mathcal{D}}_{\Omega} f \cdot \nabla \tilde{\mathcal{D}}_{\Omega} \varphi \, \mathrm{d}x.$$
 (A.22)

From Proposition A.18, we can give an explicit formula for the inverse of the Poincaré-Steklov operator using the Neumann (1-)harmonic extension operators.

Corollary A.19. Let Ω be a one-sided extension domain of \mathbb{R}^n .

- (i) The Poincaré-Steklov operator $\mathcal{A} : \mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ for the 1-harmonic solutions is isometric with inverse isometry $\mathcal{A}^{-1} = \operatorname{Tr}_i \circ S_\Omega : \mathcal{B}'(\partial\Omega) \to \mathcal{B}(\partial\Omega).$
- (ii) The Poincaré-Steklov operator $\tilde{\mathcal{A}}$ associated to $-\Delta$ is isometric from $\mathcal{B}_0(\partial\Omega)$ to $\mathcal{B}'_0(\partial\Omega)$ with inverse isometry $\tilde{\mathcal{A}}^{-1} = \operatorname{Tr}_i \circ \tilde{\mathcal{S}}_{\Omega} : \mathcal{B}'_0(\partial\Omega) \to \mathcal{B}_0(\partial\Omega).$

B Transmission problems for extension domains

B.1 Layer potential operators

Let Ω be a two-sided extension domain of \mathbb{R}^n , and consider the transmission problem described in Subsection 4.1.

For the operator $(-\Delta+1)$. For $g \in \mathcal{B}'(\partial\Omega)$, let us consider the transmission problem (4.1) with k = 1, a jump in normal derivative g, and with no jump in trace: f = 0. Let us establish its variational formulation. On Ω , by Green's formula, it holds

$$\forall v \in H^1(\Omega), \quad \langle u, v \rangle_{H^1(\Omega)} = \left\langle \frac{\partial_i u}{\partial \nu}, \operatorname{Tr}_i v \right\rangle_{\mathcal{B}', \mathcal{B}}$$

On Ω^c , we have

$$\forall w \in H^1(\Omega^c), \quad \langle u, w \rangle_{H^1(\Omega^c)} = -\left\langle \frac{\partial_e u}{\partial \nu}, \operatorname{Tr}_e w \right\rangle_{\mathcal{B}', \mathcal{B}}$$

Hence, the following variational formulation holds (since $\lambda^{(n)}(\partial \Omega) = 0$, by definition of a two-sided extension domain – Definition 2.7):

$$\forall \varphi \in H^1_{\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n), \quad \langle u, \varphi \rangle_{H^1_{\partial\Omega}(\mathbb{R}^n)} = \langle g, \mathrm{Tr}\,\varphi \rangle_{\mathcal{B}',\mathcal{B}}, \tag{B.1}$$

where

$$H^{1}_{\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^{n}) := \left\{ v \in H^{1}_{\partial\Omega}(\mathbb{R}^{n}) \mid \llbracket \mathrm{Tr} \, v \rrbracket = 0 \right\},$$

and $\operatorname{Tr} \varphi := \operatorname{Tr}_i \varphi = \operatorname{Tr}_e \varphi$. Since $\llbracket \operatorname{Tr} \cdot \rrbracket$ is continuous on $H^1_{\partial\Omega}(\mathbb{R}^n)$, the space $H^1_{\partial\Omega,[0]_{\operatorname{Tr}}}(\mathbb{R}^n)$ is a Hilbert space endowed with the same inner product $\langle \cdot, \cdot \rangle_{H^1_{\partial\Omega}(\mathbb{R}^n)}$. Therefore, applying the Riesz representation theorem, we deduce the well-posedness of the variational formulation. This allows to define the single layer potential operator in the variational sense, as the operator mapping the jump in normal derivative to the associated 1-harmonic function on $\Omega \cup \Omega^c$ with no jump in trace across $\partial\Omega$.

Theorem B.1. (Single layer potential operator for $(-\Delta + 1)$) For any two-sided extension domain Ω of \mathbb{R}^n , let

$$\begin{aligned} \mathcal{S}_{\partial\Omega} &: \mathcal{B}'(\partial\Omega) \longrightarrow \mathcal{S}_{\partial\Omega}(\mathcal{B}'(\partial\Omega)) \subset H^1_{\partial\Omega}(\mathbb{R}^n) \\ g &= \left[\left[\frac{\partial u}{\partial \nu} \right] \right] \longmapsto u = \mathcal{S}_{\partial\Omega} g, \end{aligned}$$

be the linear operator uniquely defined by

$$\forall v \in H^1_{\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n), \quad \langle \mathcal{S}_{\partial\Omega} \, g, v \rangle_{H^1_{\partial\Omega}(\mathbb{R}^n)} = \langle g, \mathrm{Tr} \, v \rangle_{\mathcal{B}',\mathcal{B}}. \tag{B.2}$$

Then $S_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to H^1_{\partial\Omega}(\mathbb{R}^n)$, called the single layer potential operator for the transmission problem (4.1) with k = 1, is a linear bounded operator:

$$\|\mathcal{S}_{\partial\Omega} g\|_{H^1_{\partial\Omega}(\mathbb{R}^n)} \le C \|g\|_{\mathcal{B}'(\partial\Omega)} \tag{B.3}$$

for some constant $C = C(\Omega) > 0$. In addition, it can be expressed as a convolution with G, the fundamental solution to $(-\Delta + 1)$ on \mathbb{R}^n :

$$\forall g \in \mathcal{B}'(\partial\Omega), \quad \forall x \in \mathbb{R}^n, \quad \mathcal{S}_{\partial\Omega} g(x) = (G * \operatorname{Tr}^* g)(x),$$

where $\operatorname{Tr}^* : \mathcal{B}'(\partial\Omega) \to [\mathcal{S}_{\partial\Omega}(\mathcal{B}'(\partial\Omega))]'$ and the convolution is on \mathbb{R}^n . Therefore,

$$\forall x \in \mathbb{R}^n, \quad \mathcal{S}_{\partial\Omega} g(x) = \langle g, \operatorname{Tr} G(x - \cdot) \rangle_{\mathcal{B}', \mathcal{B}}.$$
 (B.4)

Remark B.2. It will be pointed out in the following proof that for all $g \in \mathcal{B}'(\partial\Omega)$, $\operatorname{Tr}^* g$ can be seen as a distribution with compact support. Therefore, the convolution with the fundamental solution is well-defined, at least in the sense of distributions. We will show it is actually an element of $H^1_{\partial\Omega}(\mathbb{R}^n)$.

Proof. The well-posedness of (B.1) is sufficient to ensure that $S_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to H^1_{\partial\Omega}(\mathbb{R}^n)$ is a linear bounded operator satisfying (B.3).

Since the interior and exterior traces of the solution to the transmission problem (4.1) with k = 1 and f = 0 are equal, $S_{\partial\Omega} g$ can be defined on \mathbb{R}^n , for $g \in \mathcal{B}'(\partial\Omega)$. $S_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to S_{\partial\Omega}(\mathcal{B}'(\partial\Omega))$ is continuously invertible (with inverse $[\![\frac{\partial}{\partial\nu}]\!]$), therefore $S_{\partial\Omega}(\mathcal{B}'(\partial\Omega))$ is a Hilbert space. Let us notice that for all $\phi \in [S_{\partial\Omega}(\mathcal{B}'(\partial\Omega))]'$, it holds $G * \phi \in S_{\partial\Omega}(\mathcal{B}'(\partial\Omega))$. It is a direct corollary (adapting [17, Lemma IV.5.14]) of the well-posedness on $S_{\partial\Omega}(\mathcal{B}'(\partial\Omega))$ of the following variational problem:

$$\forall v \in \mathcal{S}_{\partial\Omega}(\mathcal{B}'(\partial\Omega)), \quad \langle u, v \rangle_{H^1_{\partial\Omega}(\mathbb{R}^n)} = \langle \phi, v \rangle_{[\mathcal{S}_{\partial\Omega}(\mathcal{B}')]', \mathcal{S}_{\partial\Omega}(\mathcal{B}')}, \tag{B.5}$$

associated with the elliptic problem (4.1) with k = 1, f = 0, g = 0 and a source term equal to ϕ . Since $u \in S_{\partial\Omega}(\mathcal{B}'(\partial\Omega))$ is its unique weak solution, by definition of the fundamental solution, it holds $G * \phi = u \in S_{\partial\Omega}(\mathcal{B}'(\partial\Omega))$.

In our case, $\operatorname{Tr}^* g \in [\mathcal{S}_{\partial\Omega}(\mathcal{B}'(\partial\Omega))]'$, hence $G * \operatorname{Tr}^* g \in \mathcal{S}_{\partial\Omega}(\mathcal{B}'(\partial\Omega))$, which means that $G * \operatorname{Tr}^* g$ does not have a jump in trace across $\partial\Omega$: $[[\operatorname{Tr}(G * \operatorname{Tr}^* g)]] = 0$.

Since $(-\Delta + 1)G$ is the Dirac distribution, it holds (see Remark B.2):

$$(-\Delta + 1)(G * \operatorname{Tr}^* g) = \operatorname{Tr}^* g.$$

For $v \in S_{\partial\Omega}(\mathcal{B}'(\partial\Omega))$ with support disjoint from $\partial\Omega$, $\langle \operatorname{Tr}^* g, v \rangle_{[S_{\partial\Omega}(\mathcal{B}')], S_{\partial\Omega}(\mathcal{B}')} = 0$, hence:

$$(-\Delta + 1)(G * \operatorname{Tr}^* g) = 0 \text{ on } \mathbb{R}^n \setminus \partial \Omega.$$

Therefore $\Delta(G * \operatorname{Tr}^* g) \in L^2(\mathbb{R}^n \setminus \partial \Omega).$

By formula (B.8), it also holds:

$$(-\Delta + 1)(G * \operatorname{Tr}^* g) = \operatorname{Tr}^* \left[\frac{\partial (G * \operatorname{Tr}^* g)}{\partial \nu} \right]$$

therefore, since $\operatorname{Tr}^* : \mathcal{B}'(\partial\Omega) \to [\mathcal{S}_{\partial\Omega}(\mathcal{B}'(\partial\Omega))]'$ is into,

$$\left[\!\left[\frac{\partial(G*\operatorname{Tr}^*g)}{\partial\nu}\right]\!\right] = g,$$

and $G * \operatorname{Tr}^* g$ is a solution (therefore, it is the unique solution) to the transmission problem (4.1) with k = 1 and f = 0 associated to $g \in \mathcal{B}'(\partial\Omega)$. Moreover, $G * \operatorname{Tr}^* g \in H^1_{\partial\Omega,[0]_{\operatorname{Tr}}}(\mathbb{R}^n)$ for $g \in \mathcal{B}'(\partial\Omega)$. In particular, if μ is a *d*-upper regular measure for $d \in [n-1, n]$ supported by $\partial \Omega$ (Definition 2.4), then [53] $L^2(\partial \Omega, \mu) \subset \mathcal{B}'(\partial \Omega)$. Therefore, if $g \in L^2(\partial \Omega, \mu)$, Equation (B.4) states that:

$$\mathcal{S}_{\partial\Omega} g(x) = \int_{\partial\Omega} G(x-y)g(y)\,\mu(\mathrm{d}y), \quad x \in \mathbb{R}^n$$

which is identical to the classical formula (see Subsection 4.1) replacing Lebesgue's measure σ with μ .

For the transmission problem (4.1) with k = 1, g = 0 and with a nonhomogeneous jump in trace $-f \in \mathcal{B}(\partial\Omega)$, we consider the space $H^1_{\partial\Omega,[-f]_{\mathrm{Tr}}}(\mathbb{R}^n)$ defined on pages iii–vi.

Proposition B.3. Let Ω be a two-sided extension domain of \mathbb{R}^n . Then the linear operator

$$\mathcal{D}_{\partial\Omega}: \mathcal{B}(\partial\Omega) \longrightarrow \mathcal{D}_{\partial\Omega}(\mathcal{B}(\partial\Omega)) \subset H^1_{\partial\Omega}(\mathbb{R}^n)$$
$$f = -\llbracket \operatorname{Tr} u \rrbracket \longmapsto u = \mathcal{D}_{\partial\Omega}f,$$

uniquely defined by the variational formulation

$$\forall v \in H^1_{\partial\Omega, [-f]}(\mathbb{R}^n), \quad \langle \mathcal{D}_{\partial\Omega}f, v \rangle_{H^1_{\partial\Omega}(\mathbb{R}^n)} = -\left\langle \frac{\partial(\mathcal{D}_{\partial\Omega}f)}{\partial\nu}, f \right\rangle_{\mathcal{B}', \mathcal{B}}, \tag{B.6}$$

where $\frac{\partial}{\partial \nu}(\mathcal{D}_{\partial\Omega}f) := \frac{\partial_i}{\partial \nu}(\mathcal{D}_{\partial\Omega}f) = \frac{\partial_e}{\partial \nu}(\mathcal{D}_{\partial\Omega}f)$, is bounded:

$$\|\mathcal{D}_{\partial\Omega}f\|_{H^1_{\partial\Omega}(\mathbb{R}^n)} \le C \|f\|_{\mathcal{B}(\partial\Omega)},\tag{B.7}$$

for some constant $C = C(\Omega) > 0$. This operator $\mathcal{D}_{\partial\Omega}$ is called the double layer potential operator for the transmission problem (4.1) with k = 1.

By the same type of construction as for the problem on a bounded domain Ω (see [38, Theorem 7]), we obtain that the transmission problem (4.1) with k = 1 and g = 0 is well-posed on $H^1_{\partial\Omega}(\mathbb{R}^n)$ for all $f \in \mathcal{B}(\partial\Omega)$. This proves Proposition B.3.

To give an explicit formula for the double layer potential operator, we introduce operators which allow to restate Green's formula.

Proposition B.4. Let Ω be a one-sided extension domain of \mathbb{R}^n . Then, there exists a unique linear continuous operator $\mathfrak{D}_i : H^1(\Omega) \to \mathcal{B}'(\partial\Omega)$ such that, for $v \in H^1(\Omega)$, it holds:

$$\forall u \in H^1_{\Delta}(\Omega), \quad \langle \mathfrak{D}_i v, \operatorname{Tr}_i u \rangle_{\mathcal{B}', \mathcal{B}} = \left\langle \frac{\partial_i u}{\partial \nu}, \operatorname{Tr}_i v \right\rangle_{\mathcal{B}', \mathcal{B}},$$

where $H^1_{\Delta}(\Omega)$ is defined by Equation (2.11).

If Ω is a two-sided extension domain of \mathbb{R}^n , then there exists a unique linear continuous operator $\mathfrak{D}_e : H^1(\Omega^c) \to \mathcal{B}'(\partial\Omega)$ such that, for $v \in H^1(\Omega^c)$, it holds:

$$\forall u \in H^1_{\Delta}(\Omega^c), \quad \langle \mathfrak{D}_e v, \operatorname{Tr}_e u \rangle_{\mathcal{B}', \mathcal{B}} = \left\langle \frac{\partial_e u}{\partial \nu}, \operatorname{Tr}_e v \right\rangle_{\mathcal{B}', \mathcal{B}}$$

Proof. Let $v \in H^1(\Omega)$. If $\phi \in \mathcal{B}'(\partial \Omega)$ is such that:

$$\forall u \in H^1_{\Delta}(\Omega), \quad \langle \phi, \operatorname{Tr}_i u \rangle_{\mathcal{B}', \mathcal{B}} = \left\langle \frac{\partial_i u}{\partial \nu}, \operatorname{Tr}_i v \right\rangle_{\mathcal{B}', \mathcal{B}}$$

then

$$\operatorname{Tr}_{i}^{*}\phi = \frac{\partial_{i}}{\partial\nu}^{*} \circ \operatorname{Tr}_{i} v,$$

where $\frac{\partial_i}{\partial \nu}^* : \mathcal{B}(\partial \Omega) \to (H^1(\Omega))'$ and $\operatorname{Tr}_i^* : \mathcal{B}'(\partial \Omega) \to (H^1(\Omega))'$. Diagram (3.7) guarantees the existence and uniqueness of $\mathfrak{D}_i v := \phi$. In the same way, we prove the existence of the linear operator $\mathfrak{D}_e : H^1(\Omega^c) \to \mathcal{B}'(\partial \Omega)$ and its uniqueness.

Let $v \in H^1(\Omega)$. For all $f \in \mathcal{B}(\partial\Omega)$, there exists $u \in H^1_{\Delta}(\Omega)$ which depends continuously on f and such that $\operatorname{Tr}_i u = f$. Then it holds:

$$\left| \langle \mathfrak{D}_{i} v, \operatorname{Tr}_{i} u \rangle_{\mathcal{B}', \mathcal{B}} \right| = \left| \left\langle \frac{\partial_{i} u}{\partial \nu}, \operatorname{Tr}_{i} v \right\rangle_{\mathcal{B}', \mathcal{B}} \right| \leq C \| u \|_{H^{1}_{\Delta}(\Omega)} \| v \|_{H^{1}(\Omega)},$$

where C > 0, independent from u and v, is linked to the continuity of the operators $\frac{\partial_i}{\partial \nu}$: $H^1_{\Delta}(\Omega) \to \mathcal{B}'(\partial \Omega)$ and Tr_i : $H^1(\Omega) \to \mathcal{B}(\partial \Omega)$, and $\|u\|_{H^1_{\Delta}(\Omega)}$ is defined in (2.13). Since u depends continuously on f, there exists C' > 0 which depends on C but neither on u or f, nor on v such that:

$$\left| \langle \mathfrak{D}_i v, \operatorname{Tr}_i u \rangle_{\mathcal{B}', \mathcal{B}} \right| \le C' \| f \|_{\mathcal{B}(\partial \Omega)} \| v \|_{H^1(\Omega)},$$

hence the continuity of $\mathfrak{D}_i : H^1(\Omega) \to \mathcal{B}'(\partial\Omega)$. In the same way, we can prove the continuity of $\mathfrak{D}_e : H^1(\Omega^c) \to \mathcal{B}'(\partial\Omega)$.

The operators \mathfrak{D}_i and \mathfrak{D}_e allow us to state a general relation between an element of $H^1_{\partial\Omega}(\mathbb{R}^n)$ and its behaviour on $\partial\Omega$, which is a generalization of a similar formula which can be found in [75] in the case of a Lipschitz domain.

Proposition B.5. Let $u \in H^1_{\partial\Omega}(\mathbb{R}^n)$, such that $-\Delta u + u = F \in L^2(\mathbb{R}^n)$ on $\mathbb{R}^n \setminus \partial\Omega$. It holds:

$$\begin{aligned} \forall v \in H^{1}(\mathbb{R}^{n}), \quad \left\langle (-\Delta + 1)u, v \right\rangle_{L^{2}(\mathbb{R}^{n})} &= \\ \left\langle F, v \right\rangle_{L^{2}(\mathbb{R}^{n})} - \left\langle \mathfrak{D}v, \llbracket \operatorname{Tr} u \rrbracket \right\rangle_{\mathcal{B}', \mathcal{B}} + \left\langle \left[\left[\frac{\partial u}{\partial \nu} \right] \right], \operatorname{Tr} v \right\rangle_{\mathcal{B}', \mathcal{B}} \end{aligned}$$

where $\operatorname{Tr} v := \operatorname{Tr}_i v = \operatorname{Tr}_e v$ as well as $\mathfrak{D}v := \mathfrak{D}_i v = \mathfrak{D}_e v$. Therefore, in $(H^1(\mathbb{R}^n))'$, it holds

$$-\Delta u + u = F - \mathfrak{D}^* \llbracket \operatorname{Tr} u \rrbracket + \operatorname{Tr}^* \llbracket \frac{\partial u}{\partial \nu} \rrbracket, \qquad (B.8)$$

,

where $\mathfrak{D}^* : \mathcal{B}(\partial\Omega) \to (H^1(\mathbb{R}^n))'$ and $\operatorname{Tr}^* : \mathcal{B}'(\partial\Omega) \to (H^1(\mathbb{R}^n))'.$

Proof. From [75], if $v \in H^1(\mathbb{R}^n)$, then $[[\operatorname{Tr} v]] = 0$ and $[[\frac{\partial v}{\partial \nu}]] = 0$, hence the notations $\operatorname{Tr} v$ and $\mathfrak{D} v$. By Green's formula, it holds:

$$\begin{aligned} \forall v \in H^1(\mathbb{R}^n), \quad \int_{\Omega \cup \Omega^c} \nabla u \cdot \nabla v \, \mathrm{d}x + \int_{\mathbb{R}^n} uv \, \mathrm{d}x \\ &= \langle F, v \rangle_{L^2(\mathbb{R}^n)} + \left\langle \left[\left[\frac{\partial u}{\partial \nu} \right] \right], \operatorname{Tr} v \right\rangle_{\mathcal{B}', \mathcal{B}}. \end{aligned}$$

Since Green's formulae can be restated using the operators \mathfrak{D}_i and \mathfrak{D}_e as:

$$\begin{cases} \langle -\Delta u + u, v \rangle_{L^{2}(\Omega)} = \langle u, v \rangle_{H^{1}(\Omega)} - \langle \mathfrak{D}v, \operatorname{Tr}_{i} u \rangle_{\mathcal{B}', \mathcal{B}} \\ \langle -\Delta u + u, v \rangle_{L^{2}(\Omega^{c})} = \langle u, v \rangle_{H^{1}(\Omega^{c})} + \langle \mathfrak{D}v, \operatorname{Tr}_{e} u \rangle_{\mathcal{B}', \mathcal{B}} \end{cases},$$

we yield the result.

We complete the definition of the double layer potential generalizing the classical formula as a convolution with the fundamental solution to $(-\Delta + 1)$.

Theorem B.6. Let Ω be a two-sided extension domain of \mathbb{R}^n . If $f \in \mathcal{B}(\partial \Omega)$, then the unique solution u to the transmission problem (4.1) for k = 1 and g = 0 associated to f can be written as:

$$\forall x \in \mathbb{R}^n \setminus \partial\Omega, \quad \mathcal{D}_{\partial\Omega}f(x) = (G * \mathfrak{D}^*f)(x) \in H^1_{\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n), \qquad (B.9)$$

where G is the fundamental solution to $(-\Delta + 1)$ on \mathbb{R}^n , $\mathfrak{D}^* : \mathcal{B}(\partial\Omega) \to (H^1(\mathbb{R}^n))'$ and the convolution is on \mathbb{R}^n . Therefore,

$$\forall x \in \mathbb{R}^n \backslash \partial \Omega \quad \mathcal{D}_{\partial \Omega} f(x) = \langle \mathfrak{D} G(x - \cdot), f \rangle_{\mathcal{B}', \mathcal{B}}.$$
 (B.10)

Proof. First of all, $(-\Delta + 1)G$ is the Dirac distribution δ_0 , therefore it holds, for $f \in \mathcal{B}(\partial\Omega)$:

$$(-\Delta+1)(G*\mathfrak{D}^*f)=\mathfrak{D}^*f.$$

Let $v \in H^1(\mathbb{R}^n)$ with a support disjoint from $\partial\Omega$. Then it holds, for any $\varphi \in H^1(\mathbb{R}^n)$ such that $\operatorname{Tr} \varphi = f$,

$$\langle \mathfrak{D}^* f, v \rangle_{(H^1)', H^1} = \left\langle \frac{\partial \varphi}{\partial \nu}, \operatorname{Tr} v \right\rangle_{\mathcal{B}', \mathcal{B}} = 0,$$

hence:

$$(-\Delta + 1)(G * \mathfrak{D}^* f) = 0$$
 on $\mathbb{R}^n \setminus \partial \Omega$.

Then, we proceed in the same way as it is done in [75, Theorem 6.11]. Let $f \in \mathcal{B}(\partial \Omega)$. Let us denote:

$$u := \begin{cases} \mathcal{D}f & \text{on } \Omega\\ 0 & \text{on } \Omega^c \end{cases} \in H^1_{\partial\Omega}(\mathbb{R}^n).$$

Then $-\Delta u + u = 0$ weakly on $\mathbb{R}^n \setminus \partial \Omega$, therefore, using formula (B.8), it holds in $(H^1(\mathbb{R}^n))'$:

$$-\Delta u + u = -\mathfrak{D}^* \llbracket \operatorname{Tr} u \rrbracket + \operatorname{Tr}^* \llbracket \frac{\partial u}{\partial \nu} \rrbracket,$$

hence, since $(-\Delta + 1)G = \delta_0$ and $S_{\partial\Omega} = G * \text{Tr}^*$, it holds:

$$u = -G * \mathfrak{D}^* \llbracket \operatorname{Tr} u \rrbracket + \mathcal{S}_{\partial \Omega} \llbracket \frac{\partial u}{\partial \nu} \rrbracket.$$

We can deduce:

$$G * \mathfrak{D}^* f = \mathcal{S}_{\partial\Omega} \circ \frac{\partial_i}{\partial\nu} (\mathcal{D}f) - u.$$

Only, for all $g \in \mathcal{B}'(\partial\Omega)$, it holds $[[\operatorname{Tr}(\mathcal{S}_{\partial\Omega} g)]] = 0$ and $[[\frac{\partial}{\partial\nu}(\mathcal{S}_{\partial\Omega} g)]] = g$, hence:

$$\llbracket \operatorname{Tr}(G * \mathfrak{D}^* f) \rrbracket = -\llbracket \operatorname{Tr} u \rrbracket = -f,$$

and

$$\left[\!\left[\frac{\partial}{\partial\nu}(G * \mathfrak{D}^* f)\right]\!\right] = \frac{\partial_i}{\partial\nu}(\mathcal{D}f) - \left[\!\left[\frac{\partial u}{\partial\nu}\right]\!\right] = 0.$$

Therefore, $G * \mathfrak{D}^* f$ is the unique solution to the transmission problem (4.1) with k = 1 and g = 0 associated to $f \in \mathcal{B}(\partial \Omega)$. Equation (B.10) holds by the definition of the operators involved.

For regular boundaries (at least C^1) the classical formula for the double layer potential operator (see Subsection 4.1) is well-defined on the space $L^2(\partial\Omega, \sigma)$: the unique solution to the transmission problem (4.1) with k = 1, g = 0 and $f = -[[\operatorname{Tr} u]] \in L^2(\partial\Omega, \sigma)$ is $\mathcal{D}_{\partial\Omega}f(x) \in H^2_{\partial\Omega}(\mathbb{R}^n)$ ((2.15) with H^2 instead of H^1), given by:

$$\forall x \in \mathbb{R}^n \setminus \partial\Omega, \quad \mathcal{D}_{\partial\Omega}f(x) = (G * \mathfrak{D}^*f)(x) = \langle \mathfrak{D}G(x - \cdot), f \rangle_{L^2(\partial\Omega,\sigma)}$$
$$= \int_{\partial\Omega} \frac{\partial G}{\partial\nu_y}(x - y)f(y)\,\sigma(dy).$$
(B.11)

Indeed, in that case, the weak solution to the transmission problem is in the space $H^2_{\partial\Omega}(\mathbb{R}^n)$, and the derivation operators $\frac{\partial_i}{\partial\nu}: H^2(\Omega) \to L^2(\partial\Omega, \sigma)$ and $\frac{\partial_e}{\partial\nu}: H^2(\Omega^c) \to L^2(\partial\Omega, \sigma)$ are linear and bounded. Therefore, the C^1 regularity of the boundary (and the fact that $\frac{\partial G}{\partial\nu} \in L^2(\partial\Omega, \sigma)$) ensures enough regularity for the weak solution so the classical integral formula coincides with the operator $\mathcal{D}_{\partial\Omega}$ in our framework. If the boundary is less regular, such as a Lipschitz or non-Lipschitz boundary, the weak solution u does not belong to H^2 anymore, only $\Delta u \in L^2(\mathbb{R}^n)$, which is not sufficient to use the integral formula. By [48], if the boundary is (exactly) Lipschitz, the domain must be convex to ensure the H^2 regularity of the weak solution to the elliptic problem with the Dirichlet boundary condition, yet the complement of a convex Lipschitz domain is not a convex domain. That regularity problem does not occur

for the C^1 -boundaries. For any two-sided extension domain Ω , Formula (B.10) generalizes the classical formula from Equation (B.11) without assuming extra regularity for the solution.

Using the definitions of both the single $S_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to H^1_{\partial\Omega}(\mathbb{R}^n)$ and the double layer potential operator $\mathcal{D}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to H^1_{\partial\Omega}(\mathbb{R}^n)$, we give the following result which is known as Green's third identity [92], generalized here in the case of a bounded two-sided extension domain.

Theorem B.7. Let Ω be a bounded two-sided extension domain of \mathbb{R}^n . The unique solution on $H^1_{\partial\Omega}(\mathbb{R}^n)$ to the transmission problem (4.1) with k = 1, for any $f \in \mathcal{B}(\partial\Omega)$ and $g \in \mathcal{B}'(\partial\Omega)$, is given by $u = \mathcal{S}_{\partial\Omega} g - \mathcal{D}_{\partial\Omega} f$.

Proof. $S_{\partial\Omega} g$ is the unique solution to the transmission problem (4.1) with k = 1 and f = 0 associated to $g \in \mathcal{B}'(\partial\Omega)$, and $\mathcal{D}_{\partial\Omega}f$ is the unique solution to the transmission problem (4.1) with k = 1 and g = 0 associated to $f \in \mathcal{B}(\partial\Omega)$. Since (4.1) is linear, the superimposition principle applies. Since the transmission problem with k = 1 is well-posed, and $u := S_{\partial\Omega} g - \mathcal{D}_{\partial\Omega} f$ is a solution to that problem, it is the unique solution.

For the operator $-\Delta$. Analogously, it is possible to define the single and double layer potential operators for the weak solutions to the transmission problem (4.1) related to $-\Delta$ and have an analogous representation of harmonic solutions as 1-harmonic ones, given in Proposition B.7. However, we need to be careful about the functional spaces on which the well-posedness of the transmission problems for $-\Delta$ occurs. The bilinear form associated to $-\Delta$:

$$a(u,v) := \int_{\Omega} \nabla u \cdot \nabla v \, \mathrm{d}x + \int_{\Omega^c} \nabla u \cdot \nabla v \, \mathrm{d}x$$

does not necessarily define an inner product in a Hilbert space resembling $H^1_{\partial\Omega}(\mathbb{R}^n)$ and reflecting the properties of the solution.

The main difference comes from whether n = 2 or $n \ge 3$. For $n \ge 3$, it is possible to impose additional conditions on u, such as a mean value zero on the bulk and at infinity, so that $a(\cdot, \cdot)$ be an inner product. Indeed, with those extra conditions, both $\|\nabla \cdot\|_{L^2(\Omega)^n}$ and $\|\nabla \cdot\|_{L^2(\Omega^c)^n}$ become norms, but only when $n \ge 3$ for the latter. In particular, the condition $n \ge 3$ ensures the existence of the limit:

$$\langle u \rangle_{\infty} := \lim_{R \to +\infty} \frac{1}{\lambda^{(n)}(B_R(0))} \int_{B_R(0)} u \, \mathrm{d}x. \tag{B.12}$$

Therefore, we consider the cases of \mathbb{R}^2 , and of \mathbb{R}^n with $n \ge 3$ separately. It is known that for $n \ge 3$ [71], the space:

$$W(\mathbb{R}^n) := \overline{C_0^{\infty}(\mathbb{R}^n)}^{\|\nabla \cdot\|_{L^2(\mathbb{R}^n)^n}} = \left\{ u \in L^2_{loc}(\mathbb{R}^n) \mid \nabla u \in L^2(\mathbb{R}^n)^n, \ \langle u \rangle_{\infty} = 0 \right\}$$
(B.13)

endowed with the inner product $\langle u, v \rangle_{W(\mathbb{R}^n)} = \int_{\mathbb{R}^n} \nabla u \cdot \nabla v \, dx$ is a Hilbert space. It is important to notice that $H^1(\mathbb{R}^n)$ is dense and continuously embedded in $W(\mathbb{R}^n)$, but $W(\mathbb{R}^n)$ is not contained in $L^2(\mathbb{R}^n)$.

If $n \geq 3$ and Ω is a two-sided extension domain of \mathbb{R}^n , then the space:

$$W_{\partial\Omega}(\mathbb{R}^n) := \left\{ u \in L^2_{loc}(\mathbb{R}^n) \mid \nabla u \mid_{\Omega} \in L^2(\Omega)^n, \ \nabla u \mid_{\Omega^c} \in L^2(\Omega^c)^n, \ \langle u \rangle_{\infty} = 0 \right\},$$
(B.14)

is a Hilbert space endowed with the norm:

$$\|u\|_{W_{\partial\Omega}}^2 := \|u\|_{H^1(\Omega)}^2 + \|\nabla u\|_{L^2(\Omega^c)^n}^2.$$
(B.15)

However, $\|\nabla \cdot\|_{L^2(\Omega)^n}$ is only a semi-norm on $H^1(\Omega)$ and could become a norm if we add a mean value zero property, for instance on Ω . Thus we consider:

$$W_{\partial\Omega,\langle 0\rangle_{\Omega}}(\mathbb{R}^{n}) := \left\{ u \in W_{\partial\Omega}(\mathbb{R}^{n}) \ \bigg| \ \int_{\Omega} u \, \mathrm{d}x = 0 \right\}$$
(B.16)

which is a Hilbert space endowed with the norm

$$\|u\|_{W_{\partial\Omega,\langle 0\rangle_{\Omega}}(\mathbb{R}^{n})}^{2} := \|\nabla u\|_{L^{2}(\Omega)^{n}}^{2} + \|\nabla u\|_{L^{2}(\Omega^{c})^{n}}^{2}.$$
 (B.17)

Similarly to the bounded case, it holds $\operatorname{Tr}_i(W_{\partial\Omega,\langle 0\rangle_\Omega}(\mathbb{R}^n)) = \mathcal{B}_0(\partial\Omega)$. Given the Dirichlet boundary condition of the transmission problem (4.1) (with k = 0and g = 0), that means if $u \in W_{\partial\Omega,\langle 0\rangle_\Omega}(\mathbb{R}^n)$ is a solution to the problem, then $\operatorname{Tr}_i u$, $\operatorname{Tr}_e u \in \mathcal{B}_0(\partial\Omega)$. Therefore, the Neumann boundary condition is to be understood in $\mathcal{B}'_0(\partial\Omega)$, which requires to define the weak exterior normal derivative $\frac{\partial_e^{\Omega} u}{\partial \nu} \in \mathcal{B}'_0(\partial\Omega)$, adapting Definition 2.13:

Definition B.8. (Weak exterior normal derivative in $\mathcal{B}'_0(\partial\Omega)$) Let $u \in H^1_{\langle 0 \rangle, \Delta}(\Omega^c)$ (defined on pages *iii-vi*). The weak exterior normal derivative in $\mathcal{B}'_0(\partial\Omega)$ of u is the element $\phi \in \mathcal{B}'_0(\partial\Omega)$ such that for all $v \in H^1_{\langle 0 \rangle}(\Omega^c)$, it holds:

$$\langle \phi, \operatorname{Tr}_{e} v \rangle_{\mathcal{B}'_{0}, \mathcal{B}_{0}} = -\int_{\Omega^{c}} (\Delta u) v \, \mathrm{d}x - \int_{\Omega^{c}} \nabla u \cdot \nabla v \, \mathrm{d}x,$$

and we denote $\frac{\partial_e^{\mathbf{o}} u}{\partial \nu} = \phi.$

Remark B.9. As it is mentioned in [5] initially for $W(\mathbb{R}^n)$, the space $W_{\partial\Omega}(\mathbb{R}^n)$ is not a subspace of $L^2(\mathbb{R}^n)$, but contains $H^1(\mathbb{R}^n)$ for $n \geq 3$ [3]. The main difference between $W_{\partial\Omega}(\mathbb{R}^n)$ and $W(\mathbb{R}^n)$ is that the former allows jumps in gradient across $\partial\Omega$, and its elements do not need to be in H^1 on the subdomains containing $\partial\Omega$: they only need to be in L^2_{loc} . In addition [5], in \mathbb{R}^3 the only difference between $H^1(\mathbb{R}^3)$ and $W(\mathbb{R}^3)$ is the speed of the decay for $|x| \to +\infty$: in $W(\mathbb{R}^3)$ the elements decay at least as $|x|^{-1/2}$, whereas elements of $H^1(\mathbb{R}^3)$ must decay at least as fast as $|x|^{-3/2}$ as $|x| \to +\infty$. If n = 2 [4, 3], $\|\nabla \cdot\|_{L^2(\mathbb{R}^n)^n}$ is not a norm on $W(\mathbb{R}^2)$, since the Poincarétype inequalities [71] do not hold: $\langle u \rangle_{\infty}$ can be infinite, should the limit even exist. Therefore, it only makes sense to consider a class of functions vanishing at infinity, as it is done [92] for the external elliptic problems:

$$V_{van}(\mathbb{R}^2) = \{ u \in H^1_{\partial\Omega}(\mathbb{R}^2) \mid u \to 0 \text{ as } |x| \to +\infty \}.$$
 (B.18)

That means we need to work in $W_{\partial\Omega}(\mathbb{R}^2) \cap L^2(\mathbb{R}^2)$ at least.

We introduce the following notation:

$$V_{\partial\Omega}(\mathbb{R}^n) := \begin{cases} V_{van}(\mathbb{R}^2) & \text{if } n = 2, \\ W_{\partial\Omega,\langle 0 \rangle_{\Omega}}(\mathbb{R}^n) & \text{if } n \ge 3. \end{cases}$$
(B.19)

The definitions of the layer potentials being directly linked to the wellposedness of the transmission problem, we begin with stating the following well-posedness result:

Theorem B.10. Let Ω be a bounded two-sided extension domain of \mathbb{R}^n . Let \tilde{G} be the fundamental solution to $-\Delta$ on \mathbb{R}^n . Then, for all $g \in \mathcal{B}'_0(\partial\Omega)$, the transmission problem (4.1) with k = 0 and f = 0 has a unique solution, given by the formula:

$$u = \tilde{G} * \operatorname{Tr}^* g \tag{B.20}$$

which goes to zero as $|x| \to +\infty$. Here, $\operatorname{Tr}^* : \mathcal{B}'_0(\partial\Omega) \to V'_{\partial\Omega,[0]_{\partial_\nu}}(\mathbb{R}^n)$, where $V_{\partial\Omega,[0]_{\partial_\nu}}(\mathbb{R}^n)$ is defined on pages iii-vi. If n=2, then $u \in V_{van}(\mathbb{R}^2)$. If $n \geq 3$, then $u \in W_{\partial\Omega,\langle 0 \rangle_{\Omega}}(\mathbb{R}^n)$ and it is the unique solution to the variational formulation

$$\forall v \in W_{\partial\Omega,\langle 0\rangle_{\Omega}}(\mathbb{R}^n), \quad \int_{\Omega} \nabla u \cdot \nabla v \, \mathrm{d}x + \int_{\Omega^c} \nabla u \cdot \nabla v \, \mathrm{d}x = \langle g, \operatorname{Tr} v \rangle_{\mathcal{B}'_0, \mathcal{B}_0}.$$
(B.21)

In addition, for all $n \geq 2$, the following a priori estimate holds:

$$\|u\|_{W_{\partial\Omega,\langle 0\rangle_{\Omega}}(\mathbb{R}^{n})} \le C \|g\|_{\mathcal{B}'_{0}(\partial\Omega)},\tag{B.22}$$

for some constant $C := C(n, \Omega) > 0$.

Proof. Let us start with the case n = 2. Firstly we verify that $\tilde{G} * \operatorname{Tr}^* g$ solves the transmission problem for all $g \in \mathcal{B}'_0(\partial\Omega)$. Since $-\Delta G$ is the Dirac distribution, it holds:

$$-\Delta(\tilde{G}*\operatorname{Tr}^*g) = \operatorname{Tr}^*g$$

For all $v \in V_{\partial\Omega,[0]_{\partial\nu}}(\mathbb{R}^n)$ which has a support disjoint from $\partial\Omega$, it holds $\langle \operatorname{Tr}^* g, v \rangle_{V'_{\partial\Omega,[0]_{\partial\nu}},V_{\partial\Omega,[0]_{\partial\nu}}} = 0$, hence:

$$-\Delta(\hat{G} * \operatorname{Tr}^* g) = 0 \quad \text{on } \mathbb{R}^n \backslash \partial\Omega.$$

By formula (B.8), it also holds:

$$-\Delta(\tilde{G}*\operatorname{Tr}^*g) = \operatorname{Tr}^*\left[\!\!\left[\frac{\partial(\tilde{G}*\operatorname{Tr}^*g)}{\partial\nu}\!\!\right]\!\!\right],$$

therefore, since $\operatorname{Tr}^* : \mathcal{B}'_0(\partial\Omega) \to V'_{\partial\Omega,[0]_{\partial\nu}}(\mathbb{R}^n)$ is into,

$$\left[\!\!\left[\frac{\partial(\tilde{G}*\operatorname{Tr}^*g)}{\partial\nu}\right]\!\!\right] = g,$$

and $G*\mathrm{Tr}^* g$ is a solution to the transmission problem associated to $g \in \mathcal{B}'_0(\partial\Omega)$.

Then, using the Kelvin transform and the uniqueness of the solution on Ω as soon as $\int_{\Omega} u \, dx = 0$, it also follows the uniqueness of the exterior problem. In addition, by [92, p. 379], its limit for $|x| \to +\infty$ is a constant. Therefore, the unique constant compatible by the Kelvin transformation with $\int_{\Omega} u \, dx = 0$, is zero (see also [92, Theorem 1, V.28.2, p. 418] ensuring that a solution to the Poisson equation is unique in the class of distributions $(C_0^{\infty}(\mathbb{R}^n))'$ with the assumption of taking the value zero at infinity).

The formula given by that theorem yields the *a priori* estimate.

For $n \geq 3$ the space $W_{\partial\Omega,\langle 0\rangle_{\Omega}}(\mathbb{R}^n)$ is a Hilbert space corresponding to the inner product associated to the norm (B.17). Then, we apply the Riesz representation theorem to the variational formulation (B.21) and obtain the well-posedness result. From the uniqueness, formula (B.20) allows to calculate the solution.

From the well-posedness of the transmission problem, we may define the single layer potential operator for the transmission problem (4.1) for k = 0 as a linear continuous operator.

Proposition B.11. Let Ω be a bounded two-sided extension domain of \mathbb{R}^n . Let \tilde{G} be the fundamental solution to $-\Delta$ on \mathbb{R}^n . Let the single layer potential operator be defined as the operator mapping the jump in normal derivative $\begin{bmatrix} \frac{\partial u}{\partial \nu} \end{bmatrix} = g \in \mathcal{B}'_0(\partial \Omega)$ to the unique solution $u \in V_{\partial \Omega}(\mathbb{R}^n)$ to the transmission problem (4.1) with k = 0 and f = 0, which can be expressed by the convolution formula (B.20):

$$\tilde{\mathcal{S}}_{\partial\Omega}: g \in \mathcal{B}'_0(\partial\Omega) \longmapsto \tilde{\mathcal{S}}_{\partial\Omega}g := \tilde{G} * \operatorname{Tr}^* g \in V_{\partial\Omega}(\mathbb{R}^n).$$
(B.23)

Then this operator is linear and continuous.

For the transmission problem (4.1) with k = 0 and g = 0, we proceed as in the bounded case. Note that since the solution u is harmonic on both Ω and Ω^c , then both $\frac{\partial_i u}{\partial \nu}$ and $\frac{\partial_e u}{\partial \nu}$ are elements of $\mathcal{B}'_{0,ext}(\partial\Omega)$ – or equivalently (Theorem A.5) elements of $\mathcal{B}'_0(\partial\Omega)$. However, though $\operatorname{Tr}_i u \in \mathcal{B}_0(\partial\Omega)$ by definition, that does not imply that $\operatorname{Tr}_e u$ must be in $\mathcal{B}_0(\partial\Omega)$ as well (a counter-example can be u = cte). Therefore, it makes sense to consider a Dirichlet boundary condition in $\mathcal{B}(\partial\Omega)$, even when it is not an element of $\mathcal{B}_0(\partial\Omega)$. For a fixed $f \in \mathcal{B}(\partial\Omega)$, we choose $\varphi \in H^1_{\partial\Omega}(\mathbb{R}^n)$ with $\Delta \varphi|_{\Omega} \in L^2(\Omega)$ and $\Delta \varphi|_{\Omega^c} \in L^2(\Omega^c)$ such that $[[\operatorname{Tr} \varphi]] = -f$ and $[[\frac{\partial \varphi}{\partial \nu}]] = 0$. Then $u \in H^1_{\partial\Omega}(\mathbb{R}^n)$ is a solution to the transmission problem if and only if $w := u - \varphi$ is a solution to:

$$\begin{cases} -\Delta w = \Delta \varphi & \text{on } \mathbb{R}^n \backslash \partial \Omega, \\ \llbracket \operatorname{Tr} w \rrbracket = 0, & \llbracket \frac{\partial w}{\partial \nu} \rrbracket = 0, \end{cases}$$

the variational form of which is:

$$\forall v \in V_{\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n), \quad \int_{\Omega \cup \Omega^c} \nabla w \cdot \nabla v \, \mathrm{d}x = \int_{\Omega \cup \Omega^c} (\Delta \varphi) v \, \mathrm{d}x,$$

where $V_{\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n)$ is defined on pages iii–vi. Using Green's formula, it can be written as:

$$\forall v \in V_{\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n), \quad \int_{\Omega \cup \Omega^c} \nabla w \cdot \nabla v \, \mathrm{d}x = \int_{\Omega} (\Delta \varphi) v \, \mathrm{d}x \\ - \int_{\Omega^c} \nabla \varphi \cdot \nabla v \, \mathrm{d}x - \left\langle \frac{\partial \varphi}{\partial \nu}, \operatorname{Tr} v \right\rangle_{\mathcal{B},\mathcal{B}}. \quad (B.24)$$

If $n \geq 3$, the left-hand side of the variational formulation (B.24) is none other than an inner product on $V_{\partial\Omega}(\mathbb{R}^n)$, therefore Riesz representation theorem applies. For n = 2, we proceed as in the case of the jump in normal derivative, constructing the solution defined by a convolution with the fundamental solution to $-\Delta$ on \mathbb{R}^n . We summarize the results in the following theorem.

Proposition B.12. Let Ω be a bounded two-sided extension domain of \mathbb{R}^n . If $f \in \mathcal{B}(\partial \Omega)$, then the unique solution $u \in V_{\partial \Omega}(\mathbb{R}^n)$ to the transmission problem (4.1) for k = 0 and g = 0 associated to f can be written as:

$$\forall x \in \mathbb{R}^n \setminus \partial\Omega, \quad u(x) = \hat{\mathcal{D}}_{\partial\Omega} f(x) := (\hat{G} * \mathfrak{D}^* f)(x), \tag{B.25}$$

where \tilde{G} is the fundamental solution to $-\Delta$ on \mathbb{R}^n , $\mathfrak{D}^* : \mathcal{B}(\partial\Omega) \to (H^1(\mathbb{R}^n))'$ and the convolution is on \mathbb{R}^n . Then the mapping $\tilde{\mathcal{D}}_{\partial\Omega} : -[[\operatorname{Tr} u]] = f \in$ $\mathcal{B}(\partial\Omega) \mapsto \tilde{\mathcal{D}}_{\partial\Omega}f := u \in V_{\partial\Omega}(\mathbb{R}^n)$ is linear bounded operator, called the double layer potential operator, and it holds:

$$\tilde{\mathcal{D}}_{\partial\Omega}f = \langle \mathfrak{D}\tilde{G}(x-\cdot), f \rangle_{\mathcal{B}',\mathcal{B}}.$$
(B.26)

Proof. The boundedness of the double layer potential operator follows from the *a priori* estimate and Riesz representation theorem for $n \ge 3$ and from the direct estimate of the convolution formula. All other statements are straightforward.

As for equation (B.11), for regular boundaries (at least C^1), the solution $\tilde{\mathcal{D}}_{\partial\Omega}f$ to the transmission problem (4.1) for k = 0 and g = 0 is in $V_{\partial\Omega}(\mathbb{R}^n) \cap H^2_{\partial\Omega}(\mathbb{R}^n)$ and the classical formula is well-defined in $L^2(\partial\Omega, \sigma)$:

$$\forall x \in \mathbb{R}^n \setminus \partial\Omega, \quad \tilde{\mathcal{D}}_{\partial\Omega}f(x) = (\tilde{G} * \mathfrak{D}^*f)(x) = (\mathfrak{D}\tilde{G}(x-\cdot), f)_{L^2(\partial\Omega,\sigma)}$$
$$= \int_{\partial\Omega} \frac{\partial\tilde{G}(x-y)}{\partial\nu_y} f(y) \,\sigma(dy).$$
(B.27)

We establish the general relation between an element of $W_{\partial\Omega,\langle 0\rangle_{\Omega}}(\mathbb{R}^n)$ and its behaviour on $\partial\Omega$ by the analogous way as in Proposition B.5.

Proposition B.13. (Representation formula for $-\Delta$) Let $u \in W_{\partial\Omega,\langle 0 \rangle_{\Omega}}(\mathbb{R}^n)$, such that $-\Delta u = f$ on $\mathbb{R}^n \setminus \partial\Omega$ with $f \in L^2(\mathbb{R}^n)$. It holds, for all $v \in H^1(\mathbb{R}^n)$:

$$\langle -\Delta u, v \rangle_{L^2(\mathbb{R}^n)} = \langle f, v \rangle_{L^2(\mathbb{R}^n)} - \langle \mathfrak{D}v, \llbracket \operatorname{Tr} u \rrbracket \rangle_{\mathcal{B}', \mathcal{B}} + \left\langle \llbracket \frac{\partial u}{\partial \nu} \rrbracket, \operatorname{Tr} v \right\rangle_{\mathcal{B}', \mathcal{B}}$$

where $\operatorname{Tr} v := \operatorname{Tr}_i v = \operatorname{Tr}_e v$ as well as $\mathfrak{D}v := \mathfrak{D}_i v = \mathfrak{D}_e v$.

Finally, as previously for $(-\Delta+1)$ in Theorem B.7, we also establish Green's third identity for $-\Delta$:

Theorem B.14. Let Ω be a two-sided extension domain \mathbb{R}^n . The unique solution on $V_{\partial\Omega}(\mathbb{R}^n)$ to transmission problem (4.1) with k = 0 for all $f \in \mathcal{B}(\partial\Omega)$ and $g \in \mathcal{B}'_0(\partial\Omega)$, is given by $u = \tilde{\mathcal{S}}_{\partial\Omega}g - \tilde{\mathcal{D}}_{\partial\Omega}f$.

B.2 Approximation by truncated problems

For the operator $(-\Delta+1)$. First, let us assume k = 1. Since problem (4.10) is understood in the weak sense, the solutions are sought in an H^1 -type space. Allowing jumps across $\partial\Omega$ and considering the homogeneous Dirichlet boundary condition on ∂U , we may define the space of solutions:

$$H^{1}_{0,\partial\Omega}(U) := \left\{ u \in L^{2}(U) \mid \nabla u|_{U \setminus \partial\Omega} \in L^{2}(U \setminus \partial\Omega)^{n}, \ \operatorname{Tr}^{\partial U} u = 0 \right\}$$
(B.28)

endowed with the norm product:

$$\|\cdot\|_{H^{1}_{0,\partial\Omega}(U)} := \sqrt{\|\cdot\|^{2}_{H^{1}(\Omega)} + \|\cdot\|^{2}_{H^{1}(U\setminus\overline{\Omega})}}.$$
 (B.29)

Remark B.15. The homogeneous Dirichlet boundary condition on U allows to extend elements of $H^1_{0,\partial\Omega}(U)$ to $H^1_{\partial\Omega}(\mathbb{R}^n)$ by setting 0 outside of U. If $u \in$ $H^1_{0,\partial\Omega}(U)$, we will keep denoting that extension by u and it holds $||u||_{H^1_{0,\partial\Omega}(U)} =$ $||u||_{H^1_{\partial\Omega}(\mathbb{R}^n)}$. Let us start with defining the single layer potential operator for the truncated problem (4.10) with k = 1 and f = 0. Proceeding in the same way as for (B.2), the variational formulation associated to the truncated transmission problem with no jump in trace is given by:

$$\forall v \in H^1_{0,\partial\Omega,[0]_{\mathrm{Tr}}}(U), \quad \langle u, v \rangle_{H^1_{0,\partial\Omega}(U)} = \langle g, \mathrm{Tr}^{\partial\Omega} v \rangle_{\mathcal{B}',\mathcal{B}}, \tag{B.30}$$

where $H^1_{0,\partial\Omega,[0]_{\mathrm{Tr}}}(U)$ is defined on pages iii–vi. By the Riesz representation theorem, the problem is well-posed.

Proposition B.16. Let Ω be a two-sided extension domain of \mathbb{R}^n , and let U be a one-sided extension domain such that $\Omega \subset \subset U$. Let

$$\begin{split} \mathcal{S}^{U}_{\partial\Omega} &: \mathcal{B}'(\partial\Omega) \longrightarrow \mathcal{S}^{U}_{\partial\Omega}(\mathcal{B}'(\partial\Omega)) \subset H^{1}_{0,\partial\Omega}(U) \\ g &= \left[\left[\frac{\partial u}{\partial \nu} \right] \right]_{\partial\Omega} \longmapsto u = \mathcal{S}^{U}_{\partial\Omega} \, g, \end{split}$$

be the linear operator uniquely defined by

$$\forall v \in H^1_{0,\partial\Omega,[0]_{\mathrm{Tr}}}(U), \quad \langle \mathcal{S}^U_{\partial\Omega} g, v \rangle_{H^1_{0,\partial\Omega}(U)} = \langle g, \mathrm{Tr} \, v \rangle_{\mathcal{B}',\mathcal{B}}. \tag{B.31}$$

Then $\mathcal{S}^{U}_{\partial\Omega}: \mathcal{B}'(\partial\Omega) \to H^{1}_{0,\partial\Omega}(U)$, called the single layer potential operator for the transmission problem truncated at U, is a linear bounded operator:

$$\|\mathcal{S}_{\partial\Omega}^U g\|_{H^1_{0,\partial\Omega}(U)} \le C \|g\|_{\mathcal{B}'(\partial\Omega)}$$
(B.32)

for some constant $C := C(n, \Omega) > 0$.

In the same way, consider problem (4.10) with k = 1 and g = 0 to define the double layer potential operator truncated at U. The variational formulation for the truncated transmission problem with no jump in normal derivative is given by (as mentioned before, all normal derivatives are on $\partial\Omega$):

$$\forall v \in H^1_{0,\partial\Omega,[-f]_{\mathrm{Tr}}}(U), \quad \langle u, v \rangle_{H^1_{0,\partial\Omega}(U)} = -\left\langle \frac{\partial u}{\partial\nu}, f \right\rangle_{\mathcal{B}',\mathcal{B}(\partial\Omega)}.$$
 (B.33)

Again, using the same method as in [38, Theorem 7] and the Riesz representation theorem, this problem is well-posed.

Proposition B.17. Let Ω be a two-sided extension domain of \mathbb{R}^n , and let U be a bounded one-sided extension domain such that $\Omega \subset \subset U$. Let

$$\mathcal{D}^{U}_{\partial\Omega}: \mathcal{B}(\partial\Omega) \longrightarrow \mathcal{D}^{U}_{\partial\Omega}(\mathcal{B}(\partial\Omega)) \subset H^{1}_{0,\partial\Omega}(U)$$
$$f = - \llbracket \operatorname{Tr} u \rrbracket_{\partial\Omega} \longmapsto u = \mathcal{D}^{U}_{\partial\Omega} f,$$

be the linear operator uniquely defined by

$$\forall v \in H^{1}_{0,\partial\Omega,[-f]_{\mathrm{Tr}}}(U), \quad \langle \mathcal{D}^{U}_{\partial\Omega}f, v \rangle_{H^{1}_{0,\partial\Omega}(U)} = -\left\langle \frac{\partial(\mathcal{D}^{U}_{\partial\Omega}f)}{\partial\nu}, f \right\rangle_{\mathcal{B}',\mathcal{B}(\partial\Omega)}, \quad (B.34)$$

where $\frac{\partial}{\partial \nu}(\mathcal{D}^{U}_{\partial\Omega}f) := \frac{\partial_{i}}{\partial \nu}(\mathcal{D}^{U}_{\partial\Omega}f) = \frac{\partial_{e}}{\partial \nu}(\mathcal{D}^{U}_{\partial\Omega}f).$ Then $\mathcal{D}^{U}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to H^{1}_{0,\partial\Omega}(U)$, called the double layer potential operator for the transmission problem truncated at U, is a linear bounded operator:

$$\|\mathcal{D}_{\partial\Omega}^{U}f\|_{H^{1}_{0,\partial\Omega}(U)} \le C\|f\|_{\mathcal{B}(\partial\Omega)}$$
(B.35)

for some constant $C := C(n, \Omega) > 0$.

We may now prove the truncated transmission problem (4.10) is an approximation of the transmission problem on the whole space (4.1) for k = 1, in the sense of the strong convergence of the associated solutions. As it is done in the Lipschitz case [3], let us consider an increasing sequence of bounded one-sided extension domains $(U_k)_{k \in \mathbb{N}}$ such that there exists an increasing sequence $(B_k)_{k \in \mathbb{N}}$ of balls such that:

$$\forall k \in \mathbb{N}, \quad \Omega \subset \subset B_k \subset U_k \quad \text{and} \quad \bigcup_{k \in \mathbb{N}} B_k = \mathbb{R}^n.$$
 (B.36)

We will denote $\mathcal{S}_{\partial\Omega}^k := \mathcal{S}_{\partial\Omega}^{U_k}$ and $\mathcal{D}_{\partial\Omega}^k := \mathcal{D}_{\partial\Omega}^{U_k}$ for $k \in \mathbb{N}$ the single and double layer potentials for the problem truncated at U_k respectively. For $f \in \mathcal{B}(\partial\Omega)$ and $g \in \mathcal{B}'(\partial\Omega)$, $\mathcal{D}_{\partial\Omega}^k f$ and $\mathcal{S}_{\partial\Omega}^k g$ will be seen as elements of $H^1_{\partial\Omega}(\mathbb{R}^n)$ by setting 0 outside of U_k .

Theorem B.18. Let Ω be a two-sided extension domain of \mathbb{R}^n , and $(U_k)_{k \in \mathbb{N}}$ be an increasing sequence of bounded one-sided extension domains such that there exists an increasing sequence $(B_k)_{k \in \mathbb{N}}$ of balls satisfying (B.36). Then,

(i) for any $g \in \mathcal{B}'(\partial \Omega)$, it holds:

$$\mathcal{S}^k_{\partial\Omega} g \xrightarrow[k \to \infty]{} \mathcal{S}_{\partial\Omega} g \quad in \ H^1_{\partial\Omega}(\mathbb{R}^n).$$

(ii) for any $f \in \mathcal{B}(\partial \Omega)$, it holds:

$$\mathcal{D}^k_{\partial\Omega}f \xrightarrow[k \to \infty]{} \mathcal{D}_{\partial\Omega}f \qquad in \ H^1_{\partial\Omega}(\mathbb{R}^n).$$

Proof. For point (i), let $g \in \mathcal{B}'(\partial\Omega)$, and let us denote $u_k := \mathcal{S}_{\partial\Omega}^k g$ and $u := \mathcal{S}_{\partial\Omega} g$. We follow the steps of the proof of [3, Theorem 4.3]. For all $k \in \mathbb{N}$, it holds:

$$\begin{aligned} \|u_k\|_{H^1_{\partial\Omega}(\mathbb{R}^n)}^2 &= \|u_k\|_{H^1_{\partial\Omega}(U_k)}^2 = \langle g, \operatorname{Tr}^{\partial\Omega} u_k \rangle_{\mathcal{B}',\mathcal{B}} \\ &\leq \|g\|_{\mathcal{B}'} \|u_k\|_{H^1_{\partial\Omega}(U_k)} = \|g\|_{\mathcal{B}'} \|u_k\|_{H^1_{\partial\Omega}(\mathbb{R}^n)} \end{aligned}$$

Therefore, $(u_k)_{k\in\mathbb{N}}$ is bounded in $H^1_{\partial\Omega}(\mathbb{R}^n)$ and has a weak accumulation point, denoted by $u_{\infty} \in H^1_{\partial\Omega}(\mathbb{R}^n)$. Since $(U_k)_{k\in\mathbb{N}}$ is an increasing sequence, considering the extensions of elements of $H^1_{\partial\Omega}(U_k)$ by setting 0 outside of U_k for all $k \in \mathbb{N}$, we can say $(H^1_{\partial\Omega}(U_k))_{k\in\mathbb{N}}$ forms an increasing sequence of Hilbert subspaces of $H^1_{\partial\Omega}(\mathbb{R}^n)$. Therefore, passing to the limit in the variational formulation (B.39) (the left hand side can be seen as an inner product on $H^1_{\partial\Omega}(\mathbb{R}^n)$), it holds:

$$\forall v \in \bigcup_{k \in \mathbb{N}} H^1_{\partial\Omega, [0]_{\mathrm{Tr}}}(U_k), \quad \langle u_{\infty}, v \rangle_{H^1_{\partial\Omega}(\mathbb{R}^n)} = \langle g, \mathrm{Tr}^{\partial\Omega} v \rangle_{\mathcal{B}', \mathcal{B}}.$$

By density of $\bigcup_{k\in\mathbb{N}} H^1_{\partial\Omega,[0]_{\mathrm{Tr}}}(U_k)$ in $H^1_{\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n)$, the previous inequality holds for all $v \in H^1_{\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n)$, hence $u = u_{\infty}$. Since u is the only weak accumulation point of the bounded sequence $(u_k)_{k\in\mathbb{N}}$ in $H^1_{\partial\Omega}(\mathbb{R}^n)$, it holds $u_k \rightharpoonup u$ in $H^1_{\partial\Omega}(\mathbb{R}^n)$ as $k \rightarrow \infty$. By the variational formulations (B.2) for the problem on the whole space and (B.31) for the truncated problems, it holds:

$$\forall k \in \mathbb{N}, \quad \|u_k\|_{H^1_{\partial\Omega}(\mathbb{R}^n)} = \|u_k\|_{H^1_{\partial\Omega}(U_k)} = \|u\|_{H^1_{\partial\Omega}(\mathbb{R}^n)},$$

hence the strong convergence.

Point (ii) is proved in a similar way, using the boundedness of the interior normal derivation operator and straightforward norm inequalities to yield the boundedness of the sequence of solutions to the truncated problem. \Box

Consequently, Green's third identity (Proposition B.7) for the truncated problem shows the solutions to the truncated problem strongly converge in $H^1_{\partial\Omega}(\mathbb{R}^n)$ to the solution of the transmission problem on the whole space as the truncation domain U grows to \mathbb{R}^n in the sense of (B.36).

For the operator $-\Delta$. Let us now focus on the case k = 0. As for the transmission problem on the whole space, uniqueness conditions need to be added to problem (4.10) for it to be well-posed. Since we will be considering the variational formulation of the transmission problems (truncated and on the whole space), we assume $n \geq 3$. Let us introduce the space on which the problem is to be solved:

$$H^{1}_{0,\partial\Omega,\langle 0\rangle_{\Omega}}(U) := \left\{ u \in L^{2}(U) \ \middle| \ \nabla u |_{U \setminus \partial\Omega} \in L^{2}(U \setminus \partial\Omega)^{n}, \\ \int_{\Omega} u \, \mathrm{d}x = 0 \text{ and } \operatorname{Tr}^{\partial U} u = 0 \right\}, \quad (B.37)$$

endowed with the inner product:

$$\|\cdot\|_{H^{1}_{0,\partial\Omega,\langle 0\rangle_{\Omega}}(U)}^{2} := \|\nabla\cdot\|_{L^{2}(\Omega)^{n}}^{2} + \|\nabla\cdot\|_{L^{2}(U\setminus\overline{\Omega})^{n}}^{2}$$
(B.38)

Remark B.19. Similarly to the case k = 1, the homogeneous Dirichlet boundary condition on U allows to extend elements of $H^1_{0,\partial\Omega,\langle 0\rangle_\Omega}(U)$ to $W_{\partial\Omega,\langle 0\rangle_\Omega}(\mathbb{R}^n)$ by setting 0 outside of U. If $u \in H^1_{0,\partial\Omega,\langle 0\rangle_\Omega}(U)$, we will keep denoting that extension by u and it holds $||u||_{H^1_{0,\partial\Omega,\langle 0\rangle_\Omega}(U)} = ||u||_{W_{\partial\Omega,\langle 0\rangle_\Omega}(\mathbb{R}^n)}$.

Let us start with defining the single layer potential operator for the truncated problem (4.10) with k = 0 and f = 0. The variational formulation associated to the truncated transmission problem with no jump in trace is given by:

$$\forall v \in H^1_{0,\partial\Omega,\langle 0\rangle_{\Omega},[0]_{\mathrm{Tr}}}(U), \quad \int_{\Omega} \nabla u \cdot \nabla v \,\mathrm{d}x + \int_{U \setminus \overline{\Omega}} \nabla u \cdot \nabla v \,\mathrm{d}x = \langle g, \mathrm{Tr}^{\partial\Omega} v \rangle_{\mathcal{B}'_0,\mathcal{B}_0},$$
(B.39)

where $H^1_{0,\partial\Omega,\langle 0\rangle_{\Omega},[0]_{\mathrm{Tr}}}(U)$ is defined on pages iii–vi. By the Riesz representation theorem, the problem is well-posed.

Proposition B.20. Let Ω be a two-sided extension domain of \mathbb{R}^n , $n \geq 3$, and let U be a bounded one-sided extension domain such that $\Omega \subset \subset U$. Let

$$\begin{split} \tilde{\mathcal{S}}^{U}_{\partial\Omega} &: \mathcal{B}'_{0}(\partial\Omega) \longrightarrow \tilde{\mathcal{S}}^{U}_{\partial\Omega}(\mathcal{B}'_{0}(\partial\Omega)) \subset H^{1}_{0,\partial\Omega,\langle 0 \rangle_{\Omega}}(U) \\ g &= \left[\left[\frac{\partial u}{\partial \nu} \right] \right] \longmapsto u = \tilde{\mathcal{S}}^{U}_{\partial\Omega} \, g, \end{split}$$

be the linear operator uniquely defined by

$$\forall v \in H^1_{0,\partial\Omega,\langle 0\rangle_{\Omega},[0]_{\mathrm{Tr}}}(U), \quad \langle \tilde{\mathcal{S}}^U_{\partial\Omega} g, v \rangle_{H^1_{0,\partial\Omega,\langle 0\rangle_{\Omega}}(U)} = \langle g, \mathrm{Tr} \, v \rangle_{\mathcal{B}'_0,\mathcal{B}_0}. \tag{B.40}$$

Then $\tilde{\mathcal{S}}^U_{\partial\Omega}: \mathcal{B}'_0(\partial\Omega) \to H^1_{0,\partial\Omega,\langle 0 \rangle_\Omega}(U)$, called the single layer potential operator for the transmission problem truncated at U, is a linear bounded operator:

$$\|\tilde{\mathcal{S}}^{U}_{\partial\Omega} g\|_{H^{1}_{0,\partial\Omega,\langle 0\rangle_{\Omega}}(U)} \le C \|g\|_{\mathcal{B}'_{0}(\partial\Omega)}$$
(B.41)

۰

for some constant $C := C(\Omega) > 0$.

In the same way, consider problem (4.10) with k = 0 and q = 0. The variational formulation for the truncated transmission problem with no jump in normal derivative is given by:

$$\forall v \in H^1_{0,\partial\Omega,\langle 0\rangle_{\Omega},[-f]_{\mathrm{Tr}}}(U), \quad \langle u,v \rangle_{H^1_{0,\partial\Omega,\langle 0\rangle_{\Omega}}(U)} = -\left\langle \frac{\partial u}{\partial\nu}, f \right\rangle_{\mathcal{B}'_0,\mathcal{B}_0}.$$
(B.42)

Using the same method as in [38, Theorem 7], this problem is well-posed.

Proposition B.21. Let Ω be a two-sided extension domain of \mathbb{R}^n , $n \geq 3$, and let U be a bounded one-sided extension domain such that $\Omega \subset \subset U$. Let

$$\begin{split} \tilde{\mathcal{D}}^{U}_{\partial\Omega} : \mathcal{B}(\partial\Omega) &\longrightarrow \tilde{\mathcal{D}}^{U}_{\partial\Omega}(\mathcal{B}(\partial\Omega)) \subset H^{1}_{0,\partial\Omega,\langle 0 \rangle_{\Omega}}(U) \\ f &= - \left[\!\!\left[\operatorname{Tr} u\right]\!\!\right]_{\partial\Omega} \longmapsto u = \tilde{\mathcal{D}}^{U}_{\partial\Omega} f, \end{split}$$

be the linear operator uniquely defined by

$$\forall v \in H^{1}_{0,\partial\Omega,\langle 0\rangle_{\Omega},[-f]_{\mathrm{Tr}}}(U), \quad \langle \tilde{\mathcal{D}}^{U}_{\partial\Omega}f,v\rangle_{H^{1}_{0,\partial\Omega,\langle 0\rangle_{\Omega}}(U)} = -\left\langle \frac{\partial(\tilde{\mathcal{D}}^{U}_{\partial\Omega}f)}{\partial\nu},f\right\rangle_{\mathcal{B}'_{0},\mathcal{B}_{0}},$$
(B.43)

where $\frac{\partial}{\partial \nu}(\tilde{\mathcal{D}}^{U}_{\partial\Omega}f) := \frac{\partial_{i}}{\partial \nu}(\tilde{\mathcal{D}}^{U}_{\partial\Omega}f) = \frac{\partial_{e}}{\partial \nu}(\tilde{\mathcal{D}}^{U}_{\partial\Omega}f).$ Then $\tilde{\mathcal{D}}^{U}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to H^{1}_{0,\partial\Omega,\langle 0 \rangle_{\Omega}}(U)$, called the double layer potential operator for the transmission problem truncated at U, is a linear bounded operator:

$$\|\tilde{\mathcal{D}}^{U}_{\partial\Omega}f\|_{H^{1}_{0,\partial\Omega,\langle 0\rangle_{\Omega}}(U)} \le C\|f\|_{\mathcal{B}(\partial\Omega)}$$
(B.44)

for some constant $C := C(\Omega) > 0$.

We may now prove the truncated transmission problem (4.10) is an approximation of (4.1) for k = 0, in the sense of the strong convergence of the associated solutions. Let us consider an increasing sequence of bounded one-sided extension domains $(U_k)_{k\in\mathbb{N}}$ in the sense of (B.36). We will denote $\tilde{\mathcal{S}}_{\partial\Omega}^k := \tilde{\mathcal{S}}_{\partial\Omega}^{U_k}$ and $\tilde{\mathcal{D}}_{\partial\Omega}^k := \tilde{\mathcal{D}}_{\partial\Omega}^{U_k}$ for $k \in \mathbb{N}$ the single and double layer potentials for the problem truncated at U_k respectively. For $f \in \mathcal{B}(\partial\Omega)$ and $g \in \mathcal{B}'_0(\partial\Omega)$, $\tilde{\mathcal{D}}_{\partial\Omega}^k f$ and $\tilde{\mathcal{S}}_{\partial\Omega}^k g$ will be seen as elements of $W_{\partial\Omega,\langle 0 \rangle_\Omega}(\mathbb{R}^n)$ by setting 0 outside of U_k .

Theorem B.22. Let Ω be a two-sided extension domain of \mathbb{R}^n , $n \geq 3$, and $(U_k)_{k \in \mathbb{N}}$ be an increasing sequence of bounded one-sided extension domains such that there exists an increasing sequence of balls satisfying (B.36). Then,

(i) for any $g \in \mathcal{B}'_0(\partial \Omega)$, it holds:

$$\tilde{\mathcal{S}}^k_{\partial\Omega} g \longrightarrow \tilde{\mathcal{S}}_{\partial\Omega} g \qquad in \ W_{\partial\Omega,\langle 0 \rangle_\Omega}(\mathbb{R}^n) \quad as \ k \to +\infty.$$

(ii) for any $f \in \mathcal{B}(\partial \Omega)$, it holds:

$$\tilde{\mathcal{D}}^k_{\partial\Omega} f \longrightarrow \tilde{\mathcal{D}}_{\partial\Omega} f \qquad in \ W_{\partial\Omega,\langle 0 \rangle_\Omega}(\mathbb{R}^n) \quad as \ k \to +\infty.$$

Proof. We proceed in the same way as for Theorem B.18. The density of $\bigcup_{k\in\mathbb{N}} H^1_{0,\partial\Omega,\langle 0\rangle_\Omega}(U_k)$ in $W_{\partial\Omega,\langle 0\rangle_\Omega}(\mathbb{R}^n)$ comes from [3, Theorem 4.3].

B.3 Neumann-Poincaré operators and boundary values

Unlike the usual definition of the Neumann-Poincaré operator as a singular integral on the boundary [77, 91], we choose to define it as the mean value of the interior and the exterior traces of the solution to the Dirichlet transmission problem.

Definition B.23. (Neumann-Poincaré operator) If Ω is a bounded twosided extension domain of \mathbb{R}^n , let us define:

(i) $\mathcal{K}_{\partial\Omega}$, the Neumann-Poincaré operator for the transmission problem (4.1), associated to $(-\Delta+1)$ for g = 0, as the average of the traces of $u = \mathcal{D}_{\partial\Omega}f$:

$$\mathcal{K}_{\partial\Omega}: \mathcal{B}(\partial\Omega) \longrightarrow \mathcal{B}(\partial\Omega)$$
$$f = -\llbracket \operatorname{Tr} u \rrbracket \longmapsto \frac{1}{2} (\operatorname{Tr}_i + \operatorname{Tr}_e) u = \frac{1}{2} (\operatorname{Tr}_i + \operatorname{Tr}_e) \circ \mathcal{D}_{\partial\Omega} f.$$

(ii) $\tilde{\mathcal{K}}_{\partial\Omega}$, the Neumann-Poincaré operator for the transmission problem (4.1) associated to $-\Delta$ for g = 0 as the average of the traces of $u = \tilde{\mathcal{D}}_{\partial\Omega}f$:

$$\mathcal{K}_{\partial\Omega}: \mathcal{B}(\partial\Omega) \longrightarrow \mathcal{B}(\partial\Omega)$$
$$f = -\llbracket \operatorname{Tr} u \rrbracket \longmapsto \frac{1}{2} (\operatorname{Tr}_i + \operatorname{Tr}_e) u = \frac{1}{2} (\operatorname{Tr}_i + \operatorname{Tr}_e) \circ \tilde{\mathcal{D}}_{\partial\Omega} f.$$

In other words, since for all $f \in \mathcal{B}(\partial\Omega)$, $[[\operatorname{Tr}(\mathcal{D}_{\partial\Omega}f)]] = -f = [[\operatorname{Tr}(\tilde{\mathcal{D}}_{\partial\Omega}f)]]$, it holds:

$$\begin{cases} \operatorname{Tr}_{i}(\mathcal{D}_{\partial\Omega}f) = \left(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega}\right)f, \\ \operatorname{Tr}_{e}(\mathcal{D}_{\partial\Omega}f) = \left(\frac{1}{2}I + \mathcal{K}_{\partial\Omega}\right)f, \\ \end{cases} \quad \text{and} \quad \begin{cases} \operatorname{Tr}_{i}(\tilde{\mathcal{D}}_{\partial\Omega}f) = \left(-\frac{1}{2}I + \tilde{\mathcal{K}}_{\partial\Omega}\right)f, \\ \operatorname{Tr}_{e}(\tilde{\mathcal{D}}_{\partial\Omega}f) = \left(\frac{1}{2}I + \tilde{\mathcal{K}}_{\partial\Omega}\right)f. \end{cases} \end{cases}$$

$$(B.45)$$

Given that definition, the Neumann-Poincaré operators are connected to the jumps in trace of the double layer potentials. The following proposition states their adjoints play the same part for the jumps in normal derivatives of the single layer potentials. Note that the sign convention $[[Tr(\mathcal{D}_{\partial\Omega}f)]] = -f$ we chose earlier induces those adjoint operators are equal to the average of the normal derivatives of the single layer potentials, instead of the opposite of that quantity.

Proposition B.24. Let Ω be a bounded two-sided extension domain of \mathbb{R}^n . Then:

(i) for all $g \in \mathcal{B}'(\partial \Omega)$, it holds:

$$\begin{cases} \frac{\partial_i (\mathcal{S}_{\partial \Omega} g)}{\partial \nu} = \left(\frac{1}{2}I + \mathcal{K}^*_{\partial \Omega}\right)g\\ \frac{\partial_e (\mathcal{S}_{\partial \Omega} g)}{\partial \nu} = \left(-\frac{1}{2}I + \mathcal{K}^*_{\partial \Omega}\right)g \end{cases}, \tag{B.46}$$

where $\mathcal{K}^*_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ denotes the adjoint operator to $\mathcal{K}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}(\partial\Omega)$. Hence, it holds:

$$\mathcal{K}^*_{\partial\Omega}: \mathcal{B}'(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$$
$$g \mapsto \frac{1}{2} \left(\frac{\partial_i}{\partial\nu} + \frac{\partial_e}{\partial\nu} \right) \circ \mathcal{S}_{\partial\Omega} g.$$

Moreover, the boundary operators $\mathcal{K}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}(\partial\Omega)$ and $\mathcal{K}^*_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ are linear and continuous;

(ii) For all $g \in \mathcal{B}'_{0,ext}(\partial\Omega)$, it holds:

$$\begin{cases} \frac{\partial_i(\tilde{\mathcal{S}}_{\partial\Omega} g)}{\partial\nu} = \left(\frac{1}{2}I + \tilde{\mathcal{K}}^*_{\partial\Omega}\right)g\\ \frac{\partial_e(\tilde{\mathcal{S}}_{\partial\Omega} g)}{\partial\nu} = \left(-\frac{1}{2}I + \tilde{\mathcal{K}}^*_{\partial\Omega}\right)g \end{cases}, \quad (B.47)$$

where $\tilde{\mathcal{K}}^*_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ denotes the adjoint operator to $\tilde{\mathcal{K}}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}(\partial\Omega)$. Thus, it holds:

$$\tilde{\mathcal{K}}^*_{\partial\Omega} : \mathcal{B}'_0(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega)$$
$$g \mapsto \frac{1}{2} \left(\frac{\partial_i}{\partial\nu} + \frac{\partial_e}{\partial\nu} \right) \circ \tilde{\mathcal{S}}_{\partial\Omega} g$$

Moreover, the boundary operators $\tilde{\mathcal{K}}_{\partial\Omega}$: $\mathcal{B}(\partial\Omega) \to \mathcal{B}(\partial\Omega)$ and $\tilde{\mathcal{K}}^*_{\partial\Omega}$: $\mathcal{B}'_0(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega)$ are linear and continuous.

Proof. We proceed as it is done in [66]. Let $f \in \mathcal{B}(\partial\Omega)$ and $g \in \mathcal{B}'(\partial\Omega)$. Let us denote $\hat{\mathcal{K}}_{\partial\Omega} := \frac{1}{2} \left(\frac{\partial_i}{\partial \nu} + \frac{\partial_e}{\partial \nu} \right) \circ \mathcal{S}_{\partial\Omega}$. Then, given the definitions of the layer potential operators $\mathcal{S}_{\partial\Omega}$ and $\mathcal{D}_{\partial\Omega}$, it holds, by Green's formula:

$$0 = \left\langle \frac{\partial_i(\mathcal{D}_{\partial\Omega}f)}{\partial\nu}, \operatorname{Tr}_i(\mathcal{S}_{\partial\Omega}g) \right\rangle_{\mathcal{B}',\mathcal{B}} - \left\langle \frac{\partial_e(\mathcal{D}_{\partial\Omega}f)}{\partial\nu}, \operatorname{Tr}_e(\mathcal{S}_{\partial\Omega}g) \right\rangle_{\mathcal{B}',\mathcal{B}}$$
$$= \int_{\mathbb{R}^n} \nabla(\mathcal{D}_{\partial\Omega}f) \cdot \nabla(\mathcal{S}_{\partial\Omega}g) \, \mathrm{d}x + \int_{\mathbb{R}^n} (\mathcal{D}_{\partial\Omega}f)(\mathcal{S}_{\partial\Omega}g) \, \mathrm{d}x$$
$$= \left\langle \frac{\partial_i(\mathcal{S}_{\partial\Omega}g)}{\partial\nu}, \operatorname{Tr}_i(\mathcal{D}_{\partial\Omega}f) \right\rangle_{\mathcal{B}',\mathcal{B}} - \left\langle \frac{\partial_e(\mathcal{S}_{\partial\Omega}g)}{\partial\nu}, \operatorname{Tr}_e(\mathcal{D}_{\partial\Omega}f) \right\rangle_{\mathcal{B}',\mathcal{B}}$$

Hence:

$$\begin{cases} \left\langle \frac{\partial_i(\mathcal{D}f)}{\partial\nu}, \operatorname{Tr}_i(\mathcal{S}_{\partial\Omega}g) \right\rangle_{\mathcal{B}',\mathcal{B}} = \left\langle \frac{\partial_e(\mathcal{D}_{\partial\Omega}f)}{\partial\nu}, \operatorname{Tr}_e(\mathcal{S}_{\partial\Omega}g) \right\rangle_{\mathcal{B}',\mathcal{B}} \\ \left\langle \frac{\partial_i(\mathcal{S}_{\partial\Omega}g)}{\partial\nu}, \operatorname{Tr}_i(\mathcal{D}f) \right\rangle_{\mathcal{B}',\mathcal{B}} = \left\langle \frac{\partial_e(\mathcal{S}_{\partial\Omega}g)}{\partial\nu}, \operatorname{Tr}_e(\mathcal{D}_{\partial\Omega}f) \right\rangle_{\mathcal{B}',\mathcal{B}} \end{cases}$$

therefore we deduce:

$$\left\langle \begin{bmatrix} \frac{\partial(\mathcal{S}_{\partial\Omega} g)}{\partial\nu} \end{bmatrix}, \operatorname{Tr}_{i}(\mathcal{D}_{\partial\Omega} f) + \operatorname{Tr}_{e}(\mathcal{D}_{\partial\Omega} f) \right\rangle_{\mathcal{B}',\mathcal{B}} = \left\langle \frac{\partial_{i}(\mathcal{S}_{\partial\Omega} g)}{\partial\nu} + \frac{\partial_{e}(\mathcal{S}_{\partial\Omega} g)}{\partial\nu}, - \llbracket \operatorname{Tr}(\mathcal{D}_{\partial\Omega} f) \rrbracket \right\rangle_{\mathcal{B}',\mathcal{B}},$$

which means:

$$\langle g, \mathcal{K}_{\partial\Omega} f \rangle_{\mathcal{B}', \mathcal{B}} = \left\langle \hat{\mathcal{K}}_{\partial\Omega} g, f \right\rangle_{\mathcal{B}', \mathcal{B}}.$$

Therefore, $\hat{\mathcal{K}}_{\partial\Omega} = \mathcal{K}^*_{\partial\Omega}$, and the fact that $\left[\!\!\left[\frac{\partial(\mathcal{S}_{\partial\Omega} g)}{\partial\nu}\right]\!\!\right] = g$ yields the formula.

The double layer potential $\mathcal{D}_{\partial\Omega} : \mathcal{B}(\bar{\partial\Omega}) \to H^{\vec{1}}_{\partial\Omega}(\mathbb{R}^n)$ is continuous, and the trace operators $\operatorname{Tr}_i, \operatorname{Tr}_e : H^1_{\partial\Omega}(\mathbb{R}^n) \to \mathcal{B}(\partial\Omega)$ are continuous. Therefore, the operator $\mathcal{K}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}(\partial\Omega)$ and its adjoint operator $\mathcal{K}^*_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to \mathcal{B}'(\partial\Omega)$ are continuous, given Definition B.23.
We proceed in the same way to obtain the relations on the operator $\tilde{\mathcal{K}}^*_{\partial\Omega}$. If $g \in \mathcal{B}'_{0,ext}(\partial\Omega)$, then $\tilde{\mathcal{S}}_{\partial\Omega} g$ is harmonic on Ω and Ω^c , and $\tilde{\mathcal{K}}^*_{\partial\Omega} g \in \mathcal{B}'_{0,ext}(\partial\Omega)$. Therefore, $\tilde{\mathcal{K}}^*_{\partial\Omega}$ can be seen as an endomorphism of $\mathcal{B}'_{0,ext}(\partial\Omega)$. By Theorem A.5, the spaces $\mathcal{B}'_0(\partial\Omega)$ and $\mathcal{B}'_{0,ext}(\partial\Omega)$ are isomorphic, which means $\tilde{\mathcal{K}}^*_{\partial\Omega}$ can be seen as a continuous endomorphism of $\mathcal{B}'_0(\partial\Omega)$ instead.

The generalization of the Neumann-Poincaré operator from Definition B.23 allows to have the same connection to the boundary values of the layer potential operator as in the Lipschitz case, when that operator is defined as a singular integral.

Remark B.25. We can give a formal definition of the operator $\tilde{\mathcal{K}}^*_{\partial\Omega}$ from $\mathcal{B}'(\partial\Omega)$ to $\mathcal{B}'(\partial\Omega)$. However, if we wish to establish the link between $\tilde{\mathcal{K}}^*_{\partial\Omega}$ and the solution to the transmission problem (4.1) for g = 0 and k = 0, as we did in the case of $(-\Delta + 1)$, then we need to restrict the domain of $\tilde{\mathcal{K}}^*_{\partial\Omega}$ to $\mathcal{B}'_{0,ext}(\partial\Omega)$, or equivalently to $\mathcal{B}'_0(\partial\Omega)$.

From that definition of $\tilde{\mathcal{K}}^*_{\partial\Omega}$ as a continuous endomorphism on $\mathcal{B}'_0(\partial\Omega)$, we can deduce its adjoint $\tilde{\mathcal{K}}_{\partial\Omega}$ defines a continuous endomorphism on $\mathcal{B}_0(\partial\Omega)$.

We adapt [2, Lemma 2.5, p. 17] (see also [44]) from the L^2 -framework to our case:

Lemma B.26. Let Ω be a bounded two-sided extension domain of \mathbb{R}^n . Then, for $\lambda \in \mathbb{R}$ such that $|\lambda| \geq \frac{1}{2}$ it holds

- (i) the operator $\lambda I + \mathcal{K}^*_{\partial\Omega}$ is an injection from $\mathcal{B}'(\partial\Omega)$ to $\mathcal{B}'(\partial\Omega)$;
- (ii) the operator $\lambda I + \tilde{\mathcal{K}}^*_{\partial\Omega}$ is an injection from $\mathcal{B}'_0(\partial\Omega)$ to $\mathcal{B}'_0(\partial\Omega)$;

Proof. Let $\lambda \in \mathbb{R}$. Let us assume there exists $\varphi \in \operatorname{Ker}(\lambda I + \mathcal{K}^*_{\partial\Omega}) \setminus \{0\}$. Then, it holds $\mathcal{S}_{\partial\Omega}\varphi \in L^2(\mathbb{R}^n \setminus \partial\Omega)$ and $\nabla \mathcal{S}_{\partial\Omega}\varphi \in L^2(\mathbb{R}^n \setminus \partial\Omega)^n$. Since $\varphi \neq 0$, it holds:

$$\int_{\Omega} |\nabla \mathcal{S}_{\partial \Omega} \varphi|^2 \, \mathrm{d}x > 0 \quad \text{or} \quad \int_{\Omega^c} |\nabla \mathcal{S}_{\partial \Omega} \varphi|^2 \, \mathrm{d}x > 0,$$

that is, using Green's formula:

$$\underbrace{\left\langle \left(\frac{1}{2}I + \mathcal{K}_{\partial\Omega}^{*}\right)\varphi, \operatorname{Tr}\left(\mathcal{S}_{\partial\Omega}\varphi\right)\right\rangle_{\mathcal{B}',\mathcal{B}}}_{A} > 0$$

or
$$\underbrace{-\left\langle \left(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega}^{*}\right)\varphi, \operatorname{Tr}\left(\mathcal{S}_{\partial\Omega}\varphi\right)\right\rangle_{\mathcal{B}',\mathcal{B}}}_{B} > 0.$$

Since $\mathcal{K}^*_{\partial\Omega}\varphi = -\lambda\varphi$, it holds:

$$A = \left(\frac{1}{2} - \lambda\right) \langle \varphi, \operatorname{Tr}\left(\mathcal{S}_{\partial\Omega}\varphi\right) \rangle_{\mathcal{B}',\mathcal{B}} \quad \text{and} \quad B = \left(\frac{1}{2} + \lambda\right) \langle \varphi, \operatorname{Tr}\left(\mathcal{S}_{\partial\Omega}\varphi\right) \rangle_{\mathcal{B}',\mathcal{B}}.$$

Note that, since $\varphi \neq 0$ and by definition of $S_{\partial\Omega}\varphi$, it holds $\langle \varphi, \operatorname{Tr}(S_{\partial\Omega}\varphi) \rangle_{\mathcal{B}',\mathcal{B}} \neq 0$.

If $\lambda = -\frac{1}{2}$, then B = 0, hence $S_{\partial\Omega}\varphi$ is constant on Ω^c . Since $S_{\partial\Omega}\varphi$ is the solution to (B.1) associated to φ , then $S_{\partial\Omega}\varphi = 0$ and $\varphi = 0$, which is contradictory.

If $\lambda \neq -\frac{1}{2}$, then $A \ge 0$ and B > 0, therefore it holds:

$$\lambda = \frac{1}{2} \frac{A - B}{A + B} \quad \in \left[-\frac{1}{2}, \frac{1}{2} \right[.$$

In the same way, we prove the result for $\tilde{\mathcal{K}}^*_{\partial\Omega} : \mathcal{B}'_0(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega)$.

Proposition B.24 highlights the link between the weak solution to our transmission problem, and the Neumann-Poincaré operators and their adjoints, which motivates our definition of those operators. Since (by definition) the single layer potential has no jump in trace and the double layer potential has no jump in normal derivative across the boundary, we may introduce the following operators:

$$\begin{aligned}
\mathcal{V}_{\partial\Omega} &:= \operatorname{Tr} \circ \mathcal{S}_{\partial\Omega} : \mathcal{B}'(\partial\Omega) \to \mathcal{B}(\partial\Omega), \\
\mathcal{W}_{\partial\Omega} &:= -\frac{\partial}{\partial\nu} \circ \mathcal{D}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}'(\partial\Omega),
\end{aligned} \tag{B.48}$$

for the operator $(-\Delta + 1)$, as well as their equivalent for the operator $-\Delta$:

$$\widetilde{\mathcal{V}}_{\partial\Omega} := \operatorname{Tr} \circ \mathcal{S}_{\partial\Omega} : \mathcal{B}'_0(\partial\Omega) \to \mathcal{B}(\partial\Omega), \\
\widetilde{\mathcal{W}}_{\partial\Omega} := -\frac{\partial}{\partial\nu} \circ \mathcal{D}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}'_0(\partial\Omega),$$
(B.49)

Those operators allow to prove symmetrization formulae for the Neumann-Poincaré operators, generalizing [80, Theorem 3.1.3].

Theorem B.27. Let Ω be a two-sided extension domain of \mathbb{R}^n . Consider the following operators:

$$C_i := \frac{1}{2}I + \mathcal{M}$$
 and $C_e := \frac{1}{2}I - \mathcal{M},$

where

$$\mathcal{M} := \begin{pmatrix} -\mathcal{K}_{\partial\Omega} & \mathcal{V}_{\partial\Omega} \\ \mathcal{W}_{\partial\Omega} & \mathcal{K}_{\partial\Omega}^* \end{pmatrix} : \mathcal{B}(\partial\Omega) \times \mathcal{B}'(\partial\Omega) \to \mathcal{B}(\partial\Omega) \times \mathcal{B}'(\partial\Omega).$$

Then, the operators C_i and C_e defined above are linear and continuous projectors, referred to as 'Calderón projectors' (interior and exterior respectively).

Proof. Let $u = S_{\partial\Omega}g - \mathcal{D}_{\partial\Omega}f$ be a solution to the transmission problem on \mathbb{R}^n , with $f \in \mathcal{B}(\partial\Omega)$ and $g \in \mathcal{B}'(\partial\Omega)$. The interior boundary values of u are given by:

$$\mathcal{C}_{i} \begin{pmatrix} \llbracket \operatorname{Tr} u \rrbracket \\ \llbracket \frac{\partial u}{\partial \nu} \rrbracket \end{pmatrix} = \begin{pmatrix} \operatorname{Tr}_{i} u \\ \frac{\partial_{i} u}{\partial \nu} \end{pmatrix} = \begin{pmatrix} \mathcal{V}_{\partial\Omega}g - \left(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega}\right)f \\ \left(\frac{1}{2}I + \mathcal{K}_{\partial\Omega}^{*}\right)g + \mathcal{W}_{\partial\Omega}f \end{pmatrix}$$

We aim at proving that $C_i^2 = C_i$. Let v be a solution to the transmission problem such that $[[\operatorname{Tr} v]] = \operatorname{Tr}_i u$ and $[[\frac{\partial v}{\partial \nu}]] = \frac{\partial_i u}{\partial \nu}$. Since the transmission problem is well-posed and $u \mathbb{1}_{\Omega}$ is a solution with the same boundary jump values as v, it holds $v = u \mathbb{1}_{\Omega}$. Therefore, it holds:

$$\mathcal{C}_{i}^{2} \begin{pmatrix} \llbracket \operatorname{Tr} u \rrbracket \\ \llbracket \frac{\partial u}{\partial \nu} \rrbracket \end{pmatrix} = \begin{pmatrix} \operatorname{Tr}_{i} v \\ \frac{\partial_{i} v}{\partial \nu} \end{pmatrix} = \begin{pmatrix} \operatorname{Tr}_{i} u \\ \frac{\partial_{i} u}{\partial \nu} \end{pmatrix} = \mathcal{C}_{i} \begin{pmatrix} \llbracket \operatorname{Tr} u \rrbracket \\ \llbracket \frac{\partial u}{\partial \nu} \rrbracket \end{pmatrix}.$$

Hence, C_i is a projector. In the same way, we prove C_e is a projector.

Since $C_i^2 = C_i$ and $C_i = \frac{1}{2}I + \mathcal{M}$, we can deduce:

$$\mathcal{M}^2 = \frac{1}{4}I,$$

which can be rewritten as:

$$\begin{cases} \mathcal{K}_{\partial\Omega}\mathcal{V}_{\partial\Omega} = \mathcal{V}_{\partial\Omega}\mathcal{K}^*_{\partial\Omega}, \\ \mathcal{W}_{\partial\Omega}\mathcal{K}_{\partial\Omega} = \mathcal{K}^*_{\partial\Omega}\mathcal{W}_{\partial\Omega}, \end{cases} \quad \text{and} \quad \begin{cases} \mathcal{K}^2_{\partial\Omega} + \mathcal{V}_{\partial\Omega}\mathcal{W}_{\partial\Omega} = \frac{1}{4}I, \\ (\mathcal{K}^*_{\partial\Omega})^2 + \mathcal{W}_{\partial\Omega}\mathcal{V}_{\partial\Omega} = \frac{1}{4}I. \end{cases} \quad (B.50)$$

In the same way, we prove the symmetrization relations for the operators associated to $-\Delta$, stated in Subsection 4.3.

B.4 Poincaré-Steklov operators for transmission problems

As explained in Subsection 4.4, we wish to generalize notion of Poincaré-Steklov operator for the transmission problem (Equation (4.1)) from [89] in the case of extension domains, using the operator we have defined in Appendix B.3. The following theorem proves both definitions given in Subsection 4.4 are equivalent and explains the connection between the solutions u and v involved.

Theorem B.28. Let Ω be a two-sided extension domain of \mathbb{R}^n .

(i) It holds:

$$\mathcal{V}_{\partial\Omega}^{-1}\left(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega}\right) = -\left(\frac{1}{2}I + \mathcal{K}_{\partial\Omega}^*\right)^{-1} \mathcal{W}_{\partial\Omega}.$$

That operator, denoted by $\mathcal{A}_{\partial\Omega,i}$ is called interior Poincaré-Steklov operator for the transmission problem associated to $(-\Delta + 1)$. It is linear continuous invertible and is described by:

$$\mathcal{A}_{\partial\Omega,i}: \mathcal{B}(\partial\Omega) \longrightarrow \mathcal{B}'(\partial\Omega)$$
$$-\llbracket \operatorname{Tr} u \rrbracket \longmapsto \llbracket \frac{\partial v}{\partial\nu} \rrbracket,$$

where $u = -\mathcal{D}_{\partial\Omega}[[\operatorname{Tr} u]]$, and $v = \mathcal{S}_{\partial\Omega}[[\frac{\partial v}{\partial \nu}]]$ is characterized by $u|_{\Omega} = v|_{\Omega}$.

(ii) It holds:

$$\mathcal{V}_{\partial\Omega}^{-1}\left(\frac{1}{2}I + \mathcal{K}_{\partial\Omega}\right) = -\left(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega}^*\right)^{-1} \mathcal{W}_{\partial\Omega}$$

That operator, denoted by $\mathcal{A}_{\partial\Omega,e}$ is called exterior Poincaré-Steklov operator for the transmission problem associated to $(-\Delta + 1)$. It is linear continuous invertible and is described by:

$$\mathcal{A}_{\partial\Omega,e}: \mathcal{B}(\partial\Omega) \longrightarrow \mathcal{B}'(\partial\Omega) \\ -\llbracket \operatorname{Tr} u \rrbracket \longmapsto \left[\left[\frac{\partial w}{\partial\nu} \right] \right],$$

where $u = -\mathcal{D}_{\partial\Omega}[[\operatorname{Tr} u]]$, and $w = \mathcal{S}_{\partial\Omega}[[\frac{\partial w}{\partial \nu}]]$ is characterized by $u|_{\Omega^c} = w|_{\Omega^c}$.

Proof. Let us prove point (i) (the proof of point (ii) is similar). Let $u = -\mathcal{D}_{\partial\Omega}[[\operatorname{Tr} u]]$ and $v = \mathcal{S}_{\partial\Omega}[[\frac{\partial v}{\partial \nu}]]$ be as in (4.15). Then it holds $\operatorname{Tr}_i u = \operatorname{Tr}_i v$, which implies $u|_{\Omega} = v|_{\Omega}$ since the problem

$$\begin{cases} (-\Delta + 1)\varphi = 0 & \text{on } \Omega, \\ \operatorname{Tr} \varphi = f \in \mathcal{B}(\partial \Omega), \end{cases}$$

is well-posed. The fact that $u|_{\Omega} = v|_{\Omega}$ and $\llbracket \operatorname{Tr} v \rrbracket = 0$ uniquely determines the solution v on \mathbb{R}^n (since the Dirichlet problem for $(-\Delta+1)$ on Ω^c is well-posed). Therefore, the operator $\mathcal{A}_{\partial\Omega,i}$ is well-defined and $\mathcal{A}_{\partial\Omega,i} = \mathcal{V}_{\partial\Omega}^{-1}(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega})$. The linearity, continuity and invertibility follow from that relation. Using the well-posedness of the Neumann problems for $(-\Delta+1)$ on Ω and on Ω^c yields $\mathcal{A}_{\partial\Omega,i} = -(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega})^{-1}\mathcal{W}_{\partial\Omega}$.

The definition of the Poincaré-Steklov operators for the transmission problem can be understood in terms of the following equivalences, where $f \in \mathcal{B}(\partial\Omega)$ and $g \in \mathcal{B}'(\partial\Omega)$:

$$\begin{aligned} \mathcal{D}_{\partial\Omega}f|_{\Omega} &= \mathcal{S}_{\partial\Omega} \left. g \right|_{\Omega} &\iff \mathcal{A}_{\partial\Omega,i}f = g, \\ \mathcal{D}_{\partial\Omega}f|_{\Omega^c} &= \mathcal{S}_{\partial\Omega} \left. g \right|_{\Omega^c} &\iff \mathcal{A}_{\partial\Omega,e}f = g. \end{aligned}$$

Those relations allow to prove the Poincaré-Steklov operator for the bounded problem (Subsection 3.2) and for the transmission problem are connected: $\mathcal{A} = -\mathcal{A}_{\partial\Omega,e}$.

B.5 Norm estimates and Neumann series

Using the boundary operators defined in Subsection 4.3, let us define the following norms on $\mathcal{B}'(\partial\Omega)$ and $\mathcal{B}(\partial\Omega)$ respectively:

$$\|\cdot\|_{\mathcal{V}_{\partial\Omega}}^{2} := \langle\cdot, \mathcal{V}_{\partial\Omega} \cdot\rangle_{\mathcal{B}',\mathcal{B}} \quad \text{and} \quad \|\cdot\|_{\mathcal{V}_{\partial\Omega}^{-1}}^{2} := \langle\mathcal{V}_{\partial\Omega}^{-1} \cdot, \cdot\rangle_{\mathcal{B}',\mathcal{B}}, \tag{B.51}$$

well-defined using the variational formulations of the transmission problems:

$$\forall g \in \mathcal{B}'(\partial\Omega), \quad \|g\|^2_{\mathcal{V}_{\partial\Omega}} = \langle g, \mathcal{V}_{\partial\Omega}g \rangle_{\mathcal{B}',\mathcal{B}} = \|\mathcal{S}_{\partial\Omega}g\|^2_{H^1_{\partial\Omega}(\mathbb{R}^n)}. \tag{B.52}$$

By continuity of the single layer potential operator and of its inverse (which is non other than $\left[\frac{\partial}{\partial \nu}\right]$), it holds, for some $\alpha > 0$:

$$\forall g \in \mathcal{B}'(\partial\Omega), \quad \frac{1}{\alpha} \|g\|^2_{\mathcal{B}'(\partial\Omega)} \le \|g\|^2_{\mathcal{V}_{\partial\Omega}} \le \alpha \|g\|^2_{\mathcal{B}'(\partial\Omega)}. \tag{B.53}$$

In the same way, let us define the following norm on $\mathcal{B}(\partial\Omega)$:

$$\|\cdot\|_{\mathcal{W}_{\partial\Omega}}^2 := \langle \mathcal{W}_{\partial\Omega} \cdot, \cdot \rangle_{\mathcal{B}', \mathcal{B}}.$$

It holds:

$$\forall f \in \mathcal{B}(\partial\Omega), \quad \|f\|_{\mathcal{W}_{\partial\Omega}}^2 = \langle \mathcal{W}_{\partial\Omega}f, f \rangle_{\mathcal{B}', \mathcal{B}} = \|\mathcal{D}_{\partial\Omega}f\|_{H^1_{\partial\Omega}(\mathbb{R}^n)}^2. \tag{B.54}$$

By continuity of the double layer potential operator and of its inverse (which is non other than $-[[Tr \cdot]]$), it holds, for some $\beta > 0$:

$$\forall f \in \mathcal{B}(\partial\Omega), \quad \frac{1}{\beta} \|f\|_{\mathcal{B}(\partial\Omega)}^2 \le \|f\|_{\mathcal{W}_{\partial\Omega}}^2 \le \beta \|f\|_{\mathcal{B}(\partial\Omega)}^2. \tag{B.55}$$

We can also define the norms $\|\cdot\|_{\tilde{\mathcal{V}}_{\partial\Omega}}$ on $\mathcal{B}(\partial\Omega)$, and $\|\cdot\|_{\tilde{\mathcal{V}}_{\partial\Omega}^{-1}}$ and $\|\cdot\|_{\tilde{\mathcal{W}}_{\partial\Omega}}$ on $\mathcal{B}'_0(\partial\Omega)$ for the problem associated to $-\Delta$, and yield the same kind of inequalities as (B.53) and (B.55) with constants $\tilde{\alpha}, \tilde{\beta} > 0$.

The norm equivalences (B.53) and (B.55) (and their restatement for $-\Delta$) place us in the framework of [89], provided the notations are adapted to the case of extension domains, which allows to state the following theorem.

Theorem B.29. Let Ω be a two-sided extension domain of \mathbb{R}^n .

- (i) The layer potential operators $\mathcal{S}_{\partial\Omega} : (\mathcal{B}'(\partial\Omega), \|\cdot\|_{\mathcal{V}_{\partial\Omega}}) \to \mathcal{S}_{\partial\Omega}(\mathcal{B}'(\partial\Omega))$ and $\mathcal{D}_{\partial\Omega} : (\mathcal{B}(\partial\Omega), \|\cdot\|_{\mathcal{W}_{\partial\Omega}}) \to \mathcal{D}_{\partial\Omega}(\mathcal{B}(\partial\Omega))$ define isometries;
- (ii) The layer potential operators $\tilde{\mathcal{S}}_{\partial\Omega} : (\mathcal{B}'_0(\partial\Omega), \|\cdot\|_{\tilde{\mathcal{V}}_{\partial\Omega}}) \to \tilde{\mathcal{S}}_{\partial\Omega}(\mathcal{B}'_0(\partial\Omega))$ and $\tilde{\mathcal{D}}_{\partial\Omega} : (\mathcal{B}(\partial\Omega), \|\cdot\|_{\tilde{\mathcal{W}}_{\partial\Omega}}) \to \tilde{\mathcal{D}}_{\partial\Omega}(\mathcal{B}(\partial\Omega))$ define isometries;

(iii) For all
$$f \in \mathcal{B}(\partial \Omega)$$
 and $\tilde{f} \in \mathcal{B}_0(\partial \Omega)$, it holds:

$$(1-c)\|f\|_{\mathcal{V}_{\partial\Omega}^{-1}} \le \left\| \left(\pm \frac{1}{2}I + \mathcal{K}_{\partial\Omega} \right) f \right\|_{\mathcal{V}_{\partial\Omega}^{-1}} \le c \|f\|_{\mathcal{V}_{\partial\Omega}^{-1}},$$

and

$$(1-\tilde{c})\|\tilde{f}\|_{\tilde{\mathcal{V}}_{\partial\Omega}^{-1}} \le \left\| \left(\pm \frac{1}{2}I + \tilde{\mathcal{K}}_{\partial\Omega} \right) \tilde{f} \right\|_{\tilde{\mathcal{V}}_{\partial\Omega}^{-1}} \le \tilde{c}\|\tilde{f}\|_{\tilde{\mathcal{V}}_{\partial\Omega}^{-1}},$$

where

$$c = \frac{1}{2} + \sqrt{\frac{1}{4} - \frac{1}{\alpha\beta}} < 1 \quad and \quad \tilde{c} = \frac{1}{2} + \sqrt{\frac{1}{4} - \frac{1}{\tilde{\alpha}\tilde{\beta}}} < 1,$$

with the constants α , β , $\tilde{\alpha}$ and $\tilde{\beta}$ from (B.53), (B.55) and their restatements for $-\Delta$ chosen large enough ($\alpha\beta, \tilde{\alpha}\tilde{\beta} > 4$). (iv) The operators $\pm \frac{1}{2}I + \mathcal{K}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}(\partial\Omega)$ and $\pm \frac{1}{2}I + \tilde{\mathcal{K}}_{\partial\Omega} : \mathcal{B}_0(\partial\Omega) \to \mathcal{B}(\partial\Omega)$ $\mathcal{B}_0(\partial\Omega)$ are isomorphisms.

Proof. Points (i) and (ii) follow from equations (B.52) and (B.54), as well as their restatements for $-\Delta$. Point *(iii)* is obtained following [89]: the norm equivalences (B.53) and (B.55) as well as the relations between the layer potentials and the Neumann-Poincaré operators yield the result. The difference between $\frac{1}{2}$ and $-\frac{1}{2}$ simply comes from the consideration of either the interior trace or the exterior trace operators. Point (iii) implies the operators $\pm \frac{1}{2}I + \mathcal{K}_{\partial\Omega} : \mathcal{B}(\partial\Omega) \to \mathcal{B}(\partial\Omega) \text{ and } \pm \frac{1}{2}I + \tilde{\mathcal{K}}_{\partial\Omega} : \mathcal{B}_0(\partial\Omega) \to \mathcal{B}_0(\partial\Omega) \text{ are injective}$ and have a closed range. Since their adjoints are injective by Lemma B.26, Point *(iv)* holds by Banach's closed range theorem.

We use a result from perturbation theory [37, Theorem 1.2.9] (Lemma B.30) below) to establish spectral properties of the Neumann-Poincaré operators.

Lemma B.30. Let X be a Banach space and $A \in \mathcal{L}(X)$ be an isomorphism. Let $\varphi \in \mathcal{L}(X)$ be such that $\|\varphi\|_{\mathcal{L}(X)} < \|A^{-1}\|_{\mathcal{L}(X)}^{-1}$. Then $A + \varphi \in \mathcal{L}(X)$ is an isomorphism.

Theorem B.31. Let Ω be a two-sided extension domain of \mathbb{R}^n . Then, for all $\lambda \in]-\infty, \frac{1}{2}] \cup [\frac{1}{2}, +\infty[:$

- (i) the operators $\lambda I + \mathcal{K}_{\partial\Omega}$ and $\lambda I + \mathcal{K}^*_{\partial\Omega}$ are invertible on $\mathcal{B}(\partial\Omega)$ and $\mathcal{B}'(\partial\Omega)$ respectively;
- (ii) the operators $\lambda I + \tilde{\mathcal{K}}_{\partial\Omega}$ and $\lambda I + \tilde{\mathcal{K}}^*_{\partial\Omega}$ are invertible on $\mathcal{B}_0(\partial\Omega)$ and $\mathcal{B}'_0(\partial\Omega)$ respectively.

Proof. To extend the result from Theorem B.29, Point (iv) to other values of λ , we use Lemma B.30 in two steps. For any $\lambda \neq -\frac{1}{2}$, it holds:

$$\lambda I + \mathcal{K}_{\partial\Omega} = \left(\lambda + \frac{1}{2}\right)I + \left(-\frac{1}{2}I + \mathcal{K}_{\partial\Omega}\right). \tag{B.56}$$

By Theorem B.29, $\| -\frac{1}{2}I + \mathcal{K}_{\partial\Omega} \| < 1$. Hence, by Lemma B.30, thinking of $-\frac{1}{2}I + \mathcal{K}_{\partial\Omega}$ as a perturbation φ of $(\lambda + \frac{1}{2})I$, $\lambda I + \mathcal{K}_{\partial\Omega}$ is invertible anytime $|\lambda + \frac{1}{2}| \geq 1$, which is valid when $\lambda \geq \frac{1}{2}$. Proceeding in the same way with $\frac{1}{2}I + \mathcal{K}_{\partial\Omega}$ instead, we find that $\lambda I + \mathcal{K}_{\partial\Omega}$ is invertible for $\lambda \leq -\frac{1}{2}$. Therefore, the real spectrum of $\mathcal{K}_{\partial\Omega}$ is included in $]-\frac{1}{2},\frac{1}{2}[$. By Banach's closed range theorem, the same holds for $\mathcal{K}^*_{\partial\Omega}$.

Proceeding in the same way, we prove the result for $\tilde{\mathcal{K}}_{\partial\Omega}$ and $\tilde{\mathcal{K}}^*_{\partial\Omega}$

Let us now look into the constants from the norm equivalences (B.53)and (B.55). The optimal constants for those inequalities are none other than the norms of the operators $\mathcal{V}_{\partial\Omega}$, $\mathcal{W}_{\partial\Omega}$ and their inverses. Consider the following norm on $\mathcal{B}(\partial \Omega)$:

$$\|\cdot\|_{\mathrm{Tr}} = \min\left\{ \|v\|_{H^1(\mathbb{R}^n)} \mid \mathrm{Tr}\, v = \cdot \right\}$$

By energy minimization, it follows that:

 $\forall v \in V_{1,\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n), \quad \|\mathrm{Tr}\,v\|_{\mathrm{Tr}} = \|v\|_{H^1(\mathbb{R}^n)},$

where $V_{1,\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n)$ is defined on pages iii–vi. Therefore, it holds:

$$\|\mathcal{V}_{\partial\Omega}\|_{\mathcal{L}(\mathcal{B}',(\mathcal{B},\|\cdot\|_{\mathrm{Tr}}))} = \|\mathcal{S}_{\partial\Omega}\|_{\mathcal{L}(\mathcal{B}',H^{1}_{\partial\Omega})}$$

For all $g \in \mathcal{B}'(\partial \Omega)$, it holds:

$$\forall v \in H^1(\mathbb{R}^n), \quad \langle \mathcal{S}_{\partial\Omega} g, v \rangle_{H^1(\mathbb{R}^n)} = \langle g, \operatorname{Tr} v \rangle_{\mathcal{B}'(\partial\Omega), \mathcal{B}(\partial\Omega)}$$

hence:

$$\|\mathcal{V}_{\partial\Omega}\|_{\mathcal{L}(\mathcal{B}',(\mathcal{B},\|\cdot\|_{\mathrm{Tr}}))} = \|\mathcal{S}_{\partial\Omega}\|_{\mathcal{L}(\mathcal{B}',H^{1}_{\partial\Omega})} = 1$$

It holds $\mathcal{S}_{\partial\Omega}^{-1} = \llbracket \frac{\partial}{\partial \nu} \rrbracket : V_{1,\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n) \to \mathcal{B}'(\partial\Omega)$. Let $u \in V_{1,\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n)$. Then, for all $v \in V_{1,\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n)$, it holds:

$$\left| \left\langle \left[\left[\frac{\partial u}{\partial \nu} \right] \right], \operatorname{Tr} v \right\rangle_{\mathcal{B}', \mathcal{B}} \right| = \left| \langle u, v \rangle_{H^1(\mathbb{R}^n)} \right|$$
$$\leq \| u \|_{H^1(\mathbb{R}^n)} \| v \|_{H^1(\mathbb{R}^n)} = \| u \|_{H^1(\mathbb{R}^n)} \| \operatorname{Tr} v \|_{\operatorname{Tr}}$$

Hence (taking u = v), $\|\mathcal{S}_{\partial\Omega}^{-1}\|_{\mathcal{L}(H^1_{\partial\Omega},\mathcal{B}')} = 1$.

We give optimal constants for the equivalence between the norms $\|\cdot\|_{\mathcal{B}(\partial\Omega)}$ and $\|\cdot\|_{\mathrm{Tr}}$.

Proposition B.32. Let Ω be a two-sided extension domain. Let E_{Ω} and E_{Ω^c} be the 1-harmonic extension operators from Ω and Ω^c respectively to \mathbb{R}^n . Then it holds:

$$\sqrt{\left(1 + \frac{1}{\|\mathbf{E}_{\Omega^c}\|^2 - 1}\right)} \|\cdot\|_{\mathcal{B}(\partial\Omega)} \le \|\cdot\|_{\mathrm{Tr}} \le \|\mathbf{E}_{\Omega}\| \|\cdot\|_{\mathcal{B}(\partial\Omega)},\tag{B.57}$$

with optimal constants.

Proof. Using the isometric properties of both norms, if $v \in V_{1,\partial\Omega,[0]_{\mathrm{Tr}}}(\mathbb{R}^n)$, then (B.57) is equivalent to:

$$\sqrt{\left(1 + \frac{1}{\|\mathbf{E}_{\Omega^c}\|^2 - 1}\right)} \|v|_{\Omega}\|_{H^1(\Omega)} \le \|v\|_{H^1(\mathbb{R}^n)} \le \|\mathbf{E}_{\Omega}\| \|v|_{\Omega}\|_{H^1(\Omega)}, \quad (B.58)$$

The second inequality and optimality of the constant follow directly from the boundedness of E_{Ω} . Moreover, it holds:

$$\frac{\|v|_{\Omega^c}\|_{H^1(\Omega^c)}^2}{\|v\|_{H^1(\mathbb{R}^n)}^2} \ge \frac{1}{\|\mathbf{E}_{\Omega^c}\|^2} \quad \text{and} \quad \|v\|_{H^1(\mathbb{R}^n)}^2 = \|v|_{\Omega}\|_{H^1(\Omega)}^2 + \|v|_{\Omega^c}\|_{H^1(\Omega^c)}^2,$$

hence

$$1 - \frac{\|v\|_{\Omega}\|_{H^1(\Omega)}^2}{\|v\|_{H^1(\mathbb{R}^n)}^2} \ge \frac{1}{\|\mathbf{E}_{\Omega^c}\|^2}.$$

The first inequality and the optimality of the constant follow.

For all $f \in \mathcal{B}(\partial \Omega)$, it holds:

$$\left\|\mathcal{D}_{\partial\Omega}f\right\|_{H^{1}_{\partial\Omega}}^{2} = \left|\left\langle\frac{\partial}{\partial\nu}\mathcal{D}_{\partial\Omega}f,f\right\rangle_{\mathcal{B}',\mathcal{B}}\right| \leq \left\|\frac{\partial}{\partial\nu}\mathcal{D}_{\partial\Omega}f\right\|_{\mathcal{B}'} \|f\|_{\mathcal{B}'}$$

Moreover, since the interior Poincaré-Steklov operator is an isometry for the norm $\|\cdot\|_{\mathcal{B}(\partial\Omega)}$, then it holds:

$$\left\| \frac{\partial}{\partial \nu} \mathcal{D}_{\partial \Omega} f \right\|_{\mathcal{B}'} = \| \operatorname{Tr}_i \mathcal{D}_{\partial \Omega} f \|_{\mathcal{B}} = \| \mathcal{D}_{\partial \Omega} f |_{\Omega} \|_{H^1(\Omega)}$$
$$\leq \sqrt{\left(1 + \frac{1}{\| \operatorname{E}_{\Omega^c} \|^2 - 1} \right)} \| \mathcal{D}_{\partial \Omega} f \|_{H^1(\mathbb{R}^n)},$$

by (B.58), hence $\|\mathcal{D}_{\partial\Omega}\| \leq \sqrt{\left(1 + \frac{1}{\|E_{\Omega^c}\|^2 - 1}\right)}$. It holds:

$$\left| \left\langle \frac{\partial u}{\partial \nu}, \llbracket \operatorname{Tr} v \rrbracket \right\rangle_{\mathcal{B}', \mathcal{B}} \right| = \left| \langle u, v \rangle_{H^1(\mathbb{R}^n)} \right|$$
$$\leq \| u \|_{H^1(\mathbb{R}^n)} \| v \|_{H^1(\mathbb{R}^n)}$$
$$\leq \| \operatorname{E}_{\Omega} \| \left\| \frac{\partial u}{\partial \nu} \right\|_{\mathcal{B}'} \| v \|_{H^1(\mathbb{R}^n)}$$

hence $\| \llbracket \operatorname{Tr} \cdot \rrbracket \| \leq \| \operatorname{E}_{\Omega} \|$. It holds for *u* harmonic on $\mathbb{R}^n \setminus \partial \Omega$:

$$\left\|\frac{\partial u}{\partial \nu}\right\|_{\mathcal{B}'} = \|\operatorname{Tr}_{i} u\|_{\mathcal{B}} = \|u|_{\Omega}\|_{H^{1}(\Omega)} \leq \sqrt{\left(1 + \frac{1}{\|\operatorname{E}_{\Omega^{c}}\|^{2} - 1}\right)} \|u\|_{H^{1}(\mathbb{R}^{n})}$$

and

$$\|u\|_{H^1(\mathbb{R}^n)} \le \|\mathbf{E}_{\Omega}\| \left\| \frac{\partial u}{\partial \nu} \right\|_{\mathcal{B}'}$$

Hence:

$$\|\mathcal{W}_{\partial\Omega}\| \le 1 + \frac{1}{\|\mathbf{E}_{\Omega^c}\|^2 - 1},$$

and

 $\|\mathcal{W}_{\partial\Omega}^{-1}\| \leq \|E_{\Omega}\|^2.$

Similar estimates hold for the operators associated to $-\Delta$.

B.6 Imaging

B.6.1 Representation formula

In this subsection, we aim at generalizing to two-sided extension domains imaging results established in the case of bounded Lipschitz domains [1, 61]. Since we will be considering two two-sided extension domains of \mathbb{R}^n , Ω and D, we shall use the following notations:

- $\operatorname{Tr}_{i}^{\partial\Omega}$: $H^{1}(\Omega) \to \mathcal{B}(\partial\Omega)$ and $\operatorname{Tr}_{e}^{\partial\Omega}$: $H^{1}(\Omega^{c}) \to \mathcal{B}(\partial\Omega)$ for the interior and exterior trace operators on the boundary of Ω , according to Definitions 2.8 and 2.10,
- $\frac{\partial_i u}{\partial \nu}\Big|_{\partial\Omega} \in \mathcal{B}'(\partial\Omega)$ and $\frac{\partial_e u}{\partial \nu}\Big|_{\partial\Omega} \in \mathcal{B}'(\partial\Omega)$ for the interior and exterior normal derivatives of $u \in H^1_{\Delta}(\mathbb{R}^n \setminus \partial\Omega)$ on the boundary of Ω , according to Definitions 2.12 and 2.13,
- $\llbracket \operatorname{Tr} u \rrbracket_{\partial\Omega} := \operatorname{Tr}_i^{\partial\Omega} u \operatorname{Tr}_e^{\partial\Omega} u \in \mathcal{B}(\partial\Omega)$ for the jump in trace across $\partial\Omega$, according to Definition 2.14,
- $\left[\left[\frac{\partial u}{\partial \nu} \right] \right]_{\partial \Omega} := \left. \frac{\partial_i u}{\partial \nu} \right|_{\partial \Omega} \left. \frac{\partial_e u}{\partial \nu} \right|_{\partial \Omega} \in \mathcal{B}'(\partial \Omega) \text{ for the jump in normal derivative across } \partial \Omega, \text{ according to Definition 2.14,}$

as well as similar notations for other two-sided extension domains, such as D.

Let Ω and D be two two-sided extension domains of \mathbb{R}^n such that $D \subset \subset \Omega$. For the transmission problem (4.1) for k = 0 across ∂D , we can define the single layer potential operator $\tilde{\mathcal{S}}_D : \mathcal{B}'_0(\partial D) \to H^1_{\partial D}(\mathbb{R}^n)$ and the double layer potential operator $\tilde{\mathcal{D}}_D : \mathcal{B}(\partial D) \to H^1(\mathbb{R}^n \setminus \partial D)$ on D, as well as the associated Neumann-Poincaré operators $\tilde{\mathcal{K}}_{\partial D} : \mathcal{B}(\partial D) \to \mathcal{B}(\partial D)$ and $\tilde{\mathcal{K}}^*_{\partial D} : \mathcal{B}'_0(\partial D) \to$ $\mathcal{B}'_0(\partial D)$. Let $k \in]0, 1[\cup]1, +\infty[$ be the conductivity of the domain D (the conductivity of Ω is assumed to be equal to 1). On the space:

$$H^{1}_{\partial D}(\Omega) := \left\{ u \in L^{2}(\Omega) \mid u|_{D} \in H^{1}(D), \ u|_{\Omega \setminus \overline{D}} \in H^{1}(\Omega \setminus \overline{D}) \right\},$$

we will consider the following two-phased transmission problem:

$$\begin{cases} \nabla \cdot \left(\left(1 + (k-1) \mathbb{1}_D \right) \nabla u \right) = 0 \quad \text{on } \Omega, \\ \frac{\partial_i u}{\partial \nu} \Big|_{\partial \Omega} = g, \\ \int_{\Omega} u \, \mathrm{d}x = 0, \end{cases}$$
(B.59)

where $g \in \mathcal{B}'_0(\partial \Omega)$. The last condition ensures the uniqueness of the solution. The associated variational formulation is:

$$\forall v \in H^{1}_{\partial D,\langle 0 \rangle}(\Omega), \ \int_{\Omega} \left(1 + (k-1) \mathbb{1}_{D} \right) \nabla u \cdot \nabla v \, \mathrm{d}x$$
$$= \left\langle \left(1 + (k-1) \mathbb{1}_{D} \right) g, \operatorname{Tr}_{i}^{\partial \Omega} v \right\rangle_{\mathcal{B}',\mathcal{B}'}, \quad (B.60)$$

where $H^1_{\partial D,\langle 0\rangle}(\Omega)$ is defined on pages iii–vi. Let us start by proving the well-posedness of the variational formulation (B.60).

Proposition B.33. The variational formulation (B.60) of the two-phased transmission problem is well-posed on $H^1_{\partial D,\langle 0\rangle}(\Omega)$.

Proof. $H^1_{\partial D,\langle 0\rangle}(\Omega)$ is a Hilbert space with $(u, v) \mapsto \langle \nabla u, \nabla v \rangle_{L^2(\Omega)}$ as an inner product which, by Poincaré's inequality, is equivalent on $H^1_{\partial D,\langle 0\rangle}(\Omega)$ to the usual inner product on $H^1(\Omega)$. Since $(1 + (k - 1))\mathbb{1}_D$ is bounded on \mathbb{R}^n , the mapping:

$$v \longmapsto \left\langle \left(1 + (k-1) \mathbb{1}_D \right) g, \operatorname{Tr}_i^{\partial \Omega} v \right\rangle_{\mathcal{B}', \mathcal{B}}$$

is continuous on $H^1_{\partial D,\langle 0\rangle}(\Omega)$, and there exists C > 0 such that:

$$\forall v \in H^1_{\partial D,\langle 0 \rangle}(\Omega), \quad \left| \left\langle \left(1 + (k-1) \mathbb{1}_D \right) g, \operatorname{Tr}_i^{\partial \Omega} v \right\rangle_{\mathcal{B}', \mathcal{B}} \right| \le C \|g\|_{\mathcal{B}'} \|v\|_{H^1_{\partial D,\langle 0 \rangle}}.$$

In addition, for $u, v \in H^1_{\partial D, \langle 0 \rangle}(\Omega)$, it holds:

$$\left| \int_{\Omega} \left(1 + (k-1) \mathbb{1}_D \right) \nabla u \cdot \nabla v \, \mathrm{d}x \right| \le (k+1) \lambda^{(n)}(\Omega) \, \|\nabla u\|_{L^2(\Omega)^n} \|\nabla v\|_{L^2(\Omega)^n}$$

one the one hand, and

$$\int_{\Omega} \left(1 + (k-1)\mathbb{1}_D \right) \nabla u \cdot \nabla u \, \mathrm{d}x \ge \min(1,k) \lambda^{(n)}(\Omega) \, \|\nabla u\|_{L^2(\Omega)^n}^2$$

on the other hand. Therefore, applying Lax-Milgram's theorem, we can deduce the existence of a unique $u \in H^1_{\partial D,\langle 0 \rangle}(\Omega)$ solution to the variational formulation (B.60). Choosing v = u in the variational formulation, it holds:

$$\min(1,k)\lambda^{(n)}(\Omega) \|u\|_{H^1_{\partial D,\langle 0\rangle}}^2 \le C \|g\|_{\mathcal{B}'} \|u\|_{H^1_{\partial D,\langle 0\rangle}},$$

where C > 0 is the previous continuity constant (which does not depend on u), that is:

$$\min(1,k)\lambda^{(n)}(\Omega) \|u\|_{H^1_{\partial D,\langle 0\rangle}} \le C \|g\|_{\mathcal{B}'},$$

hence the well-posedness of (B.60).

Knowing that the two-phased transmission problem (B.60) is well-posed, we seek a representation formula for the solution of that problem and update [1, Theorem 2.17] (see also [59, 60]).

Theorem B.34. (Representation formula) If Ω and D are two-sided extension domains of \mathbb{R}^n with $D \subset \Omega$, and if $k \in]0,1[\cup]1,+\infty[$, then the unique weak solution to the two-phased transmission problem (B.59) associated to $g \in \mathcal{B}'_0(\partial \Omega)$ is:

$$u = H - \hat{\mathcal{S}}_{\partial D}\varphi \tag{B.61}$$

where

$$H = \tilde{\mathcal{S}}_{\partial\Omega} g - \tilde{\mathcal{D}}_{\partial\Omega} f$$

with $\tilde{\mathcal{S}}_{\partial D}$ defined on $\mathcal{B}'_0(\partial D)$, $\tilde{\mathcal{S}}_{\partial \Omega}$ on $\mathcal{B}'_0(\partial \Omega)$, $\tilde{\mathcal{D}}_{\partial \Omega}$ on $\mathcal{B}(\partial \Omega)$, $f := \operatorname{Tr}_i^{\partial \Omega} u \in \mathcal{B}(\partial \Omega)$, and $\varphi \in \mathcal{B}'_0(\partial D)$ such that:

$$\left(\frac{k+1}{2(k-1)}I + \tilde{\mathcal{K}}_{\partial D}^*\right)\varphi = \left.\frac{\partial H}{\partial\nu}\right|_{\partial D},\tag{B.62}$$

-		
-		

with $\tilde{\mathcal{K}}^*_{\partial D} : \mathcal{B}'_0(\partial D) \to \mathcal{B}'_0(\partial D)$. *H* is the harmonic part of the solution, while $-\tilde{\mathcal{S}}_{\partial D}\varphi$ corresponds to the refraction part. A such decomposition is unique. In addition, it holds:

$$H - \tilde{\mathcal{S}}_{\partial D} \varphi = 0 \quad on \ \mathbb{R}^n \setminus \overline{\Omega}.$$

Proof. Let us consider the following two-phased transmission problem on the space $W_{\partial D,\partial\Omega}(\mathbb{R}^n)$, which is defined by:

$$W_{\partial D,\partial\Omega}(\mathbb{R}^n) := \left\{ u \in L^2_{loc}(\mathbb{R}^n) \mid \nabla u \mid_D \in L^2(D), \ \nabla u \mid_{\Omega \setminus \overline{D}} \in L^2(\Omega \setminus \overline{D}), \\ \nabla u \mid_{\Omega^c} \in L^2(\Omega^c), \ \langle u \rangle_{\infty} = 0 \right\},$$

$$\begin{cases} \nabla \cdot \left(\left(1 + (k-1) \mathbb{1}_D \right) \nabla u \right) = 0 & \text{on } \mathbb{R}^n \backslash \partial \Omega, \\ \llbracket \operatorname{Tr} u \rrbracket_{\partial \Omega} = f, \\ \llbracket \frac{\partial u}{\partial \nu} \rrbracket_{\partial \Omega} = g, \end{cases}$$
(B.63)

where $k \in [0, 1[\cup]], +\infty[$, $g \in \mathcal{B}'_0(\partial\Omega)$, and $f := \operatorname{Tr}_i^{\partial\Omega} u_\Omega \in \mathcal{B}(\partial\Omega)$, with u_Ω the solution to the transmission problem on Ω (B.59) associated to g.

Problem (B.63) has a unique solution on $W_{\partial D,\partial\Omega}(\mathbb{R}^n)$. Indeed, if v is a solution of that problem for homogeneous transmission conditions, then for all r > 0, it holds:

$$\begin{split} \int_{B_r(0)} |\nabla v|^2 \, \mathrm{d}x &\leq \frac{1+k}{k} \int_{B_r(0)} \left(1+(k-1)\mathbb{1}_D \right) |\nabla v|^2 \, \mathrm{d}x \\ &\leq \frac{1+k}{k} \left\langle \left(1+(k-1)\mathbb{1}_D \right) \frac{\partial_i v}{\partial \nu}, \operatorname{Tr}_i^{B_r(0)} v \right\rangle_{\mathcal{B}', \mathcal{B}} \\ &\leq -\frac{1+k}{k} \int_{B_r(0)^c} \left(1+(k-1)\mathbb{1}_D \right) |\nabla v|^2 \, \mathrm{d}x \\ &\leq -\frac{1+k}{k} \int_{B_r(0)^c} |\nabla v|^2 \, \mathrm{d}x \\ &\leq 0, \end{split}$$

applying Green's formula on $B_r(0)$ and $B_r(0)^c$. That being true for all r > 0, v is constant. Since $v \in W_{\partial D,\partial\Omega}(\mathbb{R}^n)$, it holds v = 0. Hence the uniqueness of the solution on $W_{\partial D,\partial\Omega}(\mathbb{R}^n)$. On the one hand, the function u_0 defined as $u_0|_{\Omega} = u_{\Omega}$ and $u_0|_{\Omega^c} = 0$ is a solution to the transmission problem (B.63). On the other hand, $\tilde{S}_{\partial\Omega}g - \tilde{D}_{\partial\Omega}f - \tilde{S}_{\partial D}\varphi$ is also a solution of that problem. Indeed, if u is a solution to the problem, then Green's formula on Ω states that:

$$\forall v \in H^{1}(\Omega), \quad \left\langle \left. \frac{\partial_{i} u}{\partial \nu} \right|_{\partial \Omega}, \operatorname{Tr}_{i}^{\partial \Omega} v \right\rangle_{\mathcal{B}', \mathcal{B}(\partial \Omega)} - \int_{\Omega \setminus \overline{D}} \nabla u \cdot \nabla v \, \mathrm{d}x = k \int_{D} \nabla u \cdot \nabla v \, \mathrm{d}x$$
(B.64)

Since Green's formulae on D and $\Omega \setminus \overline{D}$ respectively state that, for all $v \in H^1(\Omega)$,

$$\begin{cases} \int_{D} \nabla u \cdot \nabla v \, \mathrm{d}x = \left\langle \frac{\partial_{i} u}{\partial \nu} \right|_{\partial D}, \operatorname{Tr}^{\partial D} v \right\rangle_{\mathcal{B}', \mathcal{B}(\partial D)} \\ \int_{\Omega \setminus \overline{D}} \nabla u \cdot \nabla v \, \mathrm{d}x = \left\langle \frac{\partial_{i} u}{\partial \nu} \right|_{\partial \Omega}, \operatorname{Tr}^{\partial \Omega}_{i} v \right\rangle_{\mathcal{B}', \mathcal{B}(\partial \Omega)} - \left\langle \frac{\partial_{e} u}{\partial \nu} \right|_{\partial D}, \operatorname{Tr}^{\partial D} v \right\rangle_{\mathcal{B}', \mathcal{B}(\partial D)}, \end{cases}$$

then (B.64) becomes:

$$\forall v \in H^1(\Omega), \quad \left\langle k \left. \frac{\partial_i u}{\partial \nu} \right|_{\partial \Omega} - \left. \frac{\partial_e u}{\partial \nu} \right|_{\partial \Omega}, \operatorname{Tr}^{\partial D} v \right\rangle_{\mathcal{B}', \mathcal{B}(\partial D)} = 0,$$

that is, since the trace operator $\operatorname{Tr}^{\partial D} : H^1(\Omega) \to \mathcal{B}(\partial D)$ is onto:

$$k \left. \frac{\partial_i u}{\partial \nu} \right|_{\partial D} - \left. \frac{\partial_e u}{\partial \nu} \right|_{\partial D} = 0$$

Since $u = H - \tilde{S}_{\partial D}\varphi$, it holds:

$$k \left. \frac{\partial_i (\tilde{\mathcal{S}}_{\partial D} \varphi)}{\partial \nu} \right|_{\partial D} - \left. \frac{\partial_e (\tilde{\mathcal{S}}_{\partial D} \varphi)}{\partial \nu} \right|_{\partial D} = (k-1) \left. \frac{\partial H}{\partial \nu} \right|_{\partial D}$$

since H has no jump in normal derivative across D by Weyl's lemma, hence:

$$\left(\frac{k+1}{2(k-1)}I + \mathcal{K}_{\partial D}^*\right)\varphi = \left.\frac{\partial H}{\partial\nu}\right|_{\partial D}$$

H is harmonic on *D*, therefore, by Green's formula, $\frac{\partial H}{\partial \nu}\Big|_{\partial D} \in \mathcal{B}'_0(\partial D)$. Since $(\frac{k+1}{2(k-1)}I + \mathcal{K}^*_{\partial D})$ is invertible on $\mathcal{B}'_0(\partial D)$, such a $\varphi \in \mathcal{B}'_0(\partial D)$ exists.

We will now prove the uniqueness of the decomposition. Let us suppose that $H' \in H^1_{\partial\Omega}(\mathbb{R}^n)$ is harmonic in $\Omega, \varphi' \in \mathcal{B}'_0(\partial D)$, and that it holds:

$$H - \tilde{\mathcal{S}}_{\partial D}\varphi = H' - \tilde{\mathcal{S}}_{\partial D}\varphi' \quad \text{in } \Omega.$$

Then $\tilde{S}_{\partial D}(\varphi - \varphi')$ is harmonic on Ω , therefore, by Weyl's lemma it is real analytic: $\left[\left[\frac{\partial}{\partial \nu} \tilde{S}_{\partial D}(\varphi - \varphi') \right] \right]_{\partial D} = 0$, hence $\varphi = \varphi'$ and H = H'.

That representation formula can be simplified if n = 2 in the case of a disk.

Corollary B.35. If Ω is a two-sided extension domain of \mathbb{R}^2 , if D is a disk compactly included in Ω and if $k \in]0, 1[\cup]1, +\infty[$, then the solution to the two-phased transmission problem (B.59) is given by:

$$u = H - \frac{2(k-1)}{k+1} \tilde{\mathcal{S}}_{\partial D} \left(\frac{\partial H}{\partial \nu} \Big|_{\partial D} \right)$$
(B.65)

where

$$H = \tilde{\mathcal{S}}_{\partial\Omega} g - \tilde{\mathcal{D}}_{\partial\Omega} f$$

with $\tilde{\mathcal{S}}_{\partial D}$ defined on $L^2_0(\partial D, \sigma) \subset \mathcal{B}'_0(\partial D)$, $\tilde{\mathcal{S}}_{\partial \Omega}$ defined on $\mathcal{B}'_0(\partial \Omega)$, $\tilde{\mathcal{D}}_{\partial \Omega}$ defined on $\mathcal{B}(\partial \Omega)$ and $f = \operatorname{Tr}_i u \in \mathcal{B}(\partial \Omega)$. *Proof.* As explained in [1], since D is smooth, $\frac{\partial H}{\partial \nu}|_{\partial D} \in L^2_0(\partial D, \sigma)$. Therefore, φ defined by (B.62) is an element of $L^2_0(\partial D, \sigma)$, and it holds:

$$\tilde{\mathcal{K}}^*_{\partial D}\varphi = 0$$

Therefore, the representation formula (B.61) can be restated as (B.65) if D is a disk of \mathbb{R}^2 .

B.6.2 Subdomain identification

We state here a generalized version of the theorem about the uniqueness of the inclusion with one measurement in the monotonous case which can be found in [1] for Lipschitz boundaries, in the case of a two-sided extension domain Ω , and a two-sided extension subdomain D.

Theorem B.36. (Subdomain identification in the monotonous case) Let Ω be a two-sided extension domain of \mathbb{R}^n . Let $D_1 \subset D_2$ be two two-sided extension domains compactly included in Ω , and let $k \in]0, 1[\cup]1, +\infty[$. Let us denote by u_1 and u_2 the solutions to the two-phased transmission problem (B.59), respectively associated to $D = D_1$ and $D = D_2$. If, for some Neumann condition $g \in \mathcal{B}'_0(\partial\Omega) \setminus \{0\}$, $\operatorname{Tr}_i^{\partial\Omega} u_1 = \operatorname{Tr}_i^{\partial\Omega} u_2$, then $D_1 = D_2$.

Proof. It holds, for all $v \in H^1(\Omega)$:

$$\int_{\Omega} \nabla \cdot \left(\left(1 + (k-1)\mathbb{1}_{D_1} \right) \nabla u_1 \right) v \, \mathrm{d}x = \int_{\Omega} \nabla \cdot \left(\left(1 + (k-1)\mathbb{1}_{D_2} \right) \nabla u_2 \right) v \, \mathrm{d}x.$$

Since $\frac{\partial_i u_1}{\partial \nu}|_{\partial \Omega} = \frac{\partial_i u_2}{\partial \nu}|_{\partial \Omega} = g$ and $D_1, D_2 \subset \Omega$, it holds:

$$\left\langle \left(1 + (k-1)\mathbb{1}_{D_1} \right) \frac{\partial_i u_1}{\partial \nu} \Big|_{\partial \Omega}, \operatorname{Tr}_i^{\partial \Omega} v \right\rangle_{\mathcal{B}', \mathcal{B}} = \left\langle \left(1 + (k-1)\mathbb{1}_{D_2} \right) \frac{\partial_i u_2}{\partial \nu} \Big|_{\partial \Omega}, \operatorname{Tr}_i^{\partial \Omega} v \right\rangle_{\mathcal{B}', \mathcal{B}},$$

hence, by Green's formula:

$$\int_{\Omega} \left(1 + (k-1) \mathbb{1}_{D_1} \right) \nabla u_1 \cdot \nabla v \, \mathrm{d}x = \int_{\Omega} \left(1 + (k-1) \mathbb{1}_{D_2} \right) \nabla u_2 \cdot \nabla v \, \mathrm{d}x, \quad (B.66)$$

that is:

$$\int_{\Omega} \left(1 + (k-1)\mathbb{1}_{D_1} \right) \nabla(u_1 - u_2) \cdot \nabla v \, \mathrm{d}x = (k-1) \int_{D_2 \setminus D_1} \nabla u_2 \cdot \nabla v \, \mathrm{d}x.$$
(B.67)

Only, $\operatorname{Tr}_i^{\partial\Omega} u_1 = \operatorname{Tr}_i^{\partial\Omega} u_2$, so by Green's formula,

$$\int_{\Omega} \left(1 + (k-1)\mathbb{1}_{D_1} \right) \nabla(u_1 - u_2) \cdot \nabla u_1 \, \mathrm{d}x = 0.$$

Applying (B.67) for $v = u_1$ and $v = u_1 - u_2$, we get:

$$0 = (k-1) \int_{D_2 \setminus D_1} \nabla u_2 \cdot \nabla u_1 \, \mathrm{d}x,$$

and

$$\int_{\Omega} \left(1 + (k-1) \mathbb{1}_{D_1} \right) |\nabla(u_1 - u_2)|^2 \, \mathrm{d}x$$

= $(k-1) \int_{D_2 \setminus D_1} \nabla u_2 \cdot \nabla(u_1 - u_2) \, \mathrm{d}x,$

hence:

$$\int_{\Omega} \left(1 + (k-1)\mathbb{1}_{D_1} \right) |\nabla(u_1 - u_2)|^2 \, \mathrm{d}x + (k-1) \int_{D_2 \setminus D_1} |\nabla u_2|^2 \, \mathrm{d}x = 0$$

If k > 1, then $u_1 = u_2$ as the transmission problem (B.60) is well-posed. Therefore, since $g \neq 0$, it holds $D_1 = D_2$.

If k < 1, equation (B.66) lets us show that it holds:

$$\int_{\Omega} \left(1 + (k-1)\mathbb{1}_{D_2} \right) |\nabla(u_1 - u_2)|^2 \,\mathrm{d}x + (1-k) \int_{D_2 \setminus D_1} |\nabla u_1|^2 \,\mathrm{d}x = 0, \quad (B.68)$$

and, for the same reason, $D_1 = D_2$.

Using Theorem B.36 and Corollary B.35, we can deduce the uniqueness of the inclusion in the case of disks when Ω is a two-sided extension domain of \mathbb{R}^2 .

Theorem B.37. (Subdomain identification in the case of disks) Let Ω be a two-sided extension domain of \mathbb{R}^2 . Let D_1 and D_2 be two disks compactly included in Ω , and let $k \in]0, 1[\cup]1, +\infty[$. Let us denote by u_1 and u_2 the solutions to the two-phased transmission problem (B.59), respectively associated to $D = D_1$ and $D = D_2$. If, for some Neumann condition $g \in \mathcal{B}'_0(\partial\Omega) \setminus \{0\}$, $\operatorname{Tr}_i^{\partial\Omega} u_1 = \operatorname{Tr}_i^{\partial\Omega} u_2$, then $D_1 = D_2$.

Proof. Let us denote $f := \operatorname{Tr}_i^{\partial\Omega} u_1 = \operatorname{Tr}_i^{\partial\Omega} u_2$. The representation formula (B.65) states that, for $p \in \{1, 2\}$:

$$u_p = H - \frac{2(k-1)}{k+1} \tilde{\mathcal{S}}_{\partial D_p} \left(\frac{\partial H}{\partial \nu} \bigg|_{\partial D_p} \right)$$

with $H = \tilde{\mathcal{S}}_{\partial\Omega} g - \tilde{\mathcal{D}}_{\partial\Omega} f$.

• If $D_1 \subset D_2$ (or the other way around), then it has been proved in Theorem B.36 that $D_1 = D_2$.

On $\mathbb{R}^2 \setminus (D_1 \cup D_2)$, u_1 and u_2 are both solution to the same problem:

$$\begin{cases} \Delta u_p = 0 & \text{on } \mathbb{R}^2 \backslash (D_1 \cup D_2) \\ \llbracket \operatorname{Tr} u_p \rrbracket_{\partial \Omega} = f, \\ \llbracket \frac{\partial u_p}{\partial \nu} \rrbracket_{\partial \Omega} = g, \end{cases}$$

which has a unique solution on $V_{van}(\mathbb{R}^2 \setminus (D_1 \cup D_2)) := \{ u |_{\mathbb{R}^2 \setminus (D_1 \cup D_2)} \mid u \in V_{van}(\mathbb{R}^2) \}$, therefore $u_1 = u_2$ on $\mathbb{R}^2 \setminus (D_1 \cup D_2)$, hence it holds:

$$\tilde{\mathcal{S}}_{\partial D_1} \left(\left. \frac{\partial H}{\partial \nu} \right|_{\partial D_1} \right) = \tilde{\mathcal{S}}_{\partial D_2} \left(\left. \frac{\partial H}{\partial \nu} \right|_{\partial D_2} \right) \quad \text{on } \mathbb{R}^2 \backslash (D_1 \cup D_2). \tag{B.69}$$

• If $D_1 \cap D_2 = \emptyset$, then, by (B.69), it holds that:

$$\left[\left[\frac{\partial}{\partial \nu} \circ \tilde{\mathcal{S}}_{\partial D_1} \left(\left. \frac{\partial H}{\partial \nu} \right|_{\partial D_1} \right) \right]_{\partial D_1} = \left[\left[\frac{\partial}{\partial \nu} \circ \tilde{\mathcal{S}}_{\partial D_2} \left(\left. \frac{\partial H}{\partial \nu} \right|_{\partial D_2} \right) \right]_{\partial D_1} = 0, \quad (B.70)$$

since $\tilde{\mathcal{S}}_{\partial D_2}\left(\frac{\partial H}{\partial \nu}\Big|_{\partial D_2}\right)$ has no jump in normal derivative across ∂D_1 . Therefore, $\frac{\partial H}{\partial \nu}\Big|_{\partial D_1} = 0$ which means that H is constant \longrightarrow ABSURD. • If $D_1 \cap D_2 \notin \{\varnothing, D_1, D_2\}$, since $\frac{\partial H}{\partial \nu}\Big|_{\partial D_p} \in L^2_0(\partial D_p, \sigma)$ for $p \in \{1, 2\}$, it holds $\tilde{\mathcal{K}}^*_{D_p}\left(\frac{\partial H}{\partial \nu}\Big|_{\partial D_p}\right) = 0$, therefore:

$$\frac{\partial_i}{\partial \nu} \circ \tilde{\mathcal{S}}_{\partial D_p} \left(\frac{\partial H}{\partial \nu} \bigg|_{\partial D_p} \right) = \frac{1}{2} \left. \frac{\partial H}{\partial \nu} \right|_{\partial D_p},$$

hence there exists $c_p \in \mathbb{R}$ such that:

$$\tilde{\mathcal{S}}_{\partial D_p}\left(\left.\frac{\partial H}{\partial \nu}\right|_{\partial D_p}\right) = \frac{1}{2}H + c_p \quad \text{on } D_p.$$

Therefore, denoting $c := c_1 - c_2$, it holds:

$$\tilde{\mathcal{S}}_{\partial D_1}\left(\left.\frac{\partial H}{\partial \nu}\right|_{\partial D_1}\right) = \tilde{\mathcal{S}}_{\partial D_2}\left(\left.\frac{\partial H}{\partial \nu}\right|_{\partial D_2}\right) + c \quad \text{on } D_1 \cap D_2.$$

Since the single layer potential operator it continuous on \mathbb{R}^2 , this equality remains valid on $\partial(D_1 \cap D_2)$. Since equation (B.69) is also valid on $\partial(D_1 \cap D_2)$, we can deduce that c = 0:

$$\tilde{\mathcal{S}}_{\partial D_1}\left(\left.\frac{\partial H}{\partial \nu}\right|_{\partial D_1}\right) = \tilde{\mathcal{S}}_{\partial D_2}\left(\left.\frac{\partial H}{\partial \nu}\right|_{\partial D_2}\right) \quad \text{on } \mathbb{R}^2 \setminus (D_1 \triangle D_2),$$

where $D_1 \triangle D_2$ denotes the symmetric difference between D_1 and D_2 . Therefore, equation (B.70) holds once more and H is constant \longrightarrow ABSURD.

All in all, the only possible configuration is $D_1 = D_2$.

C Hilbert transform and Cauchy integral on extension domains

In Subsection 5.2, we introduce two possible definitions for the Cauchy integral (or equivalently, the Hilbert transform) for two-sided extension domains of \mathbb{C} . Aiming towards proving those definitions coincide, we prove any domain can be approximated by a sequence of Lipschitz domains in Appendix C.1, and in Appendix C.2, we prove the convergence of the Dirichlet (1-)harmonic extension operators, in the framework from [65] which we remind.

C.1 Lipschitz approximation of arbitrary domains

Let us consider a dyadic grid on \mathbb{R}^n :

$$\Pi_k := \{\pi_j, \ j \in \mathbb{Z}^n\},\$$

where

$$\pi_j^k = \Big\{ x \in \mathbb{R}^n \ \Big| \ \forall m \in [[0, n]], \ x \in \big[2^{-k} j_m, 2^{-k} (j_m + 1)\big] \Big\},\$$

for $k \in \mathbb{N}$ and denoting $j = (j_m)_{m \in [0,n]} \in \mathbb{Z}^n$. If $\mathcal{P} \subset \Pi_k$, denote:

$$\cup \mathcal{P} := \bigcup_{\pi \in \mathcal{P}} \pi.$$

Let us define:

$$\overline{\Omega_k} := \max_{\mathcal{P} \subset \Pi_k} \left\{ A = \cup \mathcal{P} \mid A \subset \Omega, \ \mathring{A} \text{ is connected} \right\},\$$
$$\overline{\Omega}_k^+ := \max_{\mathcal{P} \subset \Pi_k} \left\{ B = \cup \mathcal{P} \mid B \subset \Omega^c, \ \mathring{B} \text{ is connected} \right\},$$

for $k \in \mathbb{N}$ large enough (see Lemma C.1), where the maximum is understood in terms of volume. Denote their interiors by:

$$\Omega_k^- := \left(\overline{\Omega}_k^-\right)^\circ \quad \text{and} \quad \Omega_k^+ := \left(\overline{\Omega}_k^+\right)^\circ.$$
(C.1)

Those domains are referred to as interior and exterior dyadic approximations (of order k) of Ω , respectively. The connectedness condition can be written as:

$$\forall \pi_j^k \in \mathcal{P}, \ \exists i \in \mathbb{Z}^d, \quad |j-i| = 1 \text{ and } \pi_i^k \in \mathcal{P}.$$

As an example, Figure 4 represents a dyadic approximation of the Von Koch snowflake.

Naturally, the first step is to make sure the dyadic approximations are welldefined, at least for a grid thin enough (for we are interested in the limit when the grid becomes infinitely thin)

Lemma C.1. Let Ω be an arbitrary bounded domain of \mathbb{R}^n such that $\overline{\Omega}^c$ is connected too. For $k \in \mathbb{N}$ large enough, Ω_k^- and Ω_k^+ are well-defined.



Figure 4: Dyadic approximations of a Von Koch snowflake ($\partial\Omega$, in blue) in \mathbb{R}^2 . The interior approximation Ω_k^- lies inside the green dashed line; the exterior approximation Ω_k^+ lies outside the red dotted line.

Proof. Since Ω is bounded, there exists a cube \mathcal{Q} drawn on Π_0 such that $\Omega \subset \mathcal{Q}$. Then, $\overline{\Omega}_k^+$ is the largest union of cubes of Π_k of connected interior, containing \mathcal{Q} , and such that $\overline{\Omega}_k^+ \subset \Omega^c$.

For all $k \in \mathbb{N}$, denote by \mathcal{C}_k^1 any union of cubes of Π_k included in Ω , of connected interior and such that:

$$\lambda^{(n)}(\mathcal{C}^1_k) = \max_{\mathcal{P} \subset \Pi_k} \left\{ \lambda^{(n)}(A) \mid A = \bigcup \mathcal{P}, \ A \subset \Omega, \ \mathring{A} \text{ is connected} \right\}.$$

Since any dyadic cube from Π_k can be divided into 2^k dyadic cubes from Π_{k+1} , the sequence $(\lambda^{(n)}(\mathcal{C}^1_k))_{k\in\mathbb{N}}$ is non-decreasing. Therefore, there exists $\ell \leq \lambda^{(n)}(\Omega)$ positive such that:

$$\lambda^{(n)}(\mathcal{C}^1_k) \nearrow_{k \to \infty} \ell.$$

There exists $k_0 \in \mathbb{N}$ such that:

$$\forall k \ge k_0, \quad \frac{\ell}{2} < \lambda^{(n)}(\mathcal{C}_k^1) \le \ell.$$

Then, for all $k \geq k_0$, C_k^1 is uniquely defined. Otherwise, assume for some $k \geq k_0$ there were two possible choices for C_k^1 , denoted by C_1 and C_2 . Since Ω is connected, there exists a continuous path $\gamma \subset \Omega$ joining C_1 and C_2 . Since Ω is open, it holds:

$$\forall x \in \gamma, \quad d(x, \partial \Omega) > 0 \qquad i.e. \qquad \min_{x \in \gamma} d(x, \partial \Omega) > 0, \qquad (C.2)$$

by continuity with respect to x of the distance. Therefore, for $\tilde{k} \geq k$ large enough, there exists a set \mathcal{P} of cubes included in Ω from $\Pi_{\tilde{k}}$ joining C_1 and C_2 in the sense that $C_1 \cup \mathcal{P} \cup C_2$ has a connected interior. By definition of $\mathcal{C}^1_{\tilde{k}}$, we can deduce:

$$\lambda^{(n)}(\mathcal{C}^1_{\tilde{k}}) \ge \lambda^{(n)}(C_1) + \lambda^{(n)}(C_2) > \ell,$$

which is absurd since $\tilde{k} \geq k_0$. Therefore, \mathcal{C}_k^1 is uniquely defined, which means Ω_k^- is well-defined.

Remark C.2. In the proof of Lemma C.1, it is also shown $(\Omega_k^-)_{k\in\mathbb{N}}$ and $(\Omega_k^+)_{k\in\mathbb{N}}$ form non-decreasing sequences for k large enough.

We may now state that the dyadic approximations go to the approximated domain as the grid becomes thinner.

Theorem C.3. Let Ω be an arbitrary bounded domain of \mathbb{R}^n such that $\overline{\Omega}^c$ is connected too. Then $\Omega_k^- \sqcup \Omega_k^+ \longrightarrow \mathbb{R}^n \setminus \partial \Omega$.

Proof. From Remark C.2, let $k_0 \in \mathbb{N}$ be such that $(\Omega_k^-)_{k \geq k_0}$ is non-decreasing. Then it holds:

$$\Omega_k^- \xrightarrow[k \to \infty]{} \Omega_\infty^- := \bigcup_{k \ge k_0} \Omega_k^- \ \subset \ \Omega.$$

Let $x \in \Omega$. Then $d(x, \partial \Omega) > 0$. In the same way as for (C.2), there exists $k \geq k_0$ so that:

$$\{x\} \cup \Omega_{k_0}^- \subset \Omega_{\tilde{k}}^-.$$

Hence $x \in \Omega_{\infty}^{-}$, and $\Omega_{\infty}^{-} = \Omega$. To show $\Omega_{k}^{+} \to \overline{\Omega}^{c}$, we proceed in the same way considering $x \in \overline{\Omega}^{c}$ a path on $\overline{\Omega}^c$ linking x and a large square containing Ω .

Remark C.4. Lemma C.1, Remark C.2 and Proposition C.3 can easily be generalized in the case of an arbitrary (potentially unbounded) open set Ω such that Ω and $\overline{\Omega}^{c}$ have a finite number of connected components.

C.2Convergence of the Dirichlet harmonic extension operators

C.2.1 Convergence along a sequence of Hilbert spaces

In this subsection, we recall the definition of convergence of Hilbert spaces from [65] and of elements along a converging sequence of Hilbert spaces. In addition, we establish a few results to have a better understanding of that notion.

Definition C.5. (Convergence of Hilbert spaces) A sequence $(H_k)_{k \in \mathbb{N}}$ of Hilbert spaces is said to converge towards a Hilbert space H through $(C, (\Phi_k)_{k \in \mathbb{N}})$ (in the sense of Kuwae-Shioya [65]) where C is dense in H and $(\Phi_k : C \rightarrow$ $H_k)_{k\in\mathbb{N}}$ is a sequence of linear maps if it holds:

$$\forall f \in C, \quad \lim_{k \to +\infty} \|\Phi_k f\|_{H_k} = \|f\|_H. \tag{C.3}$$

Such a convergence will be denoted by: $H_k \xrightarrow{ks}_{k \to \infty} H$ through $(C, (\Phi_k)_{k \in \mathbb{N}})$.

As we could expect, that notion of convergence in all generality is not very strong. Let us point out two distasteful peculiarities of that notion of limit:

- 1. if $(H_k)_{k\in\mathbb{N}} \xrightarrow{ks} H$, then $(H_k)_{k\in\mathbb{N}} \xrightarrow{ks} V$ for all V isometric to H;
- 2. if H_k is a Hilbert subspace of V_k for all $k \in \mathbb{N}$ and $(H_k)_{k \in \mathbb{N}} \xrightarrow{ks} H$ through $(C, (\Phi_k)_{k \in \mathbb{N}})$, then $(V_k)_{k \in \mathbb{N}} \xrightarrow{ks} H$ through $(C, (\Phi_k)_{k \in \mathbb{N}})$ as well (up to changing the target of the Φ_k).

However, the convergence of Definition C.5 is compatible with certain structures such as dual spaces and direct sums of Hilbert spaces.

Lemma C.6. Assume $(H_k)_{k \in \mathbb{N}} \xrightarrow{ks} H$ through $(C, (\Phi_k)_{k \in \mathbb{N}})$ in the sense of Definition C.5. Denote by $j_k : H_k \to H'_k$ and $j : H \to H'$ the Riesz isometries between the Hilbert spaces and their dual spaces. Then it holds:

$$(H'_k)_{k\in\mathbb{N}} \xrightarrow{ks} H' \quad through \quad (j(C), (j_k \circ \Phi_k \circ j^{-1})_{k\in\mathbb{N}}).$$

Lemma C.7. Let $(H_k)_{k\in\mathbb{N}}$, $(V_k)_{k\in\mathbb{N}}$ be sequences of Hilbert spaces such that $H_k \xrightarrow{ks} H$ and $V_k \xrightarrow{ks} V$. Assume, for all $k \in \mathbb{N}$, $H_k + V_k = H_k \oplus V_k$ and $H + V = H \oplus V$, and consider the norms on those sums defined by:

$$\|\cdot\|_{H_k\oplus V_k} := \sqrt{\|\cdot\|_{H_k}^2 + \|\cdot\|_{V_k}^2} \quad and \quad \|\cdot\|_{H\oplus V} := \sqrt{\|\cdot\|_A^2 + \|\cdot\|_B^2}$$

Then it holds: $H_k \oplus V_k \xrightarrow{ks} H \oplus V$.

Proof. Assume that $H_k \xrightarrow{ks} H$ through $(\mathcal{C}_H, (\Phi_k^H)_{k \in \mathbb{N}})$, and that $V_k \xrightarrow{ks} V$ through $(\mathcal{C}_V, (\Phi_k^V)_{k \in \mathbb{N}})$. Then $\mathcal{C}_H \oplus \mathcal{C}_V$ is dense in $H \oplus V$ and defining:

$$\begin{split} \Psi_k : \mathcal{C}_H \oplus \mathcal{C}_V \longmapsto H_k \oplus V_k \\ u + v \longrightarrow \Phi_k^H u + \Phi_k^V v, \end{split}$$

then it holds: $\forall w \in \mathcal{C}_H \oplus \mathcal{C}_V, \ \|\Psi_k w\|_{H_k \oplus V_k} \longrightarrow \|w\|_{H \oplus V}.$

In our study, we are interested in the convergence of H^1 spaces of converging domains.

Proposition C.8. Let $(\Omega_k)_{k\in\mathbb{N}}$ be a sequence of two-sided extension domains and Ω be another extension domain such that $\Omega_k \xrightarrow{1} \Omega$ (in the sense of characteristic functions). Then it holds:

$$H^1(\Omega_k) \oplus H^1(\Omega_k^c) \xrightarrow[k \to \infty]{ks} H^1(\Omega) \oplus H^1(\Omega^c).$$

Proof. Let E_{Ω} denote the extension operator from $H^1(\Omega)$ to $H^1(\mathbb{R}^n)$. Let $u \in H^1(\Omega)$. From the convergence in the sense of characteristic functions, it holds:

$$\mathbb{1}_{\Omega_k} |\mathbf{E}_{\Omega} u|^2 \xrightarrow{a.e.} \mathbb{1}_{\Omega} |\mathbf{E}_{\Omega} u|^2.$$

Since $E_{\Omega} u \in L^2(\mathbb{R}^n)$, by dominated convergence, it holds:

$$\int_{\mathbb{R}^n} \mathbb{1}_{\Omega_k} |\mathcal{E}_{\Omega} u|^2 \, \mathrm{d}x \longrightarrow \int_{\mathbb{R}^n} \mathbb{1}_{\Omega} |\mathcal{E}_{\Omega} u|^2 \, \mathrm{d}x$$

In the same way:

$$\int_{\mathbb{R}^n} \mathbb{1}_{\Omega_k} |\nabla \mathcal{E}_{\Omega} u|^2 \, \mathrm{d}x \longrightarrow \int_{\mathbb{R}^n} \mathbb{1}_{\Omega} |\nabla \mathcal{E}_{\Omega} u|^2 \, \mathrm{d}x,$$

hence $\|\mathbf{E}_{\Omega} u\|_{H^1(\Omega_k)} \to \|\mathbf{E}_{\Omega} u\|_{H^1(\Omega)}$. Therefore, $H^1(\Omega_k) \xrightarrow{ks} H^1(\Omega)$. In the same way, $H^1(\Omega_k^c) \xrightarrow{ks} H^1(\Omega^c)$. By Lemma C.7, we can deduce:

$$H^1(\Omega_k) \oplus H^1(\Omega_k^c) \xrightarrow[k \to \infty]{ks} H^1(\Omega) \oplus H^1(\Omega^c).$$

In the case of a converging sequence of Hilbert spaces, we may consider a converging vectors along that sequence.

Definition C.9. (Strong convergence of vectors in moving Hilbert spaces) Let $(H_k)_{k\in\mathbb{N}}$ be a sequence of Hilbert spaces converging towards a Hilbert space H through $(C, (\Phi_k)_{k\in\mathbb{N}})$. A sequence $(f_k)_{k\in\mathbb{N}}$ with $f_k \in H_k$ – denoted by $(f_k \in H_k)_{k\in\mathbb{N}}$ – is said to converge (strongly) towards an element $f \in H$ (in the sense of Kuwae-Shioya [65]) if there exists a sequence $(g_j) \in C^{\mathbb{N}}$ with $g_j \xrightarrow{H} f$, such that:

$$\lim_{j \to +\infty} \left(\limsup_{k \to +\infty} \|\Phi_k g_j - f_k\|_{H_k} \right) = 0.$$

We shall denote such convergence by: $(f_k \in H_k)_{k \in \mathbb{N}} \xrightarrow{ks} f \in H$.

Although the notion of convergence described in Definition C.9 can seem rather complex, it becomes less intricate is we assume the convergence of the corresponding Hilbert spaces is 'nice enough'.

Lemma C.10. Let $(H_k)_{k\in\mathbb{N}}$ be a sequence of Hilbert spaces and H be another Hilbert space such that $(H_k)_{k\in\mathbb{N}} \xrightarrow{ks} H$ through $(H, (\Phi_k)_{k\in\mathbb{N}})$ where $(\Phi_k \in \mathcal{L}(H, H_k))_{k\in\mathbb{N}}$ is uniformly bounded. Then for all $(u_k \in H_k)_{k\in\mathbb{N}}$ and $u \in H$, it holds:

$$u_k \xrightarrow{ks}_{k \to \infty} u \quad \Longleftrightarrow \quad \|\Phi_k u - u_k\|_{H_k} \xrightarrow{k \to \infty} 0$$

Proof. The converse follows directly from the definition of convergence of vectors. Assume $(u_k \in H_k)_{k \in \mathbb{N}} \xrightarrow{ks} u \in H$. There exists $(v_m) \in H^{\mathbb{N}}$ such that:

$$v_m \xrightarrow[m \to \infty]{H} u$$
 and $\limsup_{k \to \infty} \|\Phi_k v_m - u_k\|_{H_k} \xrightarrow[m \to \infty]{} 0.$

From

$$\|\Phi_k u - u_k\|_{H_k} \le C_{\Phi} \|v_m - u\|_H + \|\Phi_k v_m - u_k\|_{H_k},$$

where C_{Φ} comes from the uniform boundedness of $(\Phi_k)_{k\in\mathbb{N}}$, the implication holds.

The notion of convergence of vectors allows to define the convergence of bounded operators in a natural way.

Definition C.11. (Strong convergence of bounded operators in moving Hilbert spaces) Let $(H_k)_{k\in\mathbb{N}}$ be a sequence of Hilbert spaces converging towards a Hilbert space H in the sense of Definition C.5. Let $(B_k \in \mathcal{L}(H_k))_{k\in\mathbb{N}}$ and $B \in \mathcal{L}(H)$ be linear bounded operators. It is said that the sequence $(B_k)_{k\in\mathbb{N}}$ converges towards the operator B (in the sense of Kuwae-Shioya [65]) if for every converging sequence $(f_k \in H_k)_{k\in\mathbb{N}} \xrightarrow{ks} f \in H$, it holds:

$$B_k f_k \xrightarrow[k \to \infty]{k \to \infty} Bf.$$

We shall denote such convergence by: $(B_k \in \mathcal{L}(H_k))_{k \in \mathbb{N}} \xrightarrow{ks} B \in \mathcal{L}(H).$

In the case of a single Hilbert space, the convergence of unbounded operators can be established by means of the associated quadratic forms [78]. That notion can be generalized to a sequence of Hilbert spaces as well.

Definition C.12. (Weak convergence of vectors in moving Hilbert spaces) Let $(H_k)_{k\in\mathbb{N}}$ be a sequence of Hilbert spaces converging towards a Hilbert space H in the sense of Definition C.5. A sequence $(f_k \in H_k)_{k\in\mathbb{N}}$ is said to weakly converge towards an element $f \in H$ (in the sense of Kuwae-Shioya [65]) if for any converging sequence $(g_k \in H_k)_{k\in\mathbb{N}}$ with limit $g \in H$, it holds:

$$\langle f_k, g_k \rangle_{H_k} \longrightarrow \langle f, g \rangle_H \quad as \ k \to +\infty.$$

We shall denote such convergence by: $(f_k \in H_k)_{k \in \mathbb{N}} \xrightarrow{ks} f \in H$.

Definition C.13. (Convergence of quadratic forms in the sense of Kuwae-Shioya-Mosco) Let $(H_k)_{k\in\mathbb{N}}$ be a sequence of Hilbert spaces converging towards a Hilbert space H in the sense of Definition C.5. Let $(Q_k)_{k\in\mathbb{N}}$ be a sequence of quadratic forms such that Q_k be defined on H_k , and Q be a quadratic form on H. $(Q_k)_{k\in\mathbb{N}}$ is said to converge towards Q in the sense of Kuwae-Shioya-Mosco if it holds:

(i)
$$\forall f \in H, \ \forall (f_k \in H_k)_{k \in \mathbb{N}}, \quad \left(f_k \xrightarrow{ks} f \implies \liminf_k Q_k(f_k) \ge Q(f)\right),$$

(*ii*)
$$\forall f \in H, \exists (f_k \in H_k)_{k \in \mathbb{N}}, (f_k \xrightarrow{ks} f and \limsup_k Q_k(f_k) \le Q(f)).$$

We shall denote such convergence by: $Q_k \xrightarrow{ksm}_{k\to\infty} Q$ as $k \to +\infty$.

We refer to [65] for properties of the convergence of quadratic forms in the sense of Kuwae-Shioya-Mosco.

C.2.2 Case of the Dirichlet harmonic extension operators

If we are to consider the convergence of the Dirichlet harmonic extension operators from Subsection 3.1 along a sequence of converging domains, then all of them are defined on different spaces (the trace space on each domain). To do so, we must start with proving the convergence of the trace spaces under our convergence hypotheses.

Proposition C.14. Let $(\Omega_k)_{k \in \mathbb{N}}$ be a sequence of two-sided extension domains and let Ω be another two-sided extension domain such that:

$$\forall v \in H^1(\mathbb{R}^n), \quad \|\mathrm{Tr}^{\partial\Omega_k} v\|_{\mathcal{B}(\partial\Omega_k)} \xrightarrow[k \to \infty]{} \|\mathrm{Tr}^{\partial\Omega} v\|_{\mathcal{B}(\partial\Omega)}. \tag{C.4}$$

Then $(\mathcal{B}(\partial\Omega_k))_{k\in\mathbb{N}} \xrightarrow[k\to\infty]{ks} \mathcal{B}(\partial\Omega)$ through $(\mathcal{B}(\partial\Omega), (\operatorname{Tr}^{\partial\Omega_k} \circ \operatorname{H}_{\Omega})_{k\in\mathbb{N}})$, where $\operatorname{H}_{\Omega}$: $\mathcal{B}(\partial\Omega) \to H^1(\mathbb{R}^n)$ denotes the harmonic extension from $\partial\Omega$ to \mathbb{R}^n vanishing at infinity.

Proof. For all $f \in \mathcal{B}(\partial \Omega)$, it holds:

$$\|\operatorname{Tr}^{\partial\Omega_{k}}\operatorname{H}_{\Omega}f\|_{\mathcal{B}(\partial\Omega_{k})} \xrightarrow[k\to\infty]{} \|\operatorname{Tr}^{\partial\Omega}\operatorname{H}_{\Omega}f\|_{\mathcal{B}(\partial\Omega)} = \|f\|_{\mathcal{B}(\partial\Omega)}.$$

An important remark at this point is that, by Theorem 2.9, all the trace operators $\operatorname{Tr}^{\partial\Omega_k} : H^1(\mathbb{R}^n) \to \mathcal{B}(\partial\Omega_k)$ have norm at most 1 (note the domains of the operators discussed here). Therefore, Lemma C.10 applies to the convergence of the \mathcal{B} spaces established in the previous proposition. Since we are using the harmonic extension $\operatorname{H}_{\Omega}$, we prove harmonicity is preserved along a sequence of converging domains

Lemma C.15. Let $(\Omega_k)_{k\in\mathbb{N}}$ be a sequence of two-sided extension domains of \mathbb{R}^n and Ω be another two-sided extension domain, such that $\Omega_k \xrightarrow{1,K} \Omega$. Let $(u_k)_{k\in\mathbb{N}} \in H^1(\mathbb{R}^n)^{\mathbb{N}}$ such that u_k is harmonic on $\Omega_k \cup \Omega_k^c$ for all k. Then, if $u_k \to u \in H^1(\mathbb{R}^n)$, then u is harmonic on $\Omega \cup \Omega^c$.

Proof. From the convergence in the sense of characteristic functions, it holds $H^1(\Omega_k) \to H^1(\Omega)$ and the statement makes sense. Let $\phi \in C_0^{\infty}(\Omega)$. From the compact convergence, supp $\phi \subset \Omega_k$ for k large enough, and it holds:

$$\langle u, \Delta \phi \rangle_{H^1(\Omega)} = \lim_{k \to \infty} \langle u_k, \Delta \phi \rangle_{H^1(\Omega_k)} = 0,$$

hence the harmonicity of u on Ω . In the same way, we prove the harmonicity of u on Ω^c .

The final result which we need to prove is the convergence of the Dirichlet (1-)harmonic extension operators is the convergence of the trace operators on harmonic functions.

Proposition C.16. Let $(\Omega_k)_{k\in\mathbb{N}}$ be a sequence of two-sided extension domains of \mathbb{R}^n , and Ω be another two-sided extension domain satisfying the boundary norm convergence (C.4) and such that $\Omega_k \xrightarrow{\mathbb{1},K} \Omega$. Then, if $(u_k)_{k\in\mathbb{N}} \in H^1(\mathbb{R}^n)^{\mathbb{N}}$ such that u_k is harmonic on $\Omega_k \cup \Omega_k^c$ for all k, and if $u_k \to u \in H^1(\mathbb{R}^n)$, then it holds $\operatorname{Tr}^{\partial\Omega_k} u_k \xrightarrow{ks} \operatorname{Tr}^{\partial\Omega} u$.

Proof. If $u_k \xrightarrow{ks} u \in H^1(\Omega)$, then, by Lemma C.15, u is harmonic on Ω . It holds:

$$\|\mathrm{Tr}^{\partial\Omega_k}\underbrace{\mathrm{E}_{\Omega}\mathrm{Tr}^{\partial\Omega}}_{\mathrm{id}_{H^1(\mathbb{R}^n)}}u - \mathrm{Tr}^{\partial\Omega_k}u_k\|_{\mathcal{B}(\partial\Omega_k)} \le \|u - u_k\|_{H^1(\mathbb{R}^n)},$$

and Lemma C.10 yields the result.

Using the results proved above, we establish the convergence of the Dirichlet harmonic extension operators.

Theorem C.17. Let $(\Omega_k)_{k \in \mathbb{N}}$ be a sequence of two-sided extension domains of \mathbb{R}^n , and Ω be another two-sided extension domain such that $\Omega_k \xrightarrow{\mathbb{1},K} \Omega$ and satisfying the boundary norm convergence (C.4). Then,

(i) if the H^1 -extension operators $E_{\Omega_k} : H^1(\Omega_k) \to H^1(\mathbb{R}^n), k \in \mathbb{N}$, are uniformly bounded, then:

$$\tilde{\mathcal{D}}_{\Omega_k} \in \mathcal{L}(\mathcal{B}(\partial\Omega_k), H^1(\Omega_k)) \xrightarrow{ksm} \tilde{\mathcal{D}}_{\Omega} \in \mathcal{L}(\mathcal{B}(\partial\Omega), H^1(\Omega)),$$

in the sense of Definition C.13;

(ii) if the H¹-extension operators $E_{\Omega_k^c}$: $H^1(\Omega_k^c) \to H^1(\mathbb{R}^n)$, $k \in \mathbb{N}$, are uniformly bounded, then:

$$\tilde{\mathcal{D}}_{\Omega_k^c} \in \mathcal{L}(\mathcal{B}(\partial \Omega_k), H^1(\Omega_k^c)) \xrightarrow{ksm}{k \to \infty} \tilde{\mathcal{D}}_{\Omega^c} \in \mathcal{L}(\mathcal{B}(\partial \Omega), H^1(\Omega^c)),$$

where the exterior extensions are chosen as vanishing at infinity.

Proof. (i) Consider the quadratic forms associated to the Dirichlet harmonic extension operators:

$$Q_{k}: \mathcal{B}(\partial\Omega_{k}) \longrightarrow \mathbb{R}_{+} \qquad \text{for } \tilde{\mathcal{D}}_{\alpha_{k}} f_{k} ||_{H^{1}(\Omega_{k})}^{2} = ||\mathcal{H}_{\partial\Omega_{k}}f_{k}|_{\Omega_{k}}||_{H^{1}(\Omega_{k})}^{2} \qquad \text{for } \tilde{\mathcal{D}}_{\Omega_{k}},$$
$$Q: \mathcal{B}(\partial\Omega) \longrightarrow \mathbb{R}_{+} \qquad f \longmapsto ||\tilde{\mathcal{D}}f||_{H^{1}(\Omega)}^{2} = ||\mathcal{H}_{\partial\Omega}f|_{\Omega}||_{H^{1}(\Omega)}^{2} \qquad \text{for } \tilde{\mathcal{D}}_{\Omega},$$

where $\mathcal{H}_{\partial\Omega_k}$ denotes the harmonic extension from $\partial\Omega_k$ to \mathbb{R}^n and $\mathcal{H}_{\partial\Omega}$ from $\partial\Omega$ to \mathbb{R}^n , vanishing at infinity. Let $f \in \mathcal{B}(\partial\Omega)$ and $(f_k \in \mathcal{B}(\partial\Omega_k))_{k\in\mathbb{N}}$ are such that $f_k \stackrel{ks}{\longrightarrow} f$. Without loss of generality, we may assume that the sequence

 $(Q_k(f_k)) \in \mathbb{R}^{\mathbb{N}}_+$ has a limit: that consists in considering a subsequence which goes to the lowest subsequential limit. We may also assume that:

$$L := \lim_{k \to \infty} Q_k(f_k) < +\infty, \tag{C.5}$$

otherwise condition (i) directly holds. In that case, it holds:

$$\sup_{k} \left\| \mathbf{H}_{\partial\Omega_{k}} f_{k} \right\|_{\Omega_{k}} \right\|_{H^{1}(\Omega_{k})} < +\infty$$

By uniform boundedness of the extension operators, we may deduce:

$$\sup_{k} \left\| \mathcal{H}_{\partial\Omega_{k}} f_{k} \right\|_{H^{1}(\mathbb{R}^{n})} < +\infty.$$
(C.6)

Consequently we may, passing to a subsequence, assume that there is some $v \in H^1(\mathbb{R}^n)$ such that

$$H_{\partial\Omega_k}f_k \longrightarrow v \text{ in } H^1(\mathbb{R}^n).$$
 (C.7)

From the convergence in the sense of characteristic functions, we may deduce:

$$\mathbb{1}_{\Omega_k} \mathcal{H}_{\partial \Omega_k} f_k \xrightarrow{H^1(\mathbb{R}^n)} \mathbb{1}_{\Omega} v.$$

Therefore,

$$\|v\|_{H^1(\Omega)}^2 \le \liminf_{k \to \infty} \|\mathcal{H}_{\partial\Omega_k} f_k\|_{H^1(\Omega_k)}^2 = L.$$
(C.8)

Moreover, since $H_{\partial\Omega_k} f_k$ is harmonic on Ω_k for all $k \in \mathbb{N}$, v is harmonic on Ω and it holds by Proposition C.16:

$$\operatorname{Tr}^{\partial\Omega_{k}} \operatorname{H}_{\partial\Omega_{k}} f_{k} \xrightarrow{ks} \operatorname{Tr}^{\partial\Omega} v$$

$$= f_{k} \xrightarrow{ks} f$$

hence $\operatorname{Tr}_{\partial\Omega} v = f$. By energy minimization and (C.8) it follows that:

$$Q(f) = \|\mathbf{H}_{\partial\Omega}f\|_{H^{1}(\Omega)}^{2} \le \|v\|_{H^{1}(\Omega)}^{2} \le L,$$

which proves condition (i).

Suppose now that $f \in \mathcal{B}(\partial\Omega)$. For any $k \in \mathbb{N}$, let us denote $f_k := \operatorname{Tr}_{\partial\Omega_k} \operatorname{H}_{\partial\Omega} f$. Then it holds $f_k \xrightarrow{ks} f$ and, by energy minimization:

$$Q_k(f_k) = \|\mathbf{H}_{\partial\Omega_k} f_k\|_{H^1(\Omega_k)}^2 \le \|\mathbf{H}_{\partial\Omega} f\|_{H^1(\Omega_k)}^2.$$

From the convergence in the sense of characteristic functions, we may deduce:

$$\lim_{k \to \infty} \|\mathbf{H}_{\partial\Omega} f\|_{H^1(\Omega_k)}^2 = \|\mathbf{H}_{\partial\Omega} f\|_{H^1(\Omega)}^2 = Q(f).$$

That proves the convergence in the Kuwae-Shioya-Mosco sense.

The proof of (ii) is similar.

References

- H. AMMARI AND H. KANG, Reconstruction of Small Inhomogeneities from Boundary Measurements, Springer Science & Business Media, springer-verlag ed., Aug. 2004.
- [2] H. AMMARI, H. KANG, AND H. LEE, Layer Potential Techniques in Spectral Analysis, vol. 153 of Mathematical Surveys and Monographs, American Mathematical Society, Mar. 2009.
- [3] W. ARENDT AND A. F. M. TER ELST, *The Dirichlet-to-Neumann Operator on Exterior Domains*, Potential Analysis, 43 (2015), pp. 313–340.
- [4] K. ARFI AND A. ROZANOVA-PIERRAT, Dirichlet-to-Neumann or Poincaré-Steklov operator on fractals described by d-sets, Discrete & Continuous Dynamical Systems - S, 12 (2019), pp. 1–26.
- [5] D. N. ARNOLD AND P. J. NOON, Boundary integral equations of the first kind for the heat equation, Boundary elements IX, 3 (1987), pp. 213–229.
- [6] M. BARLOW, R. BASS, T. KUMAGAI, AND A. TEPLYAEV, Uniqueness of Brownian motion on Sierpiński carpets, Journal of the European Mathematical Society, (2010), pp. 655–701.
- [7] A. BARTON, Layer potentials for general linear elliptic systems, Mar. 2017.
- [8] R. F. BASS, Probabilistic techniques in analysis, Springer Science & Business Media, 1994.
- [9] —, *Diffusions and Elliptic Operators*, Probability and its Applications, Springer-Verlag, New York, 1998.
- [10] R. F. BASS AND P. HSU, Some Potential Theory for Reflecting Brownian Motion in Holder and Lipschitz Domains, The Annals of Probability, 19 (1991), pp. 486–508.
- [11] R. BAÑUELOS, Martingale Transforms and Related Singular Integrals, Transactions of the American Mathematical Society, 293 (1986), pp. 547– 563.
- [12] R. BAÑUELOS AND F. BAUDOIN, Martingale Transforms and Their Projection Operators on Manifolds, Potential Analysis, 38 (2013), pp. 1071– 1089.
- [13] J. BEHRNDT AND A. F. M. TER ELST, Dirichlet-to-Neumann maps on bounded Lipschitz domains, Journal of Differential Equations, 259 (2015), pp. 5903–5926.

- [14] M. BIEGERT, On Traces of Sobolev Functions on the Boundary of Extension Domains, Proceedings of the American Mathematical Society, 137 (2009), pp. 4169–4176.
- [15] M. BIEGERT AND M. WARMA, Some quasi-linear elliptic equations with inhomogeneous generalized Robin boundary conditions on "bad" domains, Advances in Differential Equations, 15 (2010), pp. 893–924.
- [16] R. M. BLUMENTHAL AND R. K. GETOOR, Markov Processes and Potential Theory, no. 29 in Pure and Applied Mathematics, Academic Press, 1968.
- [17] F. BOYER AND P. FABRIE, Mathematical Tools for the Study of the Incompressible Navier-Stokes Equations and Related Models, Springer Science & Business Media, Nov. 2012.
- [18] H. BREZIS, Analyse fonctionnelle : théorie et applications, Collection Mathématiques appliquées pour la maîtrise, Masson, Paris New York Barcelone, 1987.
- [19] R. M. BROWN, Global Uniqueness in the Impedance-Imaging Problem for Less Regular Conductivities, SIAM Journal on Mathematical Analysis, 27 (1996), pp. 1049–1056.
- [20] —, Recovering the conductivity at the boundary from the Dirichlet to Neumann map: a pointwise result, Journal of Inverse and Ill-posed Problems, 9 (2001), pp. 567–574.
- [21] K. BURDZY, Z.-Q. CHEN, D. MARSHALL, AND K. RAMANAN, Obliquely reflected Brownian motion in nonsmooth planar domains, The Annals of Probability, 45 (2017), pp. 2971–3037.
- [22] A.-P. CALDERÓN, Lebesgue spaces of differentiable functions and distributions, in Proceedings of Symposia in Pure Mathematics, C. Morrey, ed., vol. 4, American Mathematical Society, Providence, Rhode Island, 1961, pp. 33–49.
- [23] A. P. CALDERÓN, Cauchy integrals on Lipschitz curves and related operators, Proceedings of the National Academy of Sciences, 74 (1977), pp. 1324–1327. Publisher: Proceedings of the National Academy of Sciences.
- [24] R. CAPITANELLI AND M. D'OVIDIO, Fractional Cauchy problem on random snowflakes, Journal of Evolution Equations, 21 (2021), pp. 2123– 2140.
- [25] S. N. CHANDLER-WILDE AND D. P. HEWETT, Wavenumber-Explicit Continuity and Coercivity Estimates in Acoustic Scattering by Planar

Screens, Integral Equations and Operator Theory, 82 (2015), pp. 423–449.

- [26] S. N. CHANDLER-WILDE, D. P. HEWETT, AND A. MOIOLA, Sobolev Spaces on Non-Lipschitz Subsets of \$\mathbar{mathbar{mathbar{s}}^n\$ with Application to Boundary Integral Equations on Fractal Screens, Integral Equations and Operator Theory, 87 (2017), pp. 179–224.
- [27] T. CHANG AND K. LEE, Spectral properties of the layer potentials on Lipschitz domains, Illinois Journal of Mathematics, 52 (2008), pp. 463– 472. Publisher: Duke University Press.
- [28] Z.-Q. CHEN, On reflected Dirichlet spaces, Probability Theory and Related Fields, 94 (1992), pp. 135–162.
- [29] —, On reflecting diffusion processes and Skorokhod decompositions, Probability Theory and Related Fields, 94 (1993), pp. 281–315.
- [30] —, Reflecting Brownian motions and a deletion result for Sobolev spaces of order (1, 2), Potential Analysis, 5 (1996), pp. 383–401.
- [31] Z.-Q. CHEN AND M. FUKUSHIMA, Symmetric Markov Processes, Time Change, and Boundary Theory, no. vol. 35 in London Mathematical Society monographs, Princeton University Press, Princeton, 2012. OCLC: ocn724663149.
- [32] K. L. CHUNG AND Z. ZHAO, From Brownian Motion to Schrödinger's Equation, Springer Berlin Heidelberg, Berlin, Heidelberg, 1995. OCLC: 851823586.
- [33] R. R. COIFMAN, P. W. JONES, AND S. SEMMES, Two Elementary Proofs of the L2 Boundedness of Cauchy Integrals on Lipschitz Curves, Journal of the American Mathematical Society, 2 (1989), pp. 553–564. Publisher: American Mathematical Society.
- [34] R. R. COIFMAN, A. MCINTOSH, AND Y. MEYER, L'intégrale de Cauchy définit un opérateur borné sur L2 pour les courbes lipschitziennes, Annals of Mathematics, 116 (1982), pp. 361–387. Publisher: Annals of Mathematics.
- [35] D. COLTON AND R. KRESS, Inverse Acoustic and Electromagnetic Scattering Theory, vol. 93 of Applied Mathematical Sciences, Springer, New York, NY, 2013.
- [36] J. DARDÉ, Méthodes de quasi-réversibilité et de lignes de niveau appliquées aux problèmes inverses elliptiques., PhD thesis, Université Paris Diderot - Paris 7, Jan. 2010.

- [37] E. B. DAVIES, *Linear Operators and their Spectra*, Cambridge Studies in Advanced Mathematics, Cambridge University Press, Cambridge, 2007.
- [38] A. DEKKERS AND A. ROZANOVA-PIERRAT, Dirichlet boundary valued problems for linear and nonlinear wave equations on arbitrary and fractal domains, Journal of Mathematical Analysis and Applications, 512 (2022), p. 126089.
- [39] A. DEKKERS, A. ROZANOVA-PIERRAT, AND A. TEPLYAEV, Mixed boundary valued problems for linear and nonlinear wave equations in domains with fractal boundaries, Calculus of Variations and Partial Differential Equations, 61 (2022), p. 75.
- [40] J. L. DOOB, Classical Potential Theory and Its Probabilistic Counterpart, Classics in Mathematics, Springer, Berlin, Heidelberg, 2001.
- [41] E. B. DYNKIN, *Markov Processes*, Springer, Berlin, Heidelberg, 1965.
- [42] —, Markov Processes, vol. 122 of Grundlehren der mathematischen Wissenschaften, Springer, Berlin, Heidelberg, 1965.
- [43] M. EGERT, R. HALLER-DINTELMANN, AND J. REHBERG, Hardy's Inequality for Functions Vanishing on a Part of the Boundary, Potential Analysis, 43 (2015), pp. 49–78.
- [44] L. ESCAURIAZA, E. B. FABES, AND G. VERCHOTA, On a regularity theorem for weak solutions to transmission problems with internal Lipschitz boundaries, Proceedings of the American Mathematical Society, 115 (1992), pp. 1069–1076.
- [45] L. C. EVANS, Partial Differential Equations, American Mathematical Society, 2010.
- [46] W. FELLER, Diffusion processes in one dimension, Transactions of the American Mathematical Society, 77 (1954), pp. 1–31.
- [47] M. FUKUSHIMA, Y. OSHIMA, AND M. TAKEDA, Dirichlet Forms and Symmetric Markov Processes, De Gruyter, Dec. 2010.
- [48] P. GRISVARD, *Elliptic problems in nonsmooth domains*, no. 69 in Classics in applied mathematics, Society for industrial and applied mathematics, Philadelphia, 2011.
- [49] G. GUANGLU, Q. MINPING, AND M. L. SILVERSTEIN, Normal Derivative for Bounded Domains with General Boundary, Transactions of the American Mathematical Society, 308 (1998).
- [50] P. HAJŁASZ, P. KOSKELA, AND H. TUOMINEN, Sobolev embeddings, extensions and measure density condition, Journal of Functional Analysis, 254 (2008), pp. 1217–1234.

- [51] A. HENROT AND M. PIERRE, Shape variation and optimization: a geometrical analysis, no. 28 in EMS tracts in mathematics, European Mathematical Society, Zürich, Switzerland, 2018.
- [52] M. HINZ, F. MAGOULÈS, A. ROZANOVA-PIERRAT, M. RYNKOVSKAYA, AND A. TEPLYAEV, On the existence of optimal shapes in architecture, Applied Mathematical Modelling, 94 (2021), pp. 676–687.
- [53] M. HINZ, A. ROZANOVA-PIERRAT, AND A. TEPLYAEV, Non-Lipschitz Uniform Domain Shape Optimization in Linear Acoustics, SIAM Journal on Control and Optimization, 59 (2021), pp. 1007–1032. Publisher: Society for Industrial and Applied Mathematics.
- [54] —, Boundary value problems on non-Lipschitz uniform domains: stability, compactness and the existence of optimal shapes, Asymptotic Analysis, (2023), pp. 1–37.
- [55] E. P. HSU, Stochastic analysis on manifolds, no. v. 38 in Graduate studies in mathematics, American Mathematical Society, Providence, R.I, 2002.
- [56] N. IKEDA AND S. WATANABE, Stochastic differential equations and diffusion processes, no. v. 24 in North-Holland mathematical library, North-Holland Pub. Co. ; Kodansha ; Sole distributors for the U.S.A. and Canada, Elsevier North-Holland, Amsterdam ; New York : Tokyo : New York, NY, 1981.
- [57] A. JONSSON AND H. WALLIN, *Function Spaces on Subsets of Rn*, Harwood Academic Publishers, 1984. Google-Books-ID: dBKoAAAAIAAJ.
- [58] —, The dual of Besov spaces on fractals, Studia Mathematica, 3 (1995), pp. 285–300.
- [59] H. KANG AND J. K. SEO, The layer potential technique for the inverse conductivity problem, Inverse Problems, 12 (1996), pp. 267–278. Publisher: IOP Publishing.
- [60] —, Identification of domains with near-extreme conductivity: global stability and error estimates, Inverse Problems, 15 (1999), pp. 851–867. Publisher: IOP Publishing.
- [61] —, Recent Progress in the Inverse Conductivity Problem with Single Measurement, in Inverse problems and related topics, Chapman and Hall/CRC, 2000. Num Pages: 12.
- [62] O. D. KELLOGG, Foundations of Potential Theory, Courier Corporation, Jan. 1953. Google-Books-ID: TxlfQi46CvEC.

- [63] D. KHAVINSON, M. PUTINAR, AND H. S. SHAPIRO, Poincaré's Variational Problem in Potential Theory, Archive for Rational Mechanics and Analysis, 185 (2007), pp. 143–184.
- [64] J. KRÁL, Integral Operators in Potential Theory, vol. 823 of Lecture Notes in Mathematics, Springer, Berlin, Heidelberg, 1980.
- [65] K. KUWAE AND T. SHIOYA, Convergence of spectral structures: a functional analytic theory and its applications to spectral geometry, Communications in Analysis and Geometry, 11 (2003), pp. 599–673. Publisher: International Press of Boston.
- [66] A. R. LALIENA AND F.-J. SAYAS, A distributional version of Kirchhoff's formula, Journal of Mathematical Analysis and Applications, 359 (2009), pp. 197–208.
- [67] M. R. LANCIA, A Transmission Problem with a Fractal Interface, Zeitschrift für Analysis und ihre Anwendungen, 21 (2002), pp. 113–133.
- [68] M. R. LANCIA AND P. VERNOLE, Venttsel' problems in fractal domains, Journal of Evolution Equations, 14 (2014), pp. 681–712. Publisher: Springer.
- [69] J.-F. LE GALL, Mouvement brownien, martingales et calcul stochastique, vol. 71 of Mathématiques et Applications, Springer, Berlin, Heidelberg, 2013.
- [70] H. LIU, M. YAMAMOTO, AND J. ZOU, Reflection principle for the Maxwell equations and its application to inverse electromagnetic scattering, Inverse Problems, 23 (2007), p. 2357.
- [71] G. LU AND B. OU, A Poincaré inequality on Rn and its application to potential fluid flows in space, IMA Preprint Series, (2004).
- [72] B. MANDELBROT, How Long Is the Coast of Britain? Statistical Self-Similarity and Fractional Dimension, Science, 156 (1967), pp. 636–638.
- [73] B. B. MANDELBROT, The Fractal Geometry of Nature, Henry Holt and Company, 1983.
- [74] J. MARSCHALL, The trace of Sobolev-Slobodeckij spaces on Lipschitz domains, manuscripta mathematica, 58 (1987), pp. 47–65.
- [75] W. C. H. MCLEAN, Strongly Elliptic Systems and Boundary Integral Equations, Cambridge University Press, Jan. 2000. Google-Books-ID: RILqjEeMfK0C.

- [76] M. MELNIKOV AND X. TOLSA, Estimate of the Cauchy Integral over Ahlfors Regular Curves, in Selected Topics in Complex Analysis, V. Y. Eiderman and M. V. Samokhin, eds., vol. 158, Birkhäuser-Verlag, Basel, 2005, pp. 159–176. Series Title: Operator Theory: Advances and Applications.
- [77] I. MITREA, On the Spectra of Elastostatic and Hydrostatic Layer Potentials on Curvilinear Polygons, Journal of Fourier Analysis and Applications, 8 (2002), pp. 443–488.
- [78] U. MOSCO, Convergence of convex sets and of solutions of variational inequalities, Advances in Mathematics, 3 (1969), pp. 510–585.
- [79] N. I. MUSKHELISHVILI, Singular Integral Equations, Springer Netherlands, Dordrecht, 1977.
- [80] J.-C. NÉDÉLEC, Acoustic and electromagnetic equations: integral representations for harmonic problems, vol. 144 of Applied Mathematical Sciences, Springer, New York, 2011. OCLC: 1064504586.
- [81] S. OLVER, Computing the Hilbert transform and its inverse, Mathematics of Computation, 80 (2011), pp. 1745–1767.
- [82] V. G. PAPANICOLAOU, The probabilistic solution of the third boundary value problem for second order elliptic equations, Probability Theory and Related Fields, 87 (1990), pp. 27–77.
- [83] K.-M. PERFEKT AND M. PUTINAR, The Essential Spectrum of the Neumann-Poincaré Operator on a Domain with Corners, Archive for Rational Mechanics and Analysis, 223 (2017), pp. 1019–1033.
- [84] A. ROZANOVA-PIERRAT, Generalization of Rellich-Kondrachov Theorem and Trace Compactness for Fractal Boundaries, in Fractals in Engineering: Theoretical Aspects and Numerical Approximations, M. R. Lancia and A. Rozanova-Pierrat, eds., SEMA SIMAI Springer Series, Cham, 2021, Springer International Publishing, pp. 155–173.
- [85] F.-J. SAYAS, Retarded Potentials and Time Domain Boundary Integral Equations, vol. 50 of Springer Series in Computational Mathematics, Springer International Publishing, Cham, 2016.
- [86] M. L. SILVERSTEIN, Symmetric Markov Processes, Springer, Nov. 2006. Google-Books-ID: 5TJ7CwAAQBAJ.
- [87] E. M. STEIN, Singular Integrals and Differentiability Properties of Functions, Princeton University Press, 1970.

- [88] —, Harmonic Analysis: Real-variable Methods, Orthogonality, and Oscillatory Integrals, Princeton University Press, Aug. 1993. Google-Books-ID: ljcOSMK7t0EC.
- [89] O. STEINBACH AND W. WENDLAND, On C. Neumann's Method for Second-Order Elliptic Systems in Domains with Non-smooth Boundaries, Journal of Mathematical Analysis and Applications, 262 (2001), pp. 733– 748.
- [90] A. D. VENTTSEL', On Boundary Conditions For Multidimensional Diffusion Processes, Theory of Probability & Its Applications, 4 (1959), pp. 164–177.
- [91] G. VERCHOTA, Layer potentials and regularity for the Dirichlet problem for Laplace's equation in Lipschitz domains, Journal of Functional Analysis, 59 (1984), pp. 572–611.
- [92] V. S. VLADIMIROV, Equations of Mathematical Physics, Marcel Dekker, Inc., 1971.
- [93] H. VOGT AND J. VOIGT, Wentzell boundary conditions in the context of Dirichlet forms, Advances in Differential Equations, 8 (2003), pp. 821– 842.
- [94] H. WALLIN, The trace to the boundary of Sobolev spaces on a snowflake, manuscripta mathematica, 73 (1991), pp. 117–125.
- [95] R. WEDER, The Aharonov-Bohm effect and time-dependent inverse scattering theory, Inverse Problems, 18 (2002), p. 1041.
- [96] H. WEYL, The method of orthogonal projection in potential theory, Duke Mathematical Journal, 7 (1940).
- [97] M. WILLEM, Analyse harmonique réelle, Collection Méthodes, Hermann, 1995.

Résumé

Que ce soit dans le domaine médical ou pour la détection d'impuretés dans les alliages métalliques, les problèmes d'imagerie trouvent de nombreux domaines d'application. D'un point de vue mathématique, ces problèmes sont modélisés par des problèmes de transmission. Dans cette thèse, on étudie ces problèmes dans le cadre de domaines dont le bord peut être irrégulier, par exemple fractal. En se plaçant sur une large classe de domaines d'extension de Sobolev, que l'on nomme domaines d'extension bilatéraux, on généralise des notions relatives au problèmes de transmission pour des bords Lipschitziens telles que les opérateurs potentiels simple et double couches, Poincaré-Steklov et Neumann-Poincaré. En particulier, on généralise la formule de représentation des solutions du problème de transmission à deux phases pour des domaines d'extension. On souligne le lien qu'ont les problèmes de transmission sur des domaines d'extension avec le concept probabiliste de chaîne de Markov, ainsi qu'avec la notion de transformée de Hilbert que l'on généralise dans le cadre non-Lipschitzien. On montre que tout domaine peut être approximé par une suite de domaines dyadiques et l'on prouve la convergence d'extensions harmoniques depuis le bord le long d'une suite convergente de domaines.

Summary

Whether it be in the medical field or for the detection of impurities in metal alloys, imaging problems have many areas of application. From a mathematical point of view, these problems are modelled by transmission problems. In this thesis, we study those problems in the context of domains with irregular boundaries such as fractals. By focusing on a large class of Sobolev extension domains, which we call two-sided extension domains, we generalize notions regarding transmission problems for Lipschitz boundaries such as the single and double layer potential, the Poincaré-Steklov and the Neumann-Poincaré operators. In particular, we generalize the representation formula for the solutions to the two-phase transmission problem to extension domains. The connection of transmission problems on extension domains with the probabilistic concept of the Markov chain is emphasised, and so is their link with the notion of the Hilbert transform, which is generalized to the non-Lipschitz framework. We establish any domain can be approximated by a sequence of dyadic domains, and prove the convergence of harmonic extensions from the boundary along a sequence of converging domains.